Online self-directed learning readiness and learning outcome in online statistics learning environment: The mediating roles of self-efficacy to learn statistics and 2x2 Achievement goal orientations

by

Sangah Lee

A dissertation submitted to the Graduate Faculty of Auburn University in partial fulfillment of the requirements for the Degree of Doctor of Philosophy

> Auburn, Alabama August 5, 2023

Keywords: self-directed learning readiness, self-efficacy, Achievement goal orientations, course satisfaction, statistics learning

Copyright 2023 by Sangah Lee

Approved by

Chih-hsuan Wang, Chair, Professor of Educational Foundations, Leadership, and Technology William Murrah, Co-chair, Associate Professor and Department Head of Educational Foundations, Leadership, and Technology Jill Salisbury-Glennon, Associate Professor of Educational Foundations, Leadership, and Technology Leslie Cordie, Associate Professor of Educational Foundations, Leadership, and Technology

Abstract

This study investigated the understanding of the effects of online self-directed learning readiness (OSDLR) on academic achievement. and course satisfaction for online learning statistics through the mediating variables of self-efficacy to learn statistics (SELS) and 2x2 achievement goal orientation. The current study sought to create awareness and inform learners and tutors engaging in online statistics courses how they can rely on self-directed learning readiness. The present study sought to investigate: i) the relationship between online self-directed learning readiness and academic outcomes as measured by course satisfaction and grades ii) how the relationship between OSDLR and learning outcomes (grade) is mediated by achievement goal orientations iv) how the relationship between OSDLR and affective learning outcome (course satisfaction) is mediated by SELS and v) how the relationship between OSDLR and affective learning outcome (course satisfaction) is mediated by SELS and v) how the relationship between orientations.

This study employs a quantitative research design and a correlational research design utilizing a cross-sectional mediation model to investigate the relationships among variables. Data collection for this study was conducted using an online survey with a set of questionnaires, including part 1 for demographics, part 2 for online learning readiness, part 3 for 2x2 achievement goal orientations, part 4 for self-efficacy to learn statistics, and part 5 for course satisfaction. The participants in the current study were students enrolled in and studying at a large public university in the Southeastern United States who had registered for at least one statistics course delivered online during the 2020 Fall semester.

Data analyses utilized path models, where a series of OLS regression analyses were performed. The analysis was divided into four different models: Model 1-1, Model 1-2, Model 2-

ii

1, and Model 2-2. The study established that OSDLR was positively related to grades. OSDLR is also positively related to SELS, and SELS was not related to the grade, and as such, it was not a significant mediator between OSDLR and grade. It is also established that OSDLR was positively related to mastery-approach goal orientation (MAP) but showed no positive relationship with any other orientations (mastery-avoidance goal orientation (MAV), performance-goal orientation (PAP), or performance-avoidance goal orientation (PAV)) The results also showed that MAP was positively related to the grade; therefore, that MAP is the significant mediator between the relationship between OSDLR and grade. MAV was negatively related to grade and neither PAP nor PAV was significantly associated with the grade. OSDLR was positively associated with course satisfaction, and SELS was positively associated with course satisfaction. Finally, none of the 2x2 achievement goal orientations were associated with course satisfaction.

This study concludes with recommendations based on these findings, limitations of the current study, and implications of the present study. The findings of the current study provide a basis upon which future studies can be developed, as well as a basis upon which better educational programs can be implemented for online learning environments by instructors and institutions of higher learning.

iii

Acknowledgments

I wish to acknowledge and express my sincere gratitude to all the people for their continued support throughout my graduate studies. First, I would like to express my sincere gratitude to my supervisor, Dr. Wang, for her thoughtful and consistent support and guidance on this dissertation. Your invaluable knowledge and plentiful research experience have encouraged me throughout all the time of my academic research and daily life. Furthermore, I would like to thank my dissertation co-chair, Dr. Murrah. Your intuitive feedback and suggestions made me improve my research and analyzing skills to a higher level. Also, I would like to thank my committee members, Dr. Glennon for her thorough feedback and suggestions on theoretical parts of the dissertation and Dr. Cordie for her invaluable opinion and suggestions of online learning and learning outcomes.

Additionally, I would like to express my gratitude to Dr. Wyckoff, the university reader, for reviewing and providing suggestions for this study.

They have all encouraged and supported me immeasurably, and I would not be able to finish this long journey without them. It would be impossible to have better committee members. They spent an enormous time reading the various drafts and always provided me with guidance, invaluable suggestions, revisions, and understanding.

Nobody has been more important to me in the pursuit of this project than the members of my family. I would like to thank my loving and supportive husband Hwanjong Jung and my upcoming babies, my beautiful twin girls who provide unending inspiration. I would like to thank my parents, whose love and guidance are with me in whatever I pursue.

iv

Table of Contents

| Abstractii |
|---|
| Acknowledgmentsiv |
| List of Tablesx |
| List of Figures xi |
| Chapter 1. Introduction1 |
| Overview1 |
| Online Statistics Learning |
| Problem Statement |
| Research Questions |
| Significance of Study7 |
| Limitations of Study7 |
| Definitions of Terms |
| Organization of Study10 |
| Chapter 2. Review of Literature |
| Overview11 |
| Problem Statement |
| Research Questions |
| Self-Directed Learning Theory |
| Self-Directed Learning Readiness |
| Online Self-Directed Learning Readiness |
| Self-efficacy Theory |
| Self-efficacy to Learn Statistics |

| Goal Orientation Theory | |
|--|----|
| Achievement Goal Orientations and Online Learning | |
| Learning Outcome | 41 |
| Course Satisfaction as Affective Learning Outcome | |
| Self-Directed Learning Readiness and Learning Outcomes | |
| Self-Directed Learning Readiness and Self-Efficacy | |
| Self-Directed Learning Readiness and Achievement Goal Orientations | |
| Self-Efficacy and Learning Outcomes | |
| Achievement Goal Orientations and Learning Outcomes | 51 |
| Summary | |
| Chapter 3. Method | 61 |
| Overview | 61 |
| Problem Statement | 61 |
| Research Questions | 62 |
| Research Design | 63 |
| Participants and Sampling Procedure | 64 |
| Instruments | 65 |
| Achievement Goal Questionnaire-Revised (AGQ-R) | 65 |
| Self-Efficacy to Learn Statistics (SELS) | 66 |
| Online Self-Directed Learning Readiness (OSDLR) | 67 |
| Course Satisfaction | 68 |
| Data Collection Procedures | |
| Data Analysis Procedures | 69 |

| The total effect of self-directed learning readiness on grade74 |
|--|
| The total effect of self-directed learning readiness on course satisfaction74 |
| The direct effect of self-directed learning readiness on grade |
| The direct effect of self-directed learning readiness on course satisfaction74 |
| The indirect effect of self-directed learning readiness on grade75 |
| The indirect effect of self-directed learning readiness on course satisfaction75 |
| Summary |
| Chapter 4. Findings |
| Overview79 |
| Problem Statement |
| Research Questions |
| Demographic Findings |
| Measures of Reliability |
| Statistical Assumptions |
| Linearity |
| Normality |
| Independence of errors |
| Homoscedasticity |
| Normally distributed errors |
| Multicollinearity |
| Multivariate outliers |
| Preliminary Analyses |
| Descriptive Findings and Correlations among Measures |

| Results | | |
|----------------------------|---------------------------------------|---------------------|
| OLS Regression | | |
| Research | Questions 1.1 and 1.2 | |
| Research | Question 1.3 | |
| Research | Questions 2.1 and 2.2 | |
| Research | Question 2.3 | |
| Summary | | |
| hapter 5. Summary, Conclus | tions, Implications, Limitations, and | Recommendations for |
| uture Research | | |
| Overview | | |
| Problem Statement | | |
| Research Questions | | |
| Study Overview | | |
| Findings | | |
| Conclusions | | |
| Research Questio | ons 1.1 and 1.2 | |
| Research Questio | on 1.3 | |
| Research Questio | ons 2.1 and 2.2 | |
| Research Questio | on 2.3 | |
| Implications | | |
| Limitations | | |
| Recommendations | | |
| References | | |

| Appendix A | |
|------------|--|
| Appendix B | |
| Appendix C | |
| Appendix D | |

List of Tables

| Table 1 | |
|----------|----|
| Table 2 | 76 |
| Table 3 | |
| Table 4 | |
| Table 5 | |
| Table 6 | |
| Table 7 | |
| Table 8 | |
| Table 9 | |
| Table 10 | |
| Table 11 | |
| Table 12 | |
| Table 13 | |
| Table 14 | |
| Table 15 | |
| Table 16 | |

List of Figures

| Figure 1 | |
|-----------|----|
| Figure 2 | 71 |
| Figure 3 | 71 |
| Figure 4 | |
| Figure 5 | |
| Figure 6 | |
| Figure 7 | |
| Figure 8 | |
| Figure 9 | |
| Figure 10 | |
| Figure 11 | |

Chapter 1 - Introduction

Overview

In recent decades, the education system has been characterized by technological advancements that have enhanced its flexibility, variability, and accessibility. Online learning has been a significant development that has expanded learners' horizons, enabling them to save resources and time, enjoy increased flexibility in interacting with their educators and peers remotely, and access learning materials without geographical or temporal barriers. The success of online learning is dependent on students' academic self-efficacy, which depends on their educators' preparedness and the quality of the learning materials (Welter et al., 2022). Self-efficacy is considered a critical factor in learning since it determines a student's belief in their own abilities (Bandura, 1986). Pham et al. (2021) developed a framework for understanding the variables affecting the effectiveness of online learning, which include proactiveness, self-study ability, perceived efficacy, the characteristics or complexity of the course, faculty capacity, and usability content.

It has been reported that the online course completion rate is 12.6%; however, based on a study of 221 Massive Open Online Courses, this rate can vary dramatically from 0.7% to 52.1%. The concept of the efficacy of online courses has provided background information to justify this variation in completion rate (MOOCs; Jordan, 2015). At the same time, face-to-face learning offers more monitoring options for students since the ability of administrators and instructors to oversee students increases learning module completion rates. Basar et al. (2018) noted that while students who attend face-to-face learning options interact with their peers, engage in in-person group exercises, and see their instructors regularly, students taking online courses must practice effective time management, be more motivated, and assume complete accountability for their

academic outcomes. This difference in the learning models' characteristics creates a difference in completion rates. The need for understanding approaches to online course completion has led to the study of the self-efficacy concept to address the psychological factors needed to increase learners' commitments, attitudes, competencies, and motivations (Hayat et al., 2020).

The ability of online learning to provide accessible materials without geographical limitations or time constraints has enabled individuals to engage in flexible learning hours and self-directed learning (SDL). Self-directed learning allows individual learners to become empowered to take more responsibility for different learning decisions and to transfer study skills and knowledge from one situation to another. Self-directed learning does not necessarily occur in isolation but does occur on a personal level since it involves activities and resources such as participation in study groups, self-guided reading, electronic dialogues, reflective writing activities, and internships. Theories of self-directed learning, such as Knowles's theory, have enabled an understanding of the characteristics needed to engage in self-directed learning and any other factors that must be considered to facilitate the process (Manning, 2007).

Numerous studies of online self-directed learning readiness (OSDLR) have suggested that OSDL success depends on self-efficacy, which is characterized by a student's attitude, selfdetermination, and motivation toward learning (Khodaei et al., 2022; Panisoara et al., 2020). Psychological factors such as anxiety, stress, and fear negatively affect students' self-efficacy, directly resulting in poor academic performance (Yilmaz, 2017). For a better understanding of OSDLR, several studies have provided an online learning readiness survey as a platform for measuring constructs such as self-efficacy, learner characteristics, online skills, independent and dependent learning preferences, time management and self-discipline, communication, and social

competencies (Martin et al., 2020; Wei & Chou, 2020). The survey instruments from these studies have aided in understanding the integrated competencies of self-directed learning, confidence, technical awareness, and communication.

Online Statistics Learning

Learning statistics online could have both advantages and disadvantages. For instance, students' opportunities to ask questions about a specific aspect of the course, a test, or calculations could be more limited. Al-Asfour (2012) recognized that while students were satisfied with online statistics courses overall, and they rated the assessment of their productivity and online instructions higher, they gave communication a lower rating. A study conducted by Blackburn (2015) revealed that online statistics courses were accepted with enthusiasm by students, who admitted that they were more interesting due to the presence of quizzes. Moreover, the online course decreased their statistics anxiety, taught them some interesting concepts, and led to positive feedback on the course overall. However, the same study showed that communication with instructors was rated lower than expected, and it was recognized that students benefit from previous knowledge of statistics before enrolling in the online course (Blackburn, 2015). Earlier research by DeVaney (2007) compared the attitudes, anxiety, and satisfaction levels of students enrolled in online statistics courses and traditional classrooms. They found that students' levels of statistics anxiety were significantly higher at the beginning of the online courses compared to the traditional classroom, yet at the end of the course, the level of statistics anxiety was lower in the online courses compared to the traditional classroom. Moreover, in some instances, anxiety increased among those learning in the traditional classroom, which affected overall satisfaction with the course (DeVaney, 2007). It is important to

acknowledge, however, that each study analyzed a specific online or traditional statistics course, and the design and content of online courses can vary dramatically.

The success of online statistics learning could depend on students' attributes as well as factors related to the design and content of the course. For example, Dunn (2014) tested the impact of academic self-regulation, intrinsic motivation, and statistics anxiety on procrastination in online statistics. The findings revealed that in an online course, academic self-regulation and intrinsic motivation are critical to reducing passive procrastination and attaining success in online statistics learning. This study emphasized the importance of intrinsic motivation (mastery approach goal orientation) in decreasing procrastination and achieving success in an online statistics course. At the same time, Mills and Raju (2011) reported that, in most cases, students prefer a face-to-face statistics course over an online course primarily due to usability concerns, a lack of communication with the instructor, and problems with design, which affected their motivation to learn.

However, it is also critical to mention that this study analyzed online courses created more than ten years ago and did not consider the changes made over the years in terms of software and hardware progress. Larwin and Larwin (2011) showed that the positive attitude toward and satisfaction with online statistics courses had increased exponentially with the development and improvement of technologies, which provided more convenient, effective, and usable course designs. Further, since the COVID-19 global pandemic started, the amount of online learning has proliferated. Koehler et al. (2018) analyzed the impact of interactive instruments on learning statistics online, including student satisfaction, and revealed that learners were particularly satisfied with the possibility of having control over their learning and accessing education remotely.

Problem Statement

The COVID-19 crisis has led to drastic changes in the education system. Students were forced to shift from physical to online classes in response to government restrictions on face-to-face interactions. These restrictions were considered critical to preventing the spread of a global respiratory disease. Aguilera-Hermida (2020) explains that while numerous challenges accompanied these changes, they created an awareness of the role and advantages of engaging in online learning. Therefore, even after the risk of viral spread was reduced, the popularity of online learning remained high. Bashir et al. (2021) reported that Aston University adopted a hybrid mode of course delivery (combining online and face-to-face courses) as a post-COVID solution that recognized the popularity of online learning. A survey cited by Kelly (2021) asserted that 73% of students preferred studying fully online after the pandemic; however, only about 53% of faculty preferred teaching online.

Statistics courses have also been transferred online, with studies such as Ritzhaupt et al. (2020) revealing that students enrolled in the coursework experienced less anxiety, making the module preferable to in-person learning. The popularity of online statistics courses post-pandemic has created a demand for understanding this mode of learning and how learners and educators can organize their courses effectively. Further, demand for understanding the learning module has been created by studies such as Figueroa-Cañas and Sancho-Vinuesa (2020), which reported that the dropout rate of online statistics courses is higher than that of traditional statistics courses. This is a significant concern because, even though most learners prefer online learning, this same online model has seen higher dropout rates. However, Knowles' self-directed learning theory has encouraged scholars to focus on self-efficacy to improve online statistics courses (Manning, 2007). The current study sought to elucidate our understanding of the existing

gap in the literature regarding the effects of self-efficacy, self-directed learning readiness (SDLR), and goal orientation on learning outcomes for an online statistics course.

Research Questions

The present study investigated the following research questions:

1. What is the extent of the relationship between online self-directed learning readiness and grades mediated by self-efficacy to learn statistics and each construct of achievement goal orientations in an online statistic learning environment?

1.1 What is the extent of the relationship between online self-directed learning readiness and grades in an online statistic learning environment?

1.2 What is the extent of the relationship between online self-directed learning readiness and grades mediated by self-efficacy to learn statistics in an online statistic learning environment?

1.3 What is the extent of the relationship between online self-directed learning readiness and grades mediated by each construct of achievement goal orientations (MAP, MAV, PAP, PAV) in an online statistic learning environment?

2. What is the extent of the relationship between online self-directed learning readiness and course satisfaction mediated by self-efficacy to learn statistics and each construct of achievement goal orientations in an online statistic learning environment?

2.1 What is the extent of the relationship between online self-directed learning readiness and course satisfaction in an online statistic learning environment?

2.2 What is the extent of the relationship between online self-directed learning readiness and course satisfaction mediated by self-efficacy to learn statistics in an online statistic learning environment?

2.3 What is the extent of the relationship between online self-directed learning readiness and course satisfaction mediated by each construct of achievement goal orientations in an online statistic learning environment?

Significance of the Study

The current study seeks to investigate the existing gap in our knowledge regarding the effects of self-efficacy, self-directed learning readiness (SDLR), and goal orientation on learning outcomes for learning statistics online. This investigation is especially relevant due to the increased demand for online courses after the pandemic. The study aims to create awareness and to inform learners and educators who engage in online statistics courses about how they can rely on self-efficacy and achievement goal orientations to improve their grade outcomes. Students, institutions, and tutors participating in online statistics courses will also understand the impact of self-directed learning readiness on online statistics course satisfaction and grade outcomes. The findings of the present study can be used to improve the dynamics of online statistics course grade outcomes and satisfaction from a psychological and academic perspective. As such, the present study has the potential to make important contributions to the body of academic literature, especially in the field of education.

Limitations of Study

This study has several limitations. First, students may not completely understand their achievement goal orientations, and some of the participants may adopt two or more goal orientations. Second, students were asked to report their grades from one of the courses they were enrolled in, but some of the participants who were enrolled in more than one statistics course could have reported their best grades regardless of their goal orientations and self-directed

learning readiness. This also applies to course satisfaction. Third, the present study includes many variables, which increases the potential for introducing confounding variables that this study might fail to control. The presence of too many variables can lead to overfitting, which results in poor predictions using new data. As such, the study results could have a level of ambiguity due to overfitting. In addition, this study used self-reported questionnaires, which increases the possibility of participant bias. At the same time, the variables measured by this study are somewhat subjective in that they require feedback from students (participants') regarding goal orientation, self-efficacy, and SDLR. Lastly, this study focuses exclusively on online statistics learning from a large southeastern research institution; therefore, the generalizability of the results to other subjects and settings might be problematic.

However, these limitations do not seriously threaten validity or reliability.

Definitions of Terms

Achievement goal orientation (AGO): This is defined as part of the need to master goals. In a recent paper, it was classified as "an individual's perception of the purpose or meaning of engaging in an achievement activity, such as to demonstrate competency or to develop competency (Elliot et al., 1999; Lin, Chen and Zhang, 2021)."

Course satisfaction: In the context of this dissertation, course satisfaction refers to the level to which students are satisfied with their online statistics course.

Grade: A student assessment tool employed by a specific educational facility (Richardson & Newby, 2006).

Mastery approach goal orientation (MAP): Students who focus on developing competence-based learning, expanding their understanding of a topic, and improving

their performance display a MAP goal orientation (Elliot et al., 1999; Mascret et al., 2017).

Mastery avoidance goal orientation (MAV): Students who focus on not losing knowledge, skills, or competence display a MAV goal orientation (Elliot & McGregor, 2001).

Online Learning Readiness Scale (OLRS): The OLRS is a scale measuring SDLR based on categories such as self-directed learning, motivation for learning, computer self-efficacy, learner control, and online communication self-efficacy (Hung et al., 2010).

Online self-directed learning readiness (OSDLR): OSDLR indicates that a learner has the attitudes, capabilities, and personality characteristics necessary for online self-directed learning (Sumuer, 2018).

Performance approach achievement goal orientation (PAP): Learners who demonstrate their competence by trying to outperform others demonstrate a PAP goal orientation (Wang et al., 2021).

Performance avoidance achievement goal orientation (PAV): Learners who focus on avoiding looking incompetent, making an error, or being outperformed by others display a PAV goal orientation (Partridge et al., 2014).

Self-efficacy: This describes people's judgments of their own capabilities to organize and execute courses of action required to attain designated types of performances (Bandura, 1986).

Self-efficacy in learning statistics (SELS): SELS refers to an individual's confidence in their ability to effectively learn the skills needed for their statistics course. It is rooted in social cognitive theory (Polm, 2016).

Organization of Study

The current study is organized into five sections: Introduction, Literature Review, Methodology, Results, and Discussion. Chapter 1 presents the introduction, which describes the research purpose, problem statement, and research question, the significance of the study, limitations of the study, the definition of terms, and the organization of the study. Chapter 2, the literature review, describes previous studies covering the research topic. This chapter not only brings together existing knowledge about self-directed learning, self-efficacy, 2x2 achievement goal orientation, and learning outcomes but also identifies an existing gap in the literature regarding the impact of the self-directed learning framework in an online learning setting and how this relationship is mediated by self-efficacy and 2x2 achievement goal orientations, which aid in establishing the unique contribution of the current study. Chapter 3 discusses the problem statement, research questions, research instruments, and participants' data collection and analysis procedures. Chapter 4 presents the demographic information of participants and survey results. The survey results will address how online self-directed learning readiness impacts learning outcomes, which are mediated by self-efficacy to learn statistics and achievement goal orientations. Finally, Chapter 5 provides implications for theory, practice, overall implications, and suggestions for future research.

Chapter 2 - Review of Literature

Overview

A review of the literature provides theoretical frameworks for the present study by discussing theories of achievement, goal orientation, self-directed learning (SDL), and self-efficacy. Self-directed learning, achievement goal orientation, and social cognitive theories are first discussed in Chapter 2. Specifically, the conceptual framework for this study includes self-directed learning readiness, online self-directed learning readiness, self-efficacy to learn statistics, and learning outcomes, such as cognitive and affective learning outcomes, as well as the achievement goal orientation theories of motivation. Following this conceptual framework, the literature review will focus on the relationship between online learning readiness, a 2x2 framework of achievement goal orientations, self-efficacy to learn statistics and learning outcomes such as academic achievement and course satisfaction.

Problem Statement

The COVID-19 global pandemic has led to drastic changes in educational systems around the world. Students were forced to shift from physical to online classes in response to government restrictions on face-to-face instruction. These restrictions were considered critical to preventing the spread of COVID-19. Aguilera-Hermida (2020) explained that while numerous challenges accompanied these changes, they also created an awareness of the role and advantages of engaging in online learning. Therefore, even after the risk of virus spread was reduced, the popularity of online learning remained high. Bashir et al. (2021) reported that Aston University adopted a hybrid mode of course delivery (combining online and face-to-face courses) as a post-COVID solution that highlighted the popularity of online learning among students. A survey cited by Kelly (2021) established that 73% of students preferred studying fully online after the pandemic; however, only about 53% of faculty preferred teaching online.

In recent years, many statistics courses have also been transferred online, with studies such as that conducted by Ritzhaupt et al. (2020) revealing that students enrolled in online coursework experience less anxiety, making online instruction preferable to in-person learning. The popularity of online statistics courses post-pandemic has created a demand for understanding this modality of learning and how learners and educators can organize their courses effectively. Further demand for understanding the learning modality has been created by studies such as those by Figueroa-Cañas and Sancho-Vinuesa (2020), which have reported that the dropout rate of online statistics courses is higher than that of traditional statistics courses. This is a significant concern because, even though most learners prefer online learning theory has encouraged scholars to focus on self-efficacy to improve online statistics courses (Manning, 2007). Thus, the present study seeks to explore the existing gap in the literature regarding the effects of self-efficacy, self-directed learning readiness (SDLR), and goal orientation on learning outcomes for learning statistics online.

Research Questions

1. What is the extent of the relationship between online self-directed learning readiness and grades mediated by self-efficacy to learn statistics and each construct of achievement goal orientations in an online statistic learning environment?

1.1 What is the extent of the relationship between online self-directed learning readiness and grades in an online statistic learning environment?

1.2 What is the extent of the relationship between online self-directed learning readiness and grades mediated by self-efficacy to learn statistics in an online statistic learning environment?

1.3 What is the extent of the relationship between online self-directed learning readiness and grades mediated by each construct of achievement goal orientations (MAP, MAV, PAP, PAV) in an online statistic learning environment?

2. What is the extent of the relationship between online self-directed learning readiness and course satisfaction mediated by self-efficacy to learn statistics and each construct of achievement goal orientations in an online statistic learning environment?

2.1 What is the extent of the relationship between online self-directed learning readiness and course satisfaction in an online statistic learning environment?

2.2 What is the extent of the relationship between online self-directed learning readiness and course satisfaction mediated by self-efficacy to learn statistics in an online statistic learning environment?

2.3 What is the extent of the relationship between online self-directed learning readiness and course satisfaction mediated by each construct of achievement goal orientations in an online statistic learning environment?

Self-Directed Learning Theory

Self-directed learning (SDL) is a theoretical framework currently used by pedagogical experts and teachers to enhance self-efficacy in learners, promote transformational learning, and utilize components of social action and justice learning. Knowles (1975) defined SDL as a process by which individuals take initiative, through the assistance of others or on their own, to diagnose learning gaps and needs, formulate goals, identify resources, and implement strategies

for learning. They also assess learning outcomes independently. Self-directed learning (SDL) allows individual learners to become empowered to take more responsibility for different learning decisions and to transfer study skills and knowledge from one situation to another (Alfaifi, 2016).

Knowles popularized the concept of self-directed learning in the US, while Tough, another scholar, embedded the term in Canadian discourse. In 1979, Tough published his book, *The Adult's Learning Projects*, an analysis of self-directed teaching activities. The term andragogy, coined by Knowles, with corresponding adult instructional processes, became popular in North America at the same time. Here, the principle of autonomy, which largely drives adult learning, is applied to more generalized notions of self-directed learning.

With the use of computer-assisted learning, many distance education efforts are using technology to facilitate self-directed learning, which requires new research and understanding (Hiemstra, 1994). In other words, it took the critical work of scholars like Gibbs (1988), Candy (1991), and Knowles (1997) to develop a framework for today's 21st-century learning institutions. Brookfield (1993) disagrees with Candy (1991), arguing that democratic educational systems should be guided by SDL rather than institutions and governments. Some proponents of nationally oriented curricula consider SDL an abomination, while education better prepares students for the dominant culture. Today's schools are stronger, more diverse, and more multicultural and multiethnic than in the past. SDL increases cultural competence and awareness, according to research (Aldemir et al., 2022). Technology makes individuals more interconnected, making meaning-sharing more crucial. Building bridges of understanding requires skills and knowledge. The literature supports Brookfield's (1993) claim that, rather than contributing to

cultural hegemony, an SDL pedagogical method assists students in a rapidly globalizing, technologizing, and diversifying environment.

Brockett and Hiemstra shaped SDL. The researchers analyzed learners' requirements, secured relevant resources, executed learning activities, and evaluated outcomes. Self-directed learning preparedness, self-concept, experience, and learning styles can influence SDL adoption. "Learner self-direction" is related to individual intentionality (Hiemstra, 1994). Hence, learners become naturally motivated if they thrive under the correct conditions and have enough confidence and self-concept to advance, interact with content, build knowledge, and use their abilities. Furthermore, Brockett and Hiemstra (1991) contributed to the discussion with their personal responsibility orientation model (PRO). The authors define PRO as taking ownership of one's actions and thoughts. The only way to take a proactive approach to learning is to accept responsibility for one's own education, and this comes from humanism, including the idea that individuals are independent, autonomous, and responsible for their own growth. In other words, it is virtually impossible for one to be independent if they are not accountable for their own actions, consequences, and choices. On the other hand, Brockett and Hiemstra (1991) believed that adult education instructors should boost student capability.

The PRO approach fits 21st-century needs, where information flows rapidly and is absorbed in large quantities, and knowledge management is crucial. Razzaq et al. (2019) observe that the public sector has recently prioritized knowledge management. As the world moves towards a service-based economy, managing knowledge is becoming a skill many professionals need. SDL may teach students early on how to manage knowledge for the economy and their jobs. Brockett and Hiemstra's (1991) PRO approach emphasizes accountability and responsibility, which are important for knowledge management.

Grow (1994) proposed the Staged Self-Directed Learning (SSDL) model, which gradually reduces an instructor's power over pupils. In the beginning, the instructor lectures, coaches, and gives criticism. At the second, "moderate" level, the pupil is independent, but still wants guidance. The teacher inspires conversation, goal setting, and lectures during this stage. At the third, "intermediate," stage, students engage in active learning. The teacher facilitates thought-inspiring dialogues by setting protocols and treating pupils equally. During this "selfdirected" stage, the teacher acts as a consultant or delegate. Grow's model better defines SDL by incorporating a gradated system that assesses pupils' independence from their teachers.

Recent research confirms Knowles' fundamental work by demonstrating SDL's influence on learner autonomy, self-efficacy, and pleasant sentiments from the two (Terry, 2006). As they advance, learners prefer independent learning. According to Tekkol and Demirel (2008), selfdirected learning requires planning, continuation, and assessment. Self-directed learning also requires independence (Hatcher, 1997; Nasri et al., 2020; Terry, 2006). According to Schweder and Raufelder (2022), positive emotions boost learning, and self-directed learning empowers students. Self-efficacy and autonomy create such good feelings and associations (Finnegan, 2022; Lemmetty et al., 2021). Nasri et al. (2020) also assert that SDL improves student competence and accountability.

Self-directed learning works well in a globalizing world, especially considering the lasting effects of COVID-19 (Liu et al., 2022). Self-directed learners engage in planning to learn, identify needs, and choose practical solutions (Gerard et al., 2022; Lai et al., 2022; Sahoo, 2016). Research supports learners supporting their own transformative learning. Internet communications technology (ICT) promotes students' participation in learning and is essential to SDL (Labonté & Smith, 2022), as portals can help students find information.

SDL andragogy is also significant. Adult learning takes place online, offline, or in hybrid adult-based learning programs. According to Pappas (2013), andragogy requires self-concept, adult learner experience, preparation, motivation, and orientation toward learning. ICT-supported learning settings combine online and remote learning with SDL, and ICT-enabled instruction aids SDL (Labonté & Smith, 2022).

In a world of instantaneous information, Guglielmino (2013) stressed the need for individual learning. Google CEO Eric Schmidt (2011) stated that at present, humans generate as much knowledge in two days as they did from the beginning of civilization through 2003. SDL helps students consume, control, and organize larger data sets, especially for classroom and professional use. Wang et al. (2021) propose that COVID-19 has forced migration to online learning, where the Internet collects and distributes vast volumes of data.

Frambach et al. (2012) emphasize that SDL can encourage learning and intellectual curiosity, but it also has drawbacks. SDL capacities and cultural and educational backgrounds were examined by medical school collaborators, who highlighted uncertainty regarding the level of student independence needed for autonomous learning, reliance on hierarchal pedagogical frameworks, and overemphasis on traditional learning modes and orientations as difficulties of the approach. Teacher-centered classrooms and pressure to get good grades were further issues faced by students in secondary schools.

Douglas (2010) assessed participants' perceptions of SDL adoption hurdles through a quantitative correlational study that identified dispositional, situational, and institutional limitations to self-directed learning. The researcher found that race, gender, class, and technology can be barriers to SDL. These can be situational, belief-based, or attitudinal.

Studies by Kohan et al. (2017) are consistent with the existing literature on SDL. While du Toit-Brits (2020) cites SDL's increasing efficacy in the 21st century, they also note its potential disadvantages and barriers. Kohan et al. (2017) studied barriers to SDL within virtual environments. Using semi-structured interviews and purposive sampling, the authors interviewed 23 postgraduate students in an Iranian medical sciences program. They then used inductive content analysis to interpret the data. Some of the identified barriers to participation in SDL included cognitive barriers, such as an overload of information and wandering minds, and barriers in communication (such as notetaking skills). Information overload was a common problem cited by the students. Participants were anxious about mismanaging large swathes of data within the eLearning context and had a difficult time sorting out which data was appropriate, feasible, and correct.

Other commonly reported issues included inadequate writing skills (i.e., preference for oral communication), time management, and role ambiguity. Role ambiguity refers to different expectations from students than those that they otherwise experience during the learning process. One post-secondary participant from Kohan et al.'s (2017) cohort suggested that a lack of face-to-face time created situations where instructors could not identify whether students had mastered important content. A common criticism among instructors using SDL was that they perceived SDL as a way of having students work completely on their own without any virtual assistance (Kohan et al., 2017). In other cases, students felt that they had more effective oral skills than writing skills but were required to submit responses to discussion questions online in text form. Students complained about time management problems, as they perceived online SDL learning to be more challenging to manage in terms of meeting deadlines and completing assignments.

It is important to remember that while SDL occurs within the context of the "self" (i.e., the individual), humans remain embedded in a cultural context, which may influence how they perceive the world around them. Thus, SDL does not and cannot occur in a vacuum. Even the most critically reflective educators cannot escape the determinants and influences of their own lives. SDL educators should reflect on the ongoing processes affecting themselves and their students while teaching. SDL can be a good way to teach students how to handle large amounts of information effectively. This type of pedagogy will become critical for fostering problem-solving and critical thinking skills in students in the 21st century.

Self-Directed Learning Readiness

Self-directed learning readiness (SDLR) has grown in popularity in recent years, especially in the field of higher education. Self-directed learners decide what they need to learn, according to Knowles (1975). They also plan where and how they will learn, which strategies will work best, and how they will measure their success. Chu and Tsai (2009) state that selfdirected learning requires self-discipline, independence, autonomy, efficient organization, clear communication, openness to constructive criticism, introspection, and self-evaluation. The learner's self-directed learning readiness depends on their attitudes, abilities, and personality traits (Wiley, 1983).

SDLR for independent study requires many skills and traits. Students must assess their family, social, and academic situations. Through SDLR, adults are taught lifelong learning habits, techniques, and skills rather than passing the standardized tests. So, people must organize their issues and develop skills to accomplish so (Brown & Palincsar, 1989; Caffarella, 1993).

Self-directed students are more aware of their self-monitoring obligations, as they seek to enrich their education. These learners want to try new things. Self-directed adult learners

perceive problems as challenges and are willing to learn and change; thus, they possess motivation, persistence, independence, self-discipline, self-confidence, and a goal-oriented perspective (Abdullah, 2007). Fisher et al. (2001) defined self-directed learning preparation as the extent to which a student takes responsibility for their education and accepts the independence that comes with focusing on relevant issues. The learner's mindset, approach, and skills determine their freedom. SDLR assumes; first, adults can self-direct, and SDL readiness can be considered a spectrum (Leach, 2000). Second, self-direction requires practice (Candy, 1991). The final premise is that SDL skills learned in one setting can be used in others. The biggest issue with defining SDL readiness is that high levels of readiness may not transfer to different circumstances and contexts (Fisher et al., 2001).

As mentioned, SDL preparedness is individualized and represents a continuum. Students who are unprepared show anxiety equivalent to that of those who are well prepared and put in a structured and instructor-guided environment.

Online Self-Directed Learning Readiness

In recent years, we have seen a rise in the idea of SDLR, especially in the context of higher education. For SDLR to be effective in independent study, a wide range of abilities and dispositions are needed. Therefore, students need to take stock of their circumstances, including their social support, study habits, and family life. There is less of a focus on standardized testing and more on developing the habits, strategies, and abilities necessary for lifelong learning in the adult population. As a result, people are required to arrange their personal challenges and cultivate traits that will serve them well in doing so (Brown & Palincsar, 1989; Caffarella, 1993). In this context, self-monitoring is an area in which self-directed students excel because they

show more awareness of their responsibilities in this area as they strive to give their education a deeper purpose.

Self-directed learners develop an appetite for knowledge and a willingness to broaden their horizons by testing new activities, ideas, and abilities. Therefore, self-directed adult learning necessitates elevated motivation, tenacity, independence, self-discipline, selfconfidence, and the attainment of a goal-oriented mindset, since independent learners see issues as challenges and are willing to learn and change (Abdullah, 2007).

As proposed by Fisher et al. (2001), the concept of self-directed learning preparedness investigates the extent to which the self-directed learner takes responsibility for their education and accepts the independence that comes with focusing on topics that are meaningful to them. The level of freedom a learner can achieve depends on their disposition, outlook, and skill set. There are assumptions regarding SDLR. First, adults have an intrinsic capacity for self-direction, and individuals' levels of SDL preparedness may be seen as falling somewhere along a spectrum (Leach, 2000). Secondly, skills in self-direction need time and effort to develop. Learning and practicing autonomous conduct is the best approach to comprehending and displaying selfdirected behavior (Candy, 1991). The last assumption is that SDL skills acquired in one setting may be transferred to similar situations in another. High degrees of SDL readiness may not always transfer to new situations and contexts, which is perhaps the biggest problem when attempting to define SDL readiness (Fisher et al., 2001).

SDL readiness is seen as very personalized and representative across the spectrum, as was previously stated. Students who are not prepared for SDL and are then given an SDL assignment exhibit significant levels of anxiety that are comparable to those shown by students who are well

prepared for SDL and are placed in surroundings with a great deal of structure and instructor guidance.

Self-Efficacy Theory

Self-efficacy is the primary element of Bandura's social-cognitive theory and has had a profound impact on the study of motivation and achievement in academic settings (Bryant, 2017). Self-efficacy describes an individual's confidence in their ability to control their motivation, behavior, and external environment, which manifests in the behaviors needed to achieve a particular performance and develops and forms as early as childhood (Bandura, 1986).

Self-efficacy is a domain-specific belief in one's ability to successfully perform a task, which influences engagement in and successful completion of the task (Bruning et al., 2013; Klassen, 2002; Pajares, 2003). Chemers et al (2001) defined Academic self-efficacy as "students" confidence in mastering academic subjects" (p. 56).

From the student's perspective, self-efficacy can be defined as a personal judgment regarding an individual's capability to reshape their behavior and accomplish academic goals. Gopal et al. (2018) defined self-efficacy theory as the cooperative interaction between the nature of the task assigned to the student and their thoughts, which determines their behavior and success rate. Self-efficacy is driven by four sources: physiological states, social persuasion, and vicarious and mastery experiences (Bandura, 1994, 1997). Mastery experiences are the major contributor to self-efficacy since they operate within a student's experiences on a subject matter. At the same time, vicarious experiences are based on what students observe and perceive when their peers perform a particular task, which aids them in manipulating their experiences through observation. A student's sense of self-efficacy is more positively impacted by others who experiences uccess if common characteristics are shared, such as age, gender, and perceived

similar abilities (Bryant, 2017). On the contrary, physiological states such as anxiety, mood swings, and stress control a student's capabilities in terms of success or failure. In such situations, a student's self-efficacy may decrease after experiencing gloomy emotions and uncertainty regarding their abilities.

Pajares and Schunk (2002) insist that self-efficacy influences learners in very different ways and is pivotal in academic self-motivation (Bandura, Martinez-Pons, & Zimmerman 1992). A meta-analysis of empirical studies conducted over the last 20 years discovered that academic self-efficacy was the single strongest predictor of students' academic achievement and performance among nine commonly researched psychosocial constructs (Artino, 2012; Bryant, 2017).

Considering the above details, students at higher education levels tend to demonstrate higher levels of self-confidence and belief in their ability to perform. In other words, older students at higher learning levels have higher self-efficacy compared to younger students at lower learning levels. The study by Gore (2006) indicates that self-efficacy is a predictor of college students' academic performance and persistence. The author attained such findings by examining first-year college students, their ACT scores, and a self-reported self-efficacy survey. Nonetheless, the results may not predict college success and could partially depend on "(a) when self-efficacy beliefs are measured, (b) what aspect of self-efficacy is being measured, and (c) what college outcome one wishes to predict" (Gore, 2006, p. 112). Gore's (2006) findings also highlight the need to give students feedback on their performance, both social and academic, for them to assess their ability to achieve academic goals. The findings by Uchida, Michael, and Mori (2018) support such a perspective by indicating that giving feedback to college students enhances their self-efficacy, as they believed in their ability to succeed in subsequent tasks.

Moreover, Schunk (1991) underscored the role self-efficacy's in allowing students to recognize the need to achieve academic goals and superior grades, surpass other students, embrace new experiences, and diligently prove their intelligence through schoolwork. Students with low self-efficacy tend to negate the influence of intelligence on academic improvement. They feel unable to succeed, which undermines their focus on goals, mastery, and performance. Bandura (1977) hypothesized that individuals develop self-efficacy by interpreting the information received from the environment. This perspective aligns with the four sources of selfefficacy: mastery experiences, social experiences, vicarious experiences, and physiological experiences. Accordingly, interpreting one's own previous performance and mastery experience is a vital source of information (Klassen, 2004; Pajares et al., 2007).

Furthermore, students of various ages have shown a positive relationship between selfefficacy and higher achievement levels. The same trend is evident between learning and adaptive academic outcomes, including improved effort and persistence with difficult tasks in experimental and correlational studies (Bandura, 1997; Linnenbrink & Pintrich, 2002; Pintrich & Schunk, 2002). Such studies indicate that students with high self-efficacy tend to work harder, persist more, and achieve better results compared to those with low self-efficacy. These explored areas imply that students with high expectations and those seeking challenging tasks tend to have stronger academic persistence compared to those with low expectations who tend to avoid or give up on their pursuit of learning (Brophy, 2004; Cleary & Chen 2009; Zeldin & Pajares, 2000). Similarly, students with high levels of self-efficacy are likely to develop challenging goals and draw on different strategies to accomplish them (Kizilgunes et al., 2009, as cited in Arquero et al., 2015). Walker et al. (2006) supports such findings by indicating a

meaningful positive relationship between students' self-efficacy and cognitive learning outcomes.

Previous research has shown that students with higher self-efficacy were more eager to engage in difficult tasks, persist longer, and work harder (Bruning & Horn, 2000; Zimmerman, 2000;). Students with higher self-efficacy are able to become more resistant to the negative impacts of failure (Bandura, 1986). Williams and Williams (2010) attest that while students with high self-efficacy feel motivated to approach complicated tasks, students with low self-efficacy develop anxiety and nervousness. Higher self-efficacy may lead students to believe that they can regulate their learning and become inclined to set personal goals. For instance, efficacious students will select rigorous coursework and have the confidence to complete challenging material. Students with low self-efficacy may even perceive a task as more difficult than it really is and give up early.

Self-efficacy is a critical predictor of students' academic success. According to Clerge (2019), self-efficacy entails a person's perceived ability to perform the tasks necessary to attain their goals. As noted above, the four sources of self-efficacy, mastery experiences, verbal persuasion, vicarious experiences, and affective and physiological states, are essential in shaping various activities and underlying academic success. In their study, Medaille et al. (2022) indicated that some academic tasks, such as thesis writing, require students to demonstrate a high level of self-efficacy that is aligned with their belief in their ability to accomplish results. Educators should create adequate opportunities for students to develop self-efficacy for optimal academic achievement. The study by Bjørnebekk et al. (2013) covers the impact of self-efficacy, motives, and achievement goals on academic achievement in post-secondary education.
The researchers analyzed factors underlying achievement motives, self-efficacy, and achievement goals to improve student performance. The findings indicate that students developed self-efficacy by staying in programs for longer. However, their fear of failure influenced their performance. Students' perceptions of their self-efficacy shape the anticipatory scenarios they create and reiterate (Norton, 2013). Consequently, the findings indicate a negative correlation between self-efficacy and the fear of failing examinations. For instance, such fear hindered science students from achieving optimal performance (Bjørnebekk et al., 2013).

Self-efficacy influences individuals' well-being, thus enhancing performance. The study by Gutiérrez and Tomás (2019) emphasizes the role of self-efficacy and school engagement in enhancing autonomy and academic achievement. Self-efficacy also incorporates emotions, which influence learners' motivation, cognitive sources, satisfaction, mental health, and learning strategies, which in turn shape academic achievement (Hayat et al., 2020). The findings obtained by Thompson and Verdino (2019) support these viewpoints by indicating that enhancing selfefficacy among community college students is essential to addressing the achievement gap by enhancing motivation levels and persistence.

Additionally, improved academic achievement is vital to enhancing students' motivation. Motivation encompasses selecting, activating, and directing behavior toward a specific goal (Landry, 2003). A study by Bjornebekk et al. (2013) indicated that as learners transition from one level to the next, they tend to focus more on academic achievement. Study outcomes by Bong et al. (2012) support such a perspective by indicating that middle school students demonstrated a higher confidence level in mathematics compared to elementary students. Yokoyama (2019) supports the explored findings by indicating that students with high self-efficacy are highly motivated to learn, leading to higher academic achievement. These details demonstrate the need

for educators to foster students' motivation by emphasizing the need for self-efficacy, selfdetermination, self-esteem, and academic identity, among other positive attributes (Gannouni & Ramboarison-Lalao, 2018).

Doménech-Betoret et al. (2017) postulate the relationship between self-efficacy, course satisfaction, and expectancy-value beliefs. According to their study, process expectations relate to students' feelings during their interactions with their teacher and are likely to play a salient role in describing learners' satisfaction levels. However, this process can be influenced by the student's self-efficacy or expected performance levels. In other words, students with robust self-efficacy beliefs are more likely to visualize success scenarios that guide performance and provide supportive resources. Consequentially, these students tend to experience more satisfaction with the learning process. Doménech-Betoret et al. (2014) focused on the mediator role played by students' beliefs, such as self-concept, self-esteem, self-efficacy, and outcomes. The study shows a positive relationship between self-efficacy and achievement expectations, cost (expected dedication), and enjoyable learning expectations, which in turn led to achievement expectations that eliminated avoidance strategies.

Self-Efficacy to Learn Statistics

Widyastuti et al. (2019) define statistics as a branch of mathematics that is used as a tool to help humans solve their problems mathematically. It can also be defined as the science that deals with presenting, collecting, analyzing, processing, and concluding data when information is available at both variable and limited levels. Beyth-Marom et al. (2008) claim that numerous studies aim to improve statistical education and practice. The concept of statistical cognition (a notion that learning simply involves developing and testing hypotheses) has aided in improving statistical education. The science of statistics has been attributed to the most normative field of

statistical cognition, which includes simple rules such as the conjunction rule of probability, models, theorems, and laws. Statistics can be understood from a descriptive, normative, or perspective approach, with the normative approach proven to be the best. In this context, selfefficacy will be considered an approach to improving statistical education through improved reasoning and information retention. The self-efficacy concept improves learning by doing, authentic learning, and situated cognition.

Widyastuti et al. (2019) established three facets of statistical cognition that aid in understanding the mediating role of self-efficacy in statistics. Normative facets represent statistical techniques that can be applied correctly within a given situation and integrate the full body of knowledge of mathematical statistics. On the other hand, descriptive facets integrate the knowledge of how people think regarding statistical concepts, misconceptions, and biases in the field and the messages they acquire when interpreting statistical information (Iwuanyanwu, 2022). At the same time, perspective statistics rely on knowledge of how to achieve successful statistical education and information.

Self-efficacy, which concerns an individual's performance capability, is dependent on judgment capability, which makes it a critical factor in statistics (Iwuanyanwu, 2022). Selfefficacy influences how individuals think, motivate themselves, and feel toward a subject by promoting fluency, ease, and confidence. Welter et al. (2022) established a correlation between the three facets of statistical cognition (prescriptive, normative, and descriptive), with the implication that students must be aware of these processes of acquiring knowledge when they attempt to explain statistical concepts. At the same time, this correlation depends on a clear perception of the difference between values and variables, which affects the student's ability to

use symmetric variables and, for example, their ability to interpret the data and represent it on the 2x2 tables.

Yousafzai et al. (2020) used the cognitive inference model to understand how selfefficacy can be used to improve statistical performance. They found that the cognitive interference model affects students' statistics anxiety, and performance. Based on the cognitive inference model, if anxiety overloads a student's working memory during an exam, this might affect their ability to perform during the examination period. Bárkányi (2021) also established a direct relationship between reduced student performance, self-efficacy, and increased anxiety levels. According to Macher et al. (2015), a student's ability to prepare for an exam can cause anxiety, which is mainly triggered by procrastination during the preparation period.

At the same time, Feldhammer-Kahr et al. (2022) showed that maladaptive coping strategies in the study process were averse to using self-regulated learning strategies, which then triggered statistical anxiety, indirectly impacting the study process. According to the researcher, self-regulated learning includes forethought, performance, and self-reflection. Anxiety reduces students' self-efficacy at the forethought stage since such students have maladaptive beliefs regarding statistics, making them struggle to keep up with classwork as early as the beginning of classes. These students avoid completing their homework and use procrastination as an escape throughout the process. Furthermore, they fail to help themselves view class materials in a less threatening way in order to learn productively. Such students are also afraid of asking their tutors for help, thereby placing themselves in situations where they experience repeated failure cycles. Self-efficacy, therefore, becomes a critical component for students aiming to achieve success in statistics, since it helps them feel that they have what it takes to develop the skills needed to master the presented concept, even if it demands that they work through setbacks (Yousafzai et

al., 2020). Samuel and Warner (2021) define a self-efficacious statistics student as one with an analytical mindset who can cultivate the skills required to meet the given learning objective since they acknowledge that the required skills can be attained with effort. From the beginning, the student can set clear, realistic, and challenging goals attributed to the mastery of objectives in the statistics course.

Self-efficacy builds the skills needed to utilize underperformance feedback as a platform for identifying new strategies for overcoming failure. Hoegler and Nelson (2018) support the claim that students with self-efficacy beliefs have an additional 7% variability in their exams because they controlled the environment of success before sitting the test. Samuel and Warner (2021) aimed to study how to improve statistical performance by improving self-efficacy and reducing anxiety levels. Based on the research, self-efficacy was judged as a state of psychological arousal that enables people to measure their anxiety levels or other emotionally arousing situations. The detrimental effect of anxiety on student statistics performance demands self-efficacy models that can enable students to prepare for exams, modify positive beliefs surrounding their coursework, and eliminate any misguiding beliefs that might prevent them from seeking help by eliminating maladaptive approaches that might hinder the preparation process.

Statistical self-efficacy is domain-specific since it includes factors relating to application and reasoning, which arise from the quantitative nature of statistical calculations. Perepiczka et al.'s (2011) study on the relationship between statistics self-efficacy and statistics anxiety focused on environmental, dispositional, and situational variables. Dispositional antecedents are intrapersonal issues that bring students into the classroom. The researchers discovered that, at the

developmental stage, factors influencing students' perceptions of their abilities affected their concerns regarding evaluation and fear of failure, which triggered statistical anxiety.

Environmental antecedents involve interpersonal factors related to the classroom experience, including interactions between the lecturer and the student. Finally, situational variables are the factors that surround the students' previous experiences with statistics. Hoegler and Nelson's (2018) study showed that students who had failed mathematics courses had a negative perception of statistics.

Attitude is also a contributing factor in self-efficacy statistics. Perepiczka et al. (2011) found that attitude determined the knowledge gap and the need for students to fill the gap provided by statistics courses. The study also reviewed the impact of gender on statistics, where males were found to have a more positive attitude than their female counterparts towards self-efficacy in statistics. Students who viewed statistics as a road towards achieving their degree and career also had better attitudes than their counterparts, which had a positive impact on their performance.

Perceived social support establishes the basis of statistical self-efficacy since it forms a personal self-identity based on family, friends, and significant others. The perceived social support process creates a potential buffering effect on attitudes and anxieties (Althauser, 2015). Walker et al. (2017) showed that statistics attitudes are not unidimensional; instead, they integrate cognitive competence, affect, interest, effort, value, and difficulty variables. Students with self-efficacy possess the attitudes required to improve throughout the semester, thereby contributing to higher performances. Widyastuti et al. (2019) focused on the role of motivation as a self-efficacy variable in statistics which could increase students' abilities to complete learning tasks. Saeid and Eslaminejad (2017) defined motivation as the driving force that triggers all

human actions. Students with high motivation levels achieve maximum statistical learning outcomes, since a motivated psychological state helps them maintain the needed intensity throughout their learning process (Althauser, 2015).

Finney and Schraw (2003) aid in developing a measure of statistical self-efficacy that can be used to examine students' self-efficacy throughout the semester. To achieve valid results in statistics, the authors recommended avoiding general measures of self-efficacy since they can decontextualize self-efficacy judgment. Instead, the design process should focus on task-specific activities, such as computing standard deviation, instead of domain-specific activities, such as learning statistics. Self-efficacy judgment offers better predictors of performance since it enables the establishment of a particular learning outcome criterion concerning what is being compared. It can be challenging to develop a direct relationship between self-efficacy and statistical achievement, which then interferes with the process of devising measurement standards (Althauser, 2015).

Nevertheless, self-efficacy to learn statistics (SELS) and current statistics self-efficacy (CSSE) have proven to be reliable methods of measuring self-efficacy levels during numerous challenges (Gomez et al., 2022). CSSE defines a student's confidence in solving tasks related to statistics, while SELS measures a student's confidence in building the skills required to solve specific statistics-related tasks. Therefore, combining the two measures can show the skill growth towards the desire to learn statistics and the psychological outcome of learning statistics. This then enables an understanding of how statistical self-efficacy can contribute to improved performance.

Goal Orientation Theory

Achievement goal-oriented theories (GOT), which became popular among educators studying academic motivation in the 1980s, may be defined as a social-cognitive theory of achievement motivation. The difference between GOT and other theories hinges on their overall focus. While other motivational theories tend to concentrate on "learners" beliefs regarding their performances, GOT focuses on what stimulates students to be concerned with and engage in learning while considering why the goal is important to them (McCollum & Kajs, 2007; Woolfolk-Hoy & Hoy, 2006). Meece et al. (1988) defined goal orientation as "a set of behavioral intentions that determine how students' approach and engage in learning activities."

Early versions of GOT focused on the dichotomy between mastery-goal and performance-goal orientations (Ames, 1992). In this dichotomous view of this process, mastery goal orientation could be identified as an intention to attain competence, expand knowledge, and improve understanding by using effortful learning (Ames, 1992). In other words, an individual focuses on improving their skills by concentrating on the process of learning attained through effort. Here, goal orientation is the product of effortful learning.

By contrast, a performance-goal orientation could be defined as the intention to achieve positive appreciation or feedback from others (Ames, 1992). In other words, learners with a performance-goal orientation focus on the opinions of their peers, society, and instructors when involved in a learning task. In the original dichotomy, learners were divided into these two categories. Performance goal orientation is generally understood as self-enhancing goal orientation (Skaalvik, 1997) and ego-involved goal orientation (Nicholls, 1984)

According to this dichotomous view, mastery-oriented learners were able to use selfregulation and self-monitoring skills by adapting to their learning process. Mastery goal

orientation allows students to complete complex tasks, as they invest their effort in the process due to their interest in the subject and the learning process (Butler, 1987). In this case, learners must be able to invest their efforts rather than have an intrinsic ability to learn. According to this dichotomous view of GOT, mastery goal orientation is an indicator of effective academic performance (Elliot & McGregor, 2001). According to Ames (1992), mastery-oriented learners have the possibility to experience pride and satisfaction with not only their results but also the entire process of learning. Overall, it was considered that learners must lean towards a mastery orientation to attain their academic goals.

As performance-oriented learners aim to outperform others, they find no intrinsic value in goal orientation, which limits their learning ability (Dweck, 1986). This dichotomous approach to learning implies that learners with a performance orientation are more likely to shift their focus from goals to acquire new skills, abilities, or knowledge to goals revolving around outperforming others, thus limiting their attention and focus. Since the main objective of performance-oriented students was to outperform others rather than appreciate the process of learning, the intrinsic value of learning was not attained (Dweck, 1986). According to the dichotomous view, performance-oriented learners relate their success or failure to the difficulty of the task or their own abilities.

Such an approach undermines the importance of effort (which is present in masteryoriented learners), which limits learners' abilities to attain success in challenging and complex tasks. According to this approach, performance-oriented learners could generate only limited information processing and use simple memorization as their primary means of learning (Dweck, 1986, Butler, 1987). Such students are more likely to avoid challenging tasks due to the risk of

failure and the requirement of effort in attaining their goals. Performance orientation was associated with a negative perception of a subject.

It is believed that students with a performance orientation tend to quit tasks early, avoid challenging assignments, and lose their motivation once they start performing better than the rest of the class (Tan & Miksza, 2019). Nevertheless, this approach to GOT has lost favor over time, particularly as researchers began to analyze people's motivations for learning.

Hence, the dichotomous approach to GOT gave way to a new model of PAV (performance-avoidance) goal orientation. For instance, Elliot and Harackiewicz's (1996) study on mastery- and performance-oriented learners produced mixed results, reporting that the latter could still manifest intrinsic orientation and invest their efforts in attaining their goals. The authors criticized the dichotomous approach for its inability to distinguish between approach and avoidance characteristics. They used Atkinson's (1957) theory to develop a new framework in which orientation characteristics are essential when classifying learners (approach vs. avoidance). This theory claims that some learners are oriented toward success as their goal (approach), while others focus on avoiding failure (avoidance).

Specifically, PAP-focused learners focus on positive feedback regarding their competencies in relation to others. By contrast, PAV-focused learners concentrate on avoiding negative feedback regarding their competencies (McCollum & Kajs, 2007). Thus, learners who use the PAP strategy try to receive better grades than their peers, while those who use the PAV strategy strive to avoid earning lower grades than their peers. Consequently, the trichotomous GOT model, including the approach and avoidance strategies, becomes more reliable when explaining learning styles (Elliot & Harackiewicz, 1996; McCollum & Kajs, 2007).

The trichotomous approach to GOT points to PAV as a negative goal orientation strategy, as avoidance integrates intrinsic motivation. Students who use this strategy lack effort and persistence; their intention to learn is triggered by their fear of failure, and they focus primarily on low-performing peers rather than on high-performing peers (Elliot et al., 1999). A PAV orientation is associated with superficial information processing, poor organization, and lower academic performance (Elliot et al., 1999). On the contrary, a PAP goal orientation has been connected to high academic performance (Barron & Harackiewicz, 2001). PAP orientation emphasizes the importance of orientation toward successful peers in the form of competition rather than considering mediocre performance as the primary goal.

The trichotomous version of GOT depicts PAV as an ineffective approach to goal orientation as it does not allow learners to attain ultimate success (Barron & Harackiewicz, 2001). In other words, fear of failure is seen as a destructive feeling that limits learners' capabilities. However, the development of research in this area has challenged theorists to develop a new and more complex approach to goal orientation and GOT. For instance, academic procrastination, which is typical of learners who utilize an avoidance goal orientation, is associated with poor academic achievement (Dikmen & Bahadır, 2021). By contrast, a goal-oriented approach mediates the relationship between academic procrastination and academic achievement by preventing or minimizing procrastination (Dikmen & Bahadır, 2021).

The next installment of GOT is the 2x2 model of goal orientation. Elliot et al. (1999) pointed out that in addition to PAV, learners could have a MAV goal orientation, wherein they avoid self- or task-referential incompetence. The 2x2 model contrasts the MAP goal orientation with the MAV goal orientation.

In this model, MAV learners strive to avoid losing their competencies, skills, and academic success, and this desire drives their learning process (Elliot & McGregor, 2001). While PAV orientation is based on avoidance of failure, MAV orientation indicates that learners want to avoid losing their attained success. While a MAV orientation indicates a lack of intrinsic motivation, it is not entirely devoid of this type of motivation, unlike PAV. Elliot and McGregor (2001) constructed a 2x2 achievement goal framework, wherein an intrapersonal approach versus a normative and valence approach versus an avoidance approach were presented in two dimensions, as shown in Figure 1. In this model, the "intrapersonal approach" refers to the goals set by individuals concerning their previous performance, whereas the "normative approach" stems from the orientation of an individual toward external stimuli (Elliot & McGregor, 2001).

Table 1

2x2 Achievement Goal Orientation Theory

| | | Intrinsic | Normative |
|---------|---------------------|---------------------------|--------------------------------|
| Valence | Approaching Success | Mastery-Approach Goal | Performance- Approach Goal |
| | Avoiding Failure | Mastery-Avoidance Goal | Performance- Avoidance Goal |

Note. This model was developed by Elliot &McGregor (2001)

The 2x2 model has similar rules to the trichotomous view of GOT. Avoiding failure is seen as a negative approach to learning and goal orientation. Regardless of whether the avoidance takes the form of MAV or PAV, these learners rely more on fear of failure or losing their status and knowledge than on intentions to gain success (Elliot & McGregor, 2001). Even

though the PAP model orients toward environmental or normative factors, it is still possible for such learners to encounter success in goal orientation (Elliot & McGregor, 2001).

Here, intrinsic orientation is not viewed as the only determinant of successful goal orientation or academic performance. Valence is also considered. In their comparison of the trichotomous and 2x2 frameworks of goal orientation, Simamora and Mutiarawati (2021) found that the 2x2 model is more suitable for and more accurate at measuring goal orientation among learners. Keklik and Keklik (2013) validated the 2x2 AGO framework by showing that the achievement goal orientation scale (AGOS) was able to distinguish among the four categories of learners (MAP, MAV, PAP, and PAV). An analysis of learners' success according to their form of goal orientation showed that self-efficacy for performance and learning was an effective tool in predicting goals, especially academic achievement goals (Keklik & Keklik, 2013).

An empirical analysis of the 2x2 framework showed that the approach method has positive but limited statistical significance. While it had a positive impact on intrinsic motivation, it was not the best measure of exam performance (Korn & Elliot, 2016). At the same time, performance-oriented goals may not be suitable for integration in one-time performance examinations as they do not allow learners to focus on their results (Korn & Elliot, 2016). Failure-avoidance goals are associated with an adverse empirical profile, as they cannot stimulate intrinsic motivation and do not encourage students to achieve good performances (Korn & Elliot, 2016).

The combination of avoidance and performance orientation is the least effective when measuring and producing outcomes since it does not allow learners to foster their efforts or motivation to learn (Korn & Elliot, 2016). Erturan et al. (2020) confirmed the efficacy and validity of the 2x2 model in the context of physical education goal orientation. The 2x2

achievement goal model has proven to be a better framework for interpreting the relationships between achievement goals and learning approaches (Üztemur, 2020). Wolters (2004) found that mastery structure (based on six items that asked students to report on whether the instructional practices in their math class emphasized learning as much as possible) and mastery orientation.

Based on five items that reflect the student's desire to learn as much as possible and engage in challenging coursework are directly associated with adaptive outcomes for all students. The facet has proven to be the most effective approach when defining goal orientation. Overall, this study recognized that, although performance success can be associated with multiple factors, goal orientation plays an essential role.

Achievement Goal Orientations and Online Learning

By using the 2x2 achievement goal orientation theory as a model of measurement of learners' goal orientation, it is possible to determine their probability of success and the possibility of attaining their academic and professional goals. As was noted earlier, students had to change their mode of learning by transferring from face-to-face to virtual learning due to the COVID-19 pandemic. Only a few studies have analyzed the 2x2 achievement goal orientation framework in the context of virtual learning.

For instance, Yeh et al. (2019) used a 2x2 framework as a basis for measuring students' online learning environments. Their study showed that students with higher mastery-approach goals were able to implement different self-regulating learning strategies and supportive online learning behaviors, which predicted their positive academic outcomes (Yeh et al., 2019). On the contrary, students with mastery-avoidance goals failed to adopt effective self-regulating learning strategies and supportive online learning behaviors, which limited their performance outcomes (Yeh et al., 2019). Moreover, performance-approach and performance-avoidance goals revealed

the limited capabilities of students applying effective self-regulating strategies and online learning behaviors, especially what concerns the avoidance model (Yeh et al., 2019). Overall, this study supports the theorization of the 2x2 model, yet it does not support a positive attitude toward performance-approach goals.

Another study conducted by Yang et al. (2016) revealed that performance-approach goals predicted online help-seeking behavior differently compared to those using the performance-avoidance approach. Students who used avoidance approach goals (mastery or performance-oriented) reported more help-seeking, while those with approach-oriented goals were less likely to seek help when learning online (Yang et al., 2016). It was implied that online courses must be designed to satisfy the demands and needs of all learners, despite the approach to goal orientation that they tend to choose (Yang et al., 2016).

Xu (2021) recognized that self-regulation of online assignment behavior among students with different goal-orientation approaches showed different results. Students with masteryapproach and performance-approach achievement goal orientations were more likely to use better self-regulating assignment behavior compared to those who were failure-avoidance oriented (Xu, 2021). Overall, studies exploring the use of goal orientation theory in the context of the online environment are limited. A 2x2 achievement goal orientation theory is the most suitable theoretical framework for this study, as it allows one to assess the goal orientation of learners in a more complex and comprehensive manner using four criteria and two levels of definition and valence. Moreover, this theory allows one to assess goal orientation among learners enrolled in online courses, which is one of the goals of this study. In addition, the 2x2 model is more detailed and comprehensive in evaluating learners' initial core orientation (to

master skills or receive praise) and their valence (avoiding failure or seeking success), which is critical for this study.

Learning Outcomes

Learning outcomes are the products of reactions triggered by experiences that aim to acquire intelligence through practice and effort (Triwahyuni et al., 2021). They are usually measured and observed at the end of the learning experience. In the field of science, the abilities a student acquires can be measured in three aspects of mastery: psychomotor, affective, and cognitive (Sinar, 2018). Cognitive development describes a student's ability to understand principles or concepts concerning the components of thinking and believe that these principles or concepts will be understood, applied, remembered, analyzed, evaluated, and developed to create learning outcomes. Suartama et al. (2020) defined cognitive outcomes as those related to acquiring information, knowledge, and intellectual skills. In this context, intellectual skills are the generic and domain-specific abilities involved in thinking, reasoning, decision-making, and broad comprehension. Broad abilities are self-regulated skills that refer to goal setting, time management, task strategy, self-evaluation, and environmental structuring. At the same time, knowledge can be measured as academic achievement or conceptual knowledge (knowledge acquired from the course under different topics and units).

On the contrary, the affective domain focuses on students' feelings, emotions, and values. It teaches reactions to stimulation, internalization, sensitivity in receiving, and the willingness to organize the values chosen. Supena (2017) explains that high and low student learning outcomes are dependent on internal factors (a student's attributes) or external factors (the school environment, learning methods, family, and so on). The achievement of cognitive learning outcomes can be assessed through written tests taken before and after the learning experience. At

the same time, affect outcomes are measured through observation during learning. Gray and DiLoreto (2016) emphasize the need for OLRS to give students quality time throughout the learning process, thereby enhancing effective outcomes.

Triwahyuni et al. (2021) hypothesized that cognitive outcomes could be measured by a student's general background and mathematical competencies in statistics. At the same time, selfefficacy, attitudes towards statistics, and satisfaction were the affective outcomes being measured since they determine the students' beliefs regarding their abilities and feelings towards the course. Instruments like surveys, user data, observations, and interviews are used as external evaluation tools to explore the different aspects of affective statistics outcomes (Wei et al., 2021). However, questionnaires are the most frequent instruments used to survey learners. User data is frequently used to obtain information regarding how students behave. This data includes the number of replies, posts, and comments students make in discussion forums, the time/duration spent watching video lectures, and their submission and completion of assessments throughout the learning process. This data can also help the researcher understand the complex and differentiated pathways of the learning process. Mahauad et al. (2018) explain that user data can be used to understand the pathway of learners' access to learning resources, which then identifies the self-regulated strategies, problem-solving patterns, and social networks that emerge in the process.

Course Satisfaction as an Affective Learning Outcome

Affective learning is focused on impacting change in learners' attitudes and beliefs. Learning strategies and student emotions have been strongly associated with learning achievement and have been considered critical elements in students' learning satisfaction (Wu et al., 2021). Satisfaction is an outcome of affective learning since it is the measure of affective

responses or aggregate feelings towards various factors (Hoque, 2017). Satisfaction in this domain manifests through outcomes such as self-esteem, confidence, and motivation. Affective learning supports learners' psychological wellbeing. Hence, affective learning has been increased to achieve positive outcomes regarding young people's mental health (Teraoka et al., 2021). Overall, satisfaction manifests through positive emotional responses, which are the core elements of affective learning. A few affective teaching strategies can be applied to increase satisfaction as a learning outcome. These strategies encourage feedback and ask deductive questions.

Gray and Di Loreto (2016) focused on the different dynamics that impact the affective elements in determining course satisfaction. The authors used course structure and organization, learner interaction, instructor presence, student engagement, infrastructural reliability, and perceived learning to investigate the elements that determine course satisfaction. The study used a survey that asked students if the online course enabled them to prepare to be future leaders or enhanced their probability of achieving their life goals. They found that students who perceived the online course as an improvement opportunity had a higher probability of being satisfied with the classes than others. This enabled them to increase their commitment to the classes.

Students have shown higher levels of course satisfaction in cases where they felt that their tutors communicated effectively, had a good rapport, and encouraged or facilitated their learning process (Allen & Seaman, 2016). Moreover, students' course satisfaction levels increased with their perceptions of the class having a sense of community and teachers' presence with asynchronous video feedback, which provided higher satisfaction levels than text feedback (Gray & Di Loreto, 2016). Students believed that video feedback made the nuance of the communication clearer and that lecturers had a higher probability of caring about them when using this form of communication.

Di Loreto and Gray (2016) explained that motivated students who excel in their courses or who are involved or invested in their desire to learn and are willing to place an extra effort in doing so have a higher probability of being engaged in their education. Course engagement extends beyond traditional approaches to understanding instructional effectiveness and includes students' mastery, retention, and satisfaction perceptions. The diverse nature of online courses has allowed the gathering of information regarding student satisfaction, which aids in improving the dynamics of online courses. Course organization and structure integrate the design of the course schedule, curriculum, instructional strategies, methodologies, and general planning of the course throughout the learning process (Allen & Seaman, 2016). Lack of technological infrastructure and reliable expertise to use the online teaching and learning resources can be a source of negative perception and can promote attitudes and behaviors that hinder independent learning and lead to course dissatisfaction.

Self-Directed Learning Readiness and Learning Outcomes

Learning and teaching experiences have evolved due to significant changes in educational technology. Using digital tools in learning encourages students to be more active in their studies outside the classroom. Learners can gain access to a plethora of materials via the intranet or the internet. This allows them to practice SDLR. SDL is a process in which learners can take the lead in determining their learning needs and managing their learning strategies and outcomes, with or without the assistance of others. Relatedly, SDLR examines the abilities, attitudes, and personality traits required for self-directed learning (Chen, 2022). Self-directed learning readiness plays a significant role in improving student outcomes, grades, and course satisfaction. Students can develop various essential aspects, such as research skills and selfdiscipline traits. Through this research, students can continually develop an urge to gain new

knowledge and skills. Gagnon et al. (2013) conducted a study wherein students in a control group were required to purchase course notes, while those in the intervention group had electronic access to similar notes and were even allowed to make copies. According to the researcher, a few students from the intervention group who did not have to purchase notes found it less favorable. Nonetheless, the study found no difference in the level of course satisfaction between the two groups, thus leading to the conclusion that the "unfair" situation did not affect outcomes. Such findings show that teaching methods lack a direct and significant effect on SDLR, knowledge acquisition, and satisfaction. Students that exhibited low motivation levels in the intervention group performed relatively better than those in the control group, while the level of motivation exhibited a positive association with course satisfaction in both groups. According to Kuo et al. (2013), the first dimension of a learner's readiness is internet self-efficacy and a computer, and this is rarely addressed. At the same time, the researchers point to existing evidence supporting the influence of SDLR and self-efficacy on satisfaction.

Mead (2011) hypothesized that students with a high level of SDLR were most likely to exhibit high satisfaction with online courses. According to the findings, the association between a student's course satisfaction rating and their level of SDLR was moderate but significant. Researchers such as Fisher et al. (2001) have also established a relatively strong association between SDLR and overall learning satisfaction. Grow (1991) proposed the staged self-directed model to encourage students to become more SD learners. In the proposed model, the researcher discovered that a student would likely become more used to a set goal to increase their SD thinking levels. With this, the researcher concluded that if a perceived grade could be termed a form of goal setting, perhaps students could discover new levels of self-directedness that, as a result, would lead to higher course satisfaction levels.

Cho and Kim (2021) found a positive correlation between self-directed learning readiness and learning satisfaction from the study investigating the effects of face-to-face flipped learning and non-face-to-face flipped learning on learning satisfaction, self-directed learning readiness, and interactions between instructors and learners among undergraduate nursing students. Online classes where the instructor and the learner operate in different times and spaces require higher levels of self-directed learning. Therefore, it is harder for teachers to manage learning for their students in these settings. The results supported Yoo's research (2020), which demonstrated the importance of self-directed learning,

In Monroe's (2016) study, which considered participants' differences as variables, there was a lack of a combination of students' grades that could overwhelmingly predict the aptitude for SDL. Considering the importance of improving SDL skills among medical students, there was a need for a more pragmatic and reliable approach to measuring the relationship between SDL and test scores. However, the data analysis offered no definite means of documenting the SDLR of the medical students based on the evaluation metrics. Regarding the academic level, Örs (2018) discovered that the SDLR scale presented a trend where students in the low-grade level received a lower mean score than their high-grade counterparts. Therefore, it can be assumed that SDLR increases with an increase in academic level.

Shokar et al. (2002) discovered a positive correlation between students' scores on the SDLR scale and their course grades. Despite the link between SDLR and a student's academic performance, other factors that constrain the association might be at work. The researcher assumed the possibility that the type of curriculum (such as the problem-based curriculum for medical school students) impacts the strength of the association, since students who are low in SDLR are more likely to be enrolled in such a curriculum.

Self-Directed Learning Readiness and Self-Efficacy

KOÇ and Turan (2018) investigated how SDLR impacts critical thinking and selfefficacy in the context of physical education and sports school. They established that SDLR affects self-efficacy and critical thinking characteristics in the same way as it affects other aspects related to education, such as course satisfaction and grades. The authors also found that SDLR predicts general self-efficacy and critical thinking by 50.5%. KOÇ and Turan (2018) argued that SDLR tends to have a positive impact on critical thinking and self-efficacy.

Tuong and Truòng (2021) also investigated SDLR and its impact on self-efficacy among students at Vietnam University and found a strong positive correlation between SDLR and selfefficacy. The authors acknowledge the limited research on the relationship between these variables. They added that the greatest impact of SDLR was on the learner's self-management factors. They recommended that by emphasizing improving students' self-efficacy, the university is likely to improve education outcomes such as course satisfaction, grades, and self-directed learning readiness.

A study by Yao (2021) regarding nursing students' ability to solve workplace problems found that SDLR had a significant positive impact on academic self-efficacy. This finding was also confirmed in other studies, including a study by Murniati et al. (2022), who also investigated university students, and Coros and Madrigal (2021), who obtained the same results in their study of high school students.

Self-Directed Learning Readiness and Achievement Goal Orientations

Achievement goal orientations measure students' general tendencies in approach, engagement, and evaluation of their academic progress and performance. Understanding SDLR as a high level of motivation to learn can help predict the success of the learning process. This can lead to satisfactory achievement in student learning (Siddiqui, 2021).

Achievement goal orientation can be a construct formed by dividing students' achievement intentions into two types: a general approach and a performance orientation. The general approach is characterized by students being well-organized, autonomous, proactive, and self-directed toward learning. By contrast, performance-oriented students can be highly persistent and focused on task completion. Well-organized students can designate clear goals, adopt a goal-oriented strategy, and practice self-confidence in their learning (Wong, 2021). This is essential, as it helps them be self-directed in the learning process. Wong (2021) suggested that self-directed learning boosts students' motivation in autonomy experiences. SDLR is controlled by learners' levels of autonomy, which is a key aspect of building strong core competence in self-direction. For instance, when students feel more autonomous about their SDLR process, they tend to become more comfortable expressing their ideas and views. However, learners who experience low levels of autonomy gradually learn to work within the framework of their instructor's or teacher's expectations. As a result, they can only work effectively under the teacher's guidance.

Learners motivated by self-direction show their willingness to take on challenging tasks (Bonk & Lee, 2017). Such students tend to adopt effective, goal-oriented strategies to improve their learning outcomes and achievement. Moreover, when SDLR is implemented based on self-direction, learners can increase their levels of persistence and motivation. For example, a learner who takes an active role in goal setting, monitoring, and evaluating their performance will be motivated to learn and produce better grades. Similarly, learners with positive self-direction experience are likely to increase their levels of persistence in tackling difficult tasks.

Ariffin et al. (2020) argued that when the learners' motivations are situated between external goal setting and performance motivation, they are likely to experience low persistence. Students tend to be less oriented when they are less motivated. This, as a result, affects their learning outcomes regarding general tendencies in approach, engagement, and evaluation of their studies. The authors emphasize that SDLR helps learners achieve goal orientation and develop their skill sets in a manner that allows them to perform well. Through SDLR, students can set achievement goals that help them improve competence and performance. Adopting SDLR aids learners in effective approach, engagement, and evaluation. This ensures that the 2x2 AGO is satisfied and that learning outcomes are improved. Simply put, SDLR is an essential part of the learning process, as it enhances academic performance and overall achievement.

Self-Efficacy and Learning Outcomes

Liu et al. (2010) integrated cooperative learning, problem-solving skills, and LEGO robotics into the design of a suitable robotics course for preservice teachers. Their findings indicated that the level of preservice teachers perceived self-efficacy positively impacted satisfaction with the learning environment, learning content, and method. In addition, Wang et al. (2013) used structural equation modeling to examine the association between SRL, student characteristics, self-efficacy in technology, and course results. Based on the findings, students who had previous e-learning experience exhibited a more effective strategy when taking their online courses. Therefore, they had a higher motivation level throughout their time as students. An increase in course satisfaction and technological self-efficacy was also noted. Lastly, students recorded better final scores with high course satisfaction and technology self-efficacy.

Jan (2015) assessed 103 graduate students who took online courses regarding various factors, including computer and academic self-efficacy, prior online learning experience, and

student satisfaction rates. Their findings indicated a significant positive association between computer self-efficacy and prior online learning experience and between student satisfaction and self-efficacy. Similarly, there was a positive association between computers and academic selfefficacy and between student satisfaction and prior online learning experiences. However, the study failed to establish a significant or positive association between student satisfaction and computer self-efficacy. Alqurashi (2016) found that internet and computer self-efficacy are strong predictors of students' performances and satisfaction with online learning, and other studies failed to exhibit this link.

Self-efficacy and emotions such as anxiety were presumed to directly impact students' grades (Barrows et al., 2013). Self-efficacy strongly influences how individuals believe in their capacities, which, as Bandura (1993) stated, affects their academic performance. It is, therefore, reasonable to assume that students with higher self-efficacy are less worried about their grades/test scores. Abdi et al. (2012) discovered similar results in their study of high school students. According to the researchers, there is a significant relationship between self-efficacy and overall test score.

Moreover, regression analyses have established that self-efficacy is an accurate predictor of academic performance. According to Barrows et al. (2013), empirical evidence supports the association between self-efficacy and overall academic success. The findings furthered the empirical literature by exhibiting an association between text anxiety, self-efficacy, and single test grades. Additionally, it was discovered that self-efficacy failed to moderate test grades or anxiety, which, interestingly, implicated future researchers in discovering the existence of a moderator between self-efficacy and test grades.

According to Gray (2018), persistence and retention rates in online classes were relatively lower compared to those of traditional online courses for adults. Based on the study's findings, online learning educators and instructional designers of online courses are likely to reduce attrition by purposely designing online courses to increase students' SDLR. For existing courses, instructors can examine students' SDLR at the beginning of the course to identify those who might be more likely to exit the course. Here, grades can aid instructors in providing additional support to students more likely to drop the course.

Achievement Goal Orientations and Learning Outcomes

Goal orientation theory has been extensively researched in educational psychology, as it tries to explain students' motivations concerning learning and postulates that a person's goal orientation will impact their insight. Studies on goal orientation have emphasized the benefits of learning and performance as two crucial goals that drive learners' motivations and boost their course satisfaction (Yeh et al., 2019). Mastery goals describe a student's motivation for obtaining knowledge and developing skills and competencies. Studies have related mastery goal orientation to active cognitive engagement, learning satisfaction, persistence, and improved effort. In their studies of high school students, Ames and Archer (1998) revealed that course satisfaction is one of the main facets of mastery goals. They explained that learners who use both mastery and performance approach goals tend to exhibit equal responses regarding learning and course satisfaction.

The two dimensions of mastery and performance goals help learners understand the need to master a task and develop higher self-competence. Achievement goal theory determines learners' desired goals for engaging in learning. It is, therefore, closely linked with their learning strategies, learning outcomes, and level of satisfaction. Dubey (2000) confirmed that only

avoidance goals were undesirably connected to online participation, satisfaction, and accomplishment.

Cho and Shen (2013) examined the roles that goal orientation and academic self-efficacy played toward a student's achievement in tandem with effort regulation and online interaction regulation. The results indicated that intrinsic goal orientation and self-efficacy were clear paths toward metacognitive self-regulation. There was no type of self-regulation prediction that could be determined from extrinsic goal orientation. This study showed that students' intrinsic goal orientations at the individual level and self-efficacy in academic courses directly impacted their overall academic achievement.

According to Standage and Treasure (2002), task and ego accomplishment goal orientations impact students' intrinsic motivations regarding physical education. The researchers noted that according to the theory of self-determination (Deci, 1985), the unidimensional construct of intrinsic motivation states that intrinsic motivation is only a single form of motivation. Therefore, this study aimed to determine the relationship between achievement goal orientations and situation-motivating factors in the case of physical education. Data was collected using structured questionnaires distributed among 182 male and 136 female middle school children with an average age of 13.2 years. The results indicated that task orientation was found to have reduced effects on self-determined motivation. The research also split the study population into goal groups. The analysis of the resulting data revealed that groups with high task orientation reported better motivational adaptation than those with low task orientation.

Klein et al. (2006) investigated how learning goal orientation (LGO), learning mode (either in physical locations or blended learning), perceived barriers, and enablers related to

students' motivation affected overall learning and course outcomes. A study size of 600 students either enrolled in physical classrooms or involved in blended learning was used. The results highlighted that students who were involved in blended learning conditions with high goal orientations and those who viewed environmental factors as enablers rather than barriers to learning recorded higher learning motivation. The motivation to learn positively impacted their academic performances in terms of grades, satisfaction, and metacognition.

Moeller et al. (2012) stated that the relationship between goal orientation and student motivation is an area of study of particular interest to researchers. Still, the connection between goal setting and the academic achievement of students at the classroom level has arguably remained unexplored. Their study analyzed the findings from a five-year quasi-experimental study that examined goal orientation and student achievement in a Spanish-language high school classroom. LinguaFolio, a portfolio focusing on self-assessment, goal setting, and evidence collection of language understanding among students, was implemented in 23 high schools with 1,273 students. The researchers used a hierarchical linear model to analyze the relationship between goal orientation and student performance at the individual levels of teachers and students. The resulting correlational analysis revealed a significant relationship between goal orientation and language proficiency achievement.

In statistics, most learning outcome assessments use assignments, tests, writing projects, and exams to test students' cognitive development. These measures are also used to understand the extent to which learners have absorbed the information presented on the subject matter. They are also integrated to be part of the cognitive assessment outcomes. Graded assignments aid in examining the required learning outcomes and compromise the mandatory final course grade. In statistics courses, explicit and standardized answers are used to assess learners. Moreover, auto-

graded and short-answer questions with feedback have proven effective for many students in understanding their performance, while exams focus more on practice and disciplinary-related skills.

Achievement goal theory tries to outline and forecast students' school-related performances. Wang et al. (2021) show a significant association between the four subscales of achievement goal orientation and academic performance. Initially, achievement goal orientation was characterized by students' task commitment. They were distinguished by two types: mastery goals, which focus on competence growth, and performance goals, which are concerned with demonstrating competence. Wang et al. (2021) proposed that goal orientation positively predicts academic adjustments and performance. In a recent study, students with high mastery-oriented goals presented a considerable aptitude for adapting to penalties and showing motivation and good academic ability compared to those with weak mastery (Wang et al., 2021). Masteryoriented students tend to spend most of their time studying and, as a result, receive good grades. This also shows that students who use mastery ability and performance-based goals have twosided advantages. They are likely to use intellectual tactics and make extra efforts to attain better academic performance due to their high motivation. Lastly, students with performance-oriented goals risk obtaining poor grades because they experience burnout, higher test anxiety, and more significant challenges compared to the group that uses mastery goals.

Achievement goal theory offers academic institutions a priceless understanding of how their students respond when they encounter academic activities. A study by Alrakaf et al. (2014) on pharmacy students in Australia shows how performance approaches correlate with good grades. The students in the first year who used the approach recorded higher grades (Alrakaf et

al., 2014), which is consistent with prior findings that indicate a positive association between academic achievements and a performance approach.

Yeh et al. (2019) examined the underlying relationship between goal orientation and academic achievement among online learners. The researchers simultaneously investigated the structural relationships between 2x2 achievement goal orientation, online courses with instructors' support, self-paced learning strategies, and the expected academic performance among several online courses with both undergraduate and graduate student respondents. The effects of self-paced learning and online instructors' supportive learning behaviors mediated the relationship between achievement goal orientations regarding students' academic expectations. Results showed that mastery-approach and avoidance goals were vital in predicting selfregulated learning strategies and online instructor-led behaviors. They also helped predict students' academic performance outcomes in their online courses. The authors noted that students with higher mastery-approach goals usually adopt multiple types of self-regulated learning strategies and utilize different supportive online learning mediums to enhance their learning experiences, further increasing the chances of better academic outcomes. By contrast, those students with higher mastery avoidance goals were less likely to adapt self-paced learning strategies and online instructor behaviors, resulting in poor academic performance.

According to Cazan (2014), online learning is usually individual and student-centered. This is because the learning environment requires learners to have a sense of self-regulation and to be learner focused. The online learning experience demands that students possess selfregulatory skills in setting up goal orientations, monitoring their progress, seeking clarification and help from their instructors whenever needed, and efficiently managing their time. According to the author, self-regulated learning positively impacts students' academic achievements and

performances in online courses. This study aimed to determine the relationship between selfregulated learning and academic performance in the case of online learning. Undergraduate student participants enrolled in online courses were the subjects of the study. The researchers noted that all learning objectives, including submitting quizzes, summative assessments, and learning resources, were done online via electronic resources. The results of an online questionnaire on self-regulated learning revealed two groups of learners: those with low levels of self-regulation and those with high levels of self-regulation. An analysis of the results indicated that self-regulated learning positively impacts academic achievement for students with high selfregulation in online courses. Moreover, linear regression analysis outlined that computer selfefficacy and online self-regulation were good predictors of grades at semester closing for the online courses. The study recommended that future research focus on epistemological beliefs, motivations for learning, and anxiety related to computers to better explain the concept of selfregulated learning in online settings.

Pulkka and Niemivirta (2013) investigated whether students' goal orientations and perceptions of their study environment affected their academic achievement. Data was collected from 169 students from the Finnish National Defense College, and students' goal orientations and learning environment evaluations were assessed twice every four months. The study group was then subdivided into four subgroups, and each group was compared against the others. The study revealed that students with different goal-orientation personalities displayed different levels of effort, attainment, and participation. Students with an orientation channeled toward increased competence and mastery goals registered more success and positive outcomes in their evaluations. On the other hand, students who avoided goal-oriented efforts displayed incompetence and poor academic achievement.

Summary

This chapter reviewed the literature on the effects of OSDLR on grades and course satisfaction for online learning statistics through the mediating variables of self-efficacy and 2x2 achievement goal orientations. The chapter was divided into two main sections. The first discussed the theoretical framework of this thesis, which uses self-directed learning, 2x2 achievement goal orientation, and self-efficacy theory, which is a subset of Bandura's social cognitive theory. The second discussed the empirical literature, which focused on the relationship between SDLR, self-efficacy, 2x2 achievement goal orientations, and learning outcomes. The learning outcomes examined in this chapter were self-efficacy, course satisfaction, grades, and 2x2 achievement goal orientation. The chapter also identified four achievement goal orientations: mastery approach (MAP), mastery avoidance (MAV), performance approach (PAP), and performance avoidance (PAV). The chapter summarized important relationships between SDLR and different learning outcomes from previous research, and hypotheses will be developed based on these findings.

Previous research has suggested that learners with a mastery goal orientation have a higher probability of learning and developing competencies, whereas those with a performance goal orientation tend to demonstrate their competencies by outperforming others. There is considerable research that exists between goal orientations and learning outcomes. It has also suggested a positive relationship between mastery goal orientations and learning. However, it has been established that, thus far, there is limited research on self-directed learning readiness and different learning outcomes, and as such, a gap in the literature exists. By investigating the effects of self-efficacy, SDLR, and goal orientation on learning outcomes for learning statistics online, this study hopes to make important contributions toward filling this gap.

The current study covers the existing gap in information regarding the effects of SDLR, self-efficacy, and goal orientations on learning outcomes for learning statistics online with the increased demand for online courses after the pandemic. The study aims to create awareness and to inform learners and tutors in online statistics courses on how they can rely on self-efficacy and 2x2 achievement goal orientations to improve their grades and course satisfaction. The audience will also understand the impact of self-directed learning readiness on online course satisfaction and grade outcomes. Even though previous studies have covered online readiness and self-efficacy as independent variables, the current study introduces the dynamics of 2x2 achievement goal orientations in improving self-directed learning readiness in the online learning environment. Previous studies such as Üztemur (2020) have only covered the 2x2 achievement goal as an independent variable in improving the performance of online statistics. Such studies focus on either a combination of two components of the AGO framework (i.e., MAP and PAV) (Korn & Elliot, 2016) or AGO as a framework for improving learning approaches and goal achievement (Üztemur, 2020).

The current study focuses on how the combination of the four elements in the AGO frameworks can improve course satisfaction and grade outcomes at the same time. It, therefore, covers the existing gaps where the previous researchers failed to focus on the dynamics of both the grade outcome and course satisfaction using the four elements of the framework. The current study will not view self-efficacy alone as a determinant of improved grade outcomes as in previous studies (Domenech-Betoret et al., 2017, Betoret et al., 2014). On the contrary, self-efficacy and the 2x2 AGO goal can act as mediators for improving online statistics course readiness, grade outcomes, and course satisfaction. Therefore, this study bridges the independent variables of self-efficacy, grade outcomes, course satisfaction, and online course readiness to

provide information on how students taking online statistics courses can improve their commitment to finishing the course and reach the goals that they established when enrolling in the classes. The study provides information from the preparation stage (online readiness) to setting goals and maintaining the required psychological and academic characteristics (selfefficacy, 2x2 AGO model) to the final stage of learning to achieve results (grade outcome and satisfaction). The study's ability to cover all stages of learning in online statistics makes it superior to other studies that only focus on single stages. For example, Betoret et al. (2014) and Walker and Brake (2017) focused on maintaining students' psychological states when participating in online courses, while Bárkányi (2021) and Macher et al. (2015) focused on the final, or outcome, stages. At the same time, Tang et al. (2021) and Yilmaz (2017) focused on the online readiness component without considering its impact throughout the whole study process. By contrast, the current study will combine the variables and cover the existing gap by showing the role of online readiness, self-efficacy, and 2x2 AGO frameworks in online statistics grade outcomes and satisfaction.

The current study will also seek to conduct research that should help fill some of the gaps in the previous literature in this area, such as how students can fully transition into the hybrid or online statistics course model after the COVID-19 pandemic or how tutors can enable their students to shift swiftly. Second, online learning has undergone drastic changes over the past decade, rendering past studies in the field obsolete. The absence of newer studies covering the topics to be analyzed emphasizes the lack of modern contexts for online learning and its impact on the learning process and students. Since online learning has become an increasingly relevant educational model, the lack of current studies in this area could be detrimental to the fields of education and statistics. Third, the instruments used in past studies differ across the research

spectrum. For example, SDLR can be measured according to problem-solving ability, creativity, ability to accept change (Guglielmino, 1978), self-management, desire for learning, self-control (Fisher et al., 2001), motivation for learning, computer self-efficacy, self-directed learning, online communication, and learner control (Hung et al., 2010). Different approaches to measuring relevant concepts create a certain level of ambiguity in analyzing the topics mentioned above. Lack of specificity in the measurement and determination of such concepts as self-efficacy and SDLR indicates the need for an in-depth analysis.

Finally, this study focuses on 2x2 achievement goal orientation and its relation to selfefficacy, SDLR, and learning outcomes in online statistics. No relevant studies in this area have been detected. Also, since this study focuses on online statistics courses, it was important to analyze studies conducted in this field. Yet, only limited inquiries have been conducted in this area, indicating a paucity of research on the topic. Consequently, this study will help fill this gap and expand knowledge on the relationships among goal orientation, self-efficacy, self-directed learning readiness, and learning outcomes (grade and course satisfaction) in online statistics learning.

Chapter 3 - Method

Overview

The purpose of this study was to investigate the relationship between online self-directed learning readiness and learning outcomes such as grade and course satisfaction through the potential mediators of self-efficacy to learn statistics and 2x2 achievement goal orientations (AGO) in online statistics learning. The present study includes an analysis of data gathered from a self-report questionnaire, which was voluntarily completed by students who were studying at a large Southeastern U.S. research institution during the Fall 2020 semester. The questionnaires chosen to collect data for this research were the online self-directed learning readiness (online SLDR) questionnaire developed by Hung et al. (2010), self-efficacy to learn statistics questionnaire (Finney and Schraw, 2003), the achievement goal questionnaire – revised version (AGQ-R) developed by Elliot and Murayama (2008), and the course satisfaction questionnaire (Kuo et al., 2013). This chapter is comprised of the following sections: 1) research questions; 2) research design; 3) participants and sampling procedures; 4) instruments; 5) data collection procedures; and 6) data analysis.

Problem Statement

COVID-19 is a health crisis that has caused drastic changes in the education system. Students were forced to shift from physical to online classes in response to government restrictions on face-to-face interactions, which were considered critical in preventing the spread of respiratory disease. Aguilera-Hermida (2020) explains that these changes have led to new challenges; however, they have also created awareness of the role and advantages of engaging in online learning. Therefore, even after the pandemic waned, the popularity of online learning remained high. Bashir et al. (2021) reported that a university adopted a hybrid (a combination of
online and face-to-face courses) mode of course delivery as a post-COVID solution that recognized the popularity of online learning. Kelly's (2021) survey reported that 73% of students prefer to take fully online courses, and 68% prefer learning using a hybrid solution.

Statistics courses have also been transferred online, with studies such as Ritzhaupt et al. (2020) revealing that students enrolled in the class experience less anxiety, making the module preferred. The popularity of online statistics post-pandemic has created a demand for understanding this learning module and how learners and educators can organize the course effectively. Further demand for understanding the learning module has been created by studies such as those conducted by Figueroa-Cañas and Sancho-Vinuesa (2020), which have reported that the dropout rate of online statistics courses is higher than that of traditional statistics courses. However, through grounded research in theories such as those espoused by Knowles on self-directed learning, scholars have focused on self-efficacy as an approach to improving online statistics. The current study seeks to help close the existing knowledge gap regarding the effects of self-efficacy, SDLR, and goal orientation on learning outcomes for learning statistics online.

Research Questions

1. What is the extent of the relationship between online self-directed learning readiness and grades mediated by self-efficacy to learn statistics and each construct of achievement goal orientations in an online statistic learning environment?

1.1 What is the extent of the relationship between online self-directed learning readiness and grades in an online statistic learning environment?

1.2 What is the extent of the relationship between online self-directed learning readiness and grades mediated by self-efficacy to learn statistics in an online statistic learning environment?

1.3 What is the extent of the relationship between online self-directed learning readiness and grades mediated by each construct of achievement goal orientations (MAP, MAV, PAP, PAV) in an online statistic learning environment?

2. What is the extent of the relationship between online self-directed learning readiness and course satisfaction mediated by self-efficacy to learn statistics and each construct of achievement goal orientations in an online statistic learning environment?

2.1 What is the extent of the relationship between online self-directed learning readiness and course satisfaction in an online statistic learning environment?

2.2 What is the extent of the relationship between online self-directed learning readiness and course satisfaction mediated by self-efficacy to learn statistics in an online statistic learning environment?

2.3 What is the extent of the relationship between online self-directed learning readiness and course satisfaction mediated by each construct of achievement goal orientations in an online statistic learning environment?

Research Design

This study employs a quantitative survey research design. Specifically, a correlational research design utilizing a cross-sectional mediation model was used in this quantitative study to investigate the relationships among variables of online self-directed learning readiness, self-efficacy to learn statistics, 2x2 achievement goal orientations, course satisfaction, and grades in online statistics learning. A correlational study is non-experimental and focuses on collecting data to determine the degree to which a relationship exists between at least two variables where little, or no effort has been made to control extraneous variables (Mohajan, 2020). The mediation role employed during the current study investigated how OSDLR affects learning outcomes such

as grades and course satisfaction through 2x2 AGO and SELS as mediating variables. Previous studies have shown that correlational studies aim to predict possible outcomes and explain human behaviors (Fraenkel et al., 2014), justifying the appropriateness of using a correlational research design in the current study. According to recommendations by Pollack et al. (2012) and Verhulst et al. (2012), correlation studies do not establish causal relations between variables; instead, mediation analysis should be applied to test the connection between meditating, dependent, and independent variables. As such, mediation analysis is the most suitable statistical approach for examining the research questions in this study.

A cross-sectional study design simultaneously collects data on relevant variables from different subjects, people, or phenomena (Spector, 2019). The single-data collection process usually takes place within a short time frame. The process enables researchers to gather preliminary evidence for the relationships that may exist among constructs and provide the foundation for future avenues of research using cross-sectional correlation analysis (Martin et al., 2019). However, the outcome variable (variable Y, i.e., the explained, response, dependent, or predicted variable) and the exposure variable (variable X, i.e., predictor, explanatory, independent variable, which, at times, is referred to as a factor), which are measured simultaneously, are the most critical elements of cross-sectional studies (Gregorich et al., 2021). Nevertheless, in the current study, it was impossible to ensure that the exposure preceded the outcome because there was no follow-up over time.

Participants and Sampling Procedure

The present study explored the relationships among students' OSDLR, SELS, and AGO, and learning outcomes included course satisfaction and grade during the online statistics learning process. The current study used convenience sampling, which is a type of non-probability sampling method where the researcher selects participants who are easily accessible and readily available to participate in the study. This method is commonly used in research studies with time, budget, or logistical constraints (Jager et al., 2017). Convenience sampling is a quick and easy way to collect data, as participants are selected based on their availability and willingness to participate. It can be a cost-effective research method, as there are no extra expenses related to recruiting participants, such as advertising or incentives (Bhardwaj, 2019). This method can also be helpful for researchers with limited access to a large population or difficulty recruiting participants from a specific population (Jawale, 2012). The current study participants were students enrolled in a large public university in the Southeastern United States who had registered for at least one statistics course delivered online during the 2020 fall semester. These students were selected as possible participants because they were enrolled as students at the university and were 19 years old or older.

Instruments

Achievement Goal Questionnaire-Revised (AGQ-R)

Elliot and McGregor (2001) are the pioneers of the Achievement Goal Questionnaire (AGQ) instrument, which aims to evaluate AGOs conceptualized in the 2x2 achievement goal framework. However, in 2008, the questionnaire was revised and improved by Elliot and

Murayama (2008), as cited by Sánchez Rosas (2015), who renamed it the Achievement Goal Questionnaire-Revised Version (AGQ-R). The updated version was designed as a hierarchical model that integrated approach and avoidance achievement motivations. The new version allows for the inclusion of the four AGOs, including MAP-goal orientation, MAV-goal orientation, PAP-goal orientation, and PAV-goal orientation, as it is grounded in a 2x2 achievement goal orientation theoretical framework. Learners with a mastery goal orientation have a higher probability of learning and developing competencies, whereas those with performance goal orientations tend to demonstrate their competencies by outperforming others (Lin, 2021). Therefore, MAP goal orientation focuses on understanding and learning course materials, whereas MAV goal orientation focuses on avoiding losing one's competencies or skills. PAP goal-oriented learners focus on outperforming their peers, whereas PAV goal-oriented learners strive not to appear incompetent (Elliot & McGregor, 2001).

The AGQ-R survey uses a 12-item, 7-point Likert scale that ranges from 1 (strongly disagree) to 7 (strongly agree; Diaconu-Gherasim et al., 2019). The items evaluate students' goals concerning their academic performances; therefore, they rely upon statements such as, "When studying, I aim at mastering the *materials* presented in class completely," and "My studying objective is to perform better than other students." Different studies have used the AGQ scale when measuring AGO. The Cronbach's alpha of mastery approach-goal orientation, mastery avoidance-goal orientation, performance approach-goal orientation, and performance avoidance-goal orientation are 0.84, 0.88, 0.92, and 0.94, respectively (Diaconu-Gherasim et al., 2019; Rosas, 2015). Overall, the results suggested that the AGQ-R is a reliable instrument.

Self-Efficacy to Learn Statistics (SELS)

The self-efficacy to learn statistics (SELS) scale was developed by Finney and Schraw (2003) to measure students' confidence and ability to learn statistics. The instrument has 14 items, and each item is based on a 6-point Likert scale ranging from 1 (no confidence) to 6 (complete confidence) using questions associated with statistics-related tasks, with scores ranging from 14 (low level of statistical self-efficacy) to 84 (high level of statistical self-efficacy). Alhazzani et al.'s (2021) study measured the scale's internal consistency with a

Cronbach's alpha value of 0.975 which shows the scale's good reliability. The study established a negative correlation between SELS and anxiety scores. At the same time, a positive correlation was established between attitudes toward statistics and SELS.

Online Self-Directed Learning Readiness (OSDLR)

Hung et al.'s (2010) self-reported, 18-item scale, the Online Learning Readiness Scale (OLRS), was used in the survey to evaluate online readiness. The OLRS is a multidimensional scale consisting of five sub-dimensions associated with online readiness: self-directed learning (5-item), motivation to learn (4-item), learner control (3-item), computer/internet self-efficacy (3-item), and online communication self-efficacy (3-item). Answers are provided on a 5-point Likert scale ranging from 1 (strongly disagree) to 5 (strongly agree). Self-Directed Learning consists of the first part of OLRS, which seeks to explore how learners take the initiative to understand their learning needs, establish learning goals, identify the required learning resources, choose, and implement appropriate learning strategies, and evaluate learning outcomes (Knowles, 1975). Computer/internet self-efficacy assesses students' abilities to use technology to accomplish online learning. This includes the ability to use computers to accomplish certain tasks and to apply higher-level internet self-efficacy skills. Motivation for learning is concerned with motivational and cognitive variables influencing online learning. This includes both intrinsic and extrinsic motivation. The learner control dimension is concerned with the autonomy and flexibility of the learner in using study materials. The online communication self-efficacy dimension measures the computer-mediated communication involved in online learning. The reliability of the questionnaire was validated with a composite reliability between 0.73 and 0.87 (Hung et al., 2010). Liu (2019) supported the previous study on the role of the OLRS survey in evaluating online readiness, where a Cronbach's alpha value of 0.82 was found. This means

OLRS is a reliable instrument. Items such as: "I repeated the online learning materials based on my needs," "I am open to new ideas when learning online," and "I feel confident in using online tools to communicate with others" were used to measure OLRS.

Course Satisfaction

Course satisfaction cannot be measured using a single template across the different studies (Gray & DiLoreto, 2016; Kuo et al., 2013) instead, it is adjusted according to the course structure, which enables an understanding of students' opinions regarding online learning settings. Kuo et al.'s (2013) study relied on a self-reported, 5-item blended-learning satisfaction scale with answers ranging from 1 (strongly disagree) to 5 (strongly agree). The study demonstrated high internal consistency, which concludes that this scale is reliable, as shown by Cronbach's alpha of 0.8. The study provided items such as "I enjoyed being an online student," "Online learning met my needs as a student," and "I find the experience beneficial," which was used in the current study.

Data Collection Procedures

Data collection for this study was conducted using an online survey with a set of questionnaires, including part 1 for demographics, part 2 for online learning readiness, part 3 for 2x2 achievement goal orientations, part 4 for self-efficacy to learn statistics, and part 5 for course satisfaction. Emails were sent to the Department of Mathematics and Statistics, which was asked to distribute the survey through group emails to the students who were enrolled in at least one statistics course. Invitation emails were then sent through a third party every two weeks over a period of six weeks to all students enrolled in the Department of Mathematics and Statistics. The survey was approved by the Institutional Review Board (IRB) (See Appendix D).

At the beginning of the data collection process, participants were informed of the purpose of the research and the expected time required to take the survey. They were also informed that their participation in the survey was completely anonymous and voluntary. Participants were informed that no foreseeable risks were associated with the study. Furthermore, they were asked to describe how well the survey statements described their online learning readiness, selfefficacy to learn statistics, achievement goal orientations, course satisfaction, and self-reported grades. They were assured that there were no right or wrong answers for each item. In addition, participants were informed that all their personal information and responses would be kept confidential. One-hundred-sixty-eight students participated in answering the survey in the fall semester of 2020, and 121 respondents, or 72.02%, were usable.

Data Analysis Procedures

Data collected from the online survey will be analyzed using IBM SPSS Statistics (Version 26). Statistical analyses will include descriptive statistical analysis, correlational analysis, reliability estimates, and a series of OLS regressions to analyze the path models. First, descriptive statistics for age, gender, grade levels, ethnic/racial identification, major (STEM/Non-STEM), and self-reported grade in one of the statistics courses students were enrolled in were used to portray the characteristics of this research sample. A correlational research design utilizing a mediation model was used to investigate how OSDLR transmits its effect on learning outcomes (grade and course satisfaction) through intervening variables or "mediators" such as SELS and 2x2 AGO. Pearson's correlations will be computed to see whether there are significant relationships among variables. To test the proposed mediation model as diagrammed in Figure 1 through Figure 4 and answer Research Questions 1.1, 1.2, 1.3, 2.1, 2.2, and 2.3 path analysis will be used to investigate the mediating roles of SELS and 2x2 AGO.

There are four different models to answer research questions in the current study. The first two regression analyses from Model 1-1 examined whether the antecedent variable of self-directed learning readiness was associated with the consequent variables of cognitive outcome (grade) (Equation 1) and self-efficacy to learn statistics (i.e., the potential mediator) (Equation 2). The third regression analysis from Model 1-1 included self-directed learning readiness and selfefficacy to learn statistics as predictors of cognitive learning outcomes (Equation 3). In Model 1-2: 2x2 achievement goal orientations of mastery-approach, mastery-avoidance, performanceapproach, and performance-avoidance are included as potential mediators in Equations 5 through 8, which show how self-directed learning readiness is associated with each mediator. The last equation from Model 1-2 is the regression analysis that probes whether grade can be predicted by self-directed learning readiness and each construct of 2x2 achievement goal orientations together (Equation 9). Regression analyses from Models 2-1 and 2-2 examined how self-directed learning readiness was directly associated with affective learning outcomes (course satisfaction) and the potential mediators of 2x2 achievement goal orientation and self-efficacy to learn statistics, respectively. The following equations for each model represent the estimation of the statistical diagram of the mediation model, where a1, a2, a3, a4, a5, a6, a7, a8, a9, a10, b1, b2, b3, b4, b5, b6, b7, b8, b9, c1, c2, c3, c4, c'1, c'2, c'3, and c'4 are the unstandardized regression coefficients given to the predictors in the proposed mediation model in the estimation of the outcomes, and el through el 7 denote the errors in estimation. The first and last regression analyses yielded direct effect and total effect, respectively, and the rest of the analyses indicated the indirect effect of self-directed learning readiness on affective learning outcomes (course satisfaction) and cognitive learning outcomes (Grade) (See Figures 1, 2, 3, and 4):

Figure 1

Model 1-1: The statistical model in which the effect of online self-directed learning readiness on grade is mediated by self-efficacy to learn statistics



Figure 2

Model 1-2: The statistical model in which the effect of online self-directed learning readiness on

grade is mediated by each construct of $2x^2$ achievement goal orientations



Figure 3

Model 2-1: The statistical model in which the effect of online self-directed learning readiness on course satisfaction is mediated by self-efficacy to learn statistics





Figure 4

Model 2-2: The statistical model in which the effect of online self-directed learning readiness on course satisfaction is mediated by each construct of 2x2 achievement goal orientations



Model 1-1



Self – efficacy to Learn Statistics = $\beta 0 + (a1 \times self - b)$

$$directed \ learning \ readiness) \ + \ e1 \tag{2}$$

 $Grade = \beta 0 + (c'1 \times self - directed learning readiness) + (b1 \times SELS) + e2$ (3)

Model 1-2

 $Mastery - Approach = \beta 0 + (a2 \times self - directed \ learning \ readiness) + e3$ (4) $Mastery - Avoidance = \beta 0 + (a3 \times self - directed \ learning \ readiness) + e4$ (5) $Performance - Approach = \beta 0 + (a4 \times self - directed \ learning \ readiness) + e5$ (6)

Performance – Avoidance = $\beta 0$ + (a5 × self – directed learning readiness) + e6 (7)

 $Grade = \beta 0 + (c'2 \times self - directed learning readiness) + (b2 \times MAP) +$

 $(b3 \times MAV) + (b4 \times PAP) + (b5 \times PAV) + e_7(8)$

Model 2-1

Course Satisfaction = $\beta 0$ + (c3 × self – directed learning readiness) + e8(9) Self – efficacy to Learn Statistics = $\beta 0$ + (a6 × self –

directed learning readiness) + e9 (10)

Course Satisfaction = $\beta 0$ + (c'3 × self – directed learning readiness) +

 $(b6 \times SELS) + e10$ (11)

Model 2-2

 $Mastery - Approach = \beta 0 + (a7 \times self - directed \ learning \ readiness) + e11 \ (12)$ $Mastery - Avoidance = \beta 0 + (a8 \times self - directed \ learning \ readiness) + e12 \ (13)$

Performance – Approach = $\beta 0 + (a9 \times self - directed learning readiness) + e13$ (14)

Performance – Avoidance = $\beta 0$ + (a10 × self – directed learning readiness) + e14 (15)

Course Satisfaction = $\beta 0$ + (c'4 × self - directed learning readiness) + (b7 × MAP) + (b8 × MAV) + (b9 × PAP) + (b10 × PAV) + e15(16)

The total effect of self-directed learning readiness on grades

The total effect of SDLR on grades is simply quantified and estimated with the regression coefficient c in the proposed mediation model shown in Equations 1. The interpretation of the total effect is that two cases that differ by one unit on SDLR are estimated to differ by c units on grade.

The total effect of self-directed learning readiness on course satisfaction

The total effect of SDLR on course satisfaction is simply quantified and estimated with the regression coefficient c in the proposed mediation model shown in Equations 9. The interpretation of the total effect is that two cases that differ by one unit on SDLR are estimated to differ by c units on course satisfaction.

The direct effect of self-directed learning readiness on grades

The direct effect of self-directed learning readiness on grade is quantified as c' and is interpreted as the extent to which two cases differ by one-unit self-directed learning readiness but are equal on self-efficacy to learn statistics (Equation 3), and 2x2 achievement goal orientations (Equation 8) are estimated to differ by c' units on grade.

The direct effect of self-directed learning readiness on course satisfaction

The direct effect of self-directed learning readiness on course satisfaction is quantified as c' and is interpreted as how the extent to which two cases differ by one-unit self-directed learning readiness but are equal on self-efficacy to learn statistics (Equation 11), and 2x2 achievement goal orientations (Equation 16) are estimated to differ by c' units on course satisfaction.

The indirect effect of self-directed learning readiness on grades

The effect of self-directed learning readiness on grades has yielded indirect effects in the proposed mediation model (see Figure 1 and 2). The indirect effect of self-directed learning readiness through self-efficacy to learn statistics only (Indirect effect $1 = a1 \times b1$; Equation 3), mastery-approach only (Indirect effect $2 = a2 \times b2$; Equation 8), mastery-avoidance only (Indirect effect $3 = a3 \times b3$; Equation 8), performance-approach only (Indirect effect $3 = a4 \times b4$; Equation 8) and performance-avoidance only (Indirect effect $4 = a5 \times b5$; Equation 8). Each indirect effect is used to test the research questions.

The indirect effect of self-directed learning readiness on course satisfaction

The effect of self-directed learning readiness on course satisfaction has yielded indirect effects in the proposed mediation model (see Figure 3 and 4). The indirect effect of self-directed learning readiness through self-efficacy to learn statistics only (Indirect effect $5 = a6 \times b6$; Equation 11), mastery-approach only (Indirect effect $6 = a7 \times b7$; Equation 16), masteryavoidance only (Indirect effect $7 = a8 \times b8$; Equation 16), performance-approach only (Indirect effect $8 = a9 \times b9$; Equation 16) and performance-avoidance only (Indirect effect $9 = a10 \times b10$; Equation 16). Each indirect effect is used to test the research questions.

Table 2

| Research Question | Model | Outcome | Mediator | Path Model |
|----------------------|-------|-----------------------------|-------------------|---|
| RQ 1.1 | 1-1 | | Direct effect | $SDL \rightarrow Grade$ |
| RQ 1.2 | 11 | Cognitive | | $SDL \rightarrow SELS \rightarrow Grade$ |
| RQ 1.3. | 1-2 | Learning Outcome (Grade) | Indirect Effect | $SDL \rightarrow MAP \rightarrow Grade$ $SDL \rightarrow MAV \rightarrow Grade$ $SDL \rightarrow PAP \rightarrow Grade$ |
| RQ 2.1 | 2-1 | | Direct effect | $SDL \rightarrow PAV \rightarrow Grade$ $SDL \rightarrow Satisfaction$ |
| RQ 2.2 | | Learning | Le l'ac et Effect | $SDL \rightarrow SELS \rightarrow Satisfaction$ $SDL \rightarrow MAP \rightarrow Grade$ |
| RQ 2.3. | 2-2 | Satisfaction) | Indirect Effect | $SDL \rightarrow MAV \rightarrow Grade$ $SDL \rightarrow PAP \rightarrow Grade$ $SDL \rightarrow PAV \rightarrow Grade$ |

Explanation of Variables of Models and Research Questions

The current study can use the bootstrapping method and beta coefficient to achieve its objective of understanding the statistical significance of the relationship between variables. Kim and Park (2019) defined statistical significance as the level at which the measure of the probability of the null hypothesis will be true or vary as per accepted standards. The beta coefficients estimate the strength and direction of the relationship between latent or observed variables and the associated errors. Indirect effects are calculated by acquiring the difference between total effects and direct effects. The direct effects are predictor variables of an outcome that do not account for the effects of the mediating variables in the relationship. For the current study, the beta coefficient will be determined as the product of the beta coefficient of online self-directed learning readiness and mediating variables (2x2 achievement goal orientations and self-efficacy to learn statistics). The effect of online self-directed learning readiness on grades and course satisfaction will yield indirect effects in the proposed mediation model.

Bootstrapping is a popular method to infer mediation (Shrout & Bolger, 2002) wherein the indirect effect is computed using a specified re-sampling method (e.g., 5000 iterations). This method generates a p-value, confidence intervals, and a standard error, which are used to interpret mediation. If the confidence interval does not include 0, one may conclude that the indirect effect differs from 0 and is statistically significant at the .05 level (Kenny, 2018). The model parameters are known as the unstandardized regression coefficients that are quantified and estimated using PROCESS version 4.3 with the 95% bootstrap confidence interval, as re-sampling methods implemented for the indexes of mediation inference (Hayes, 2013, 2018; Preacher & Hayes, 2008).

Bootstrapping is a widespread technique in research for several reasons. First, it does not require normalization assumptions. Unlike traditional statistical inference methods, such as t-tests or ANOVA, bootstrapping does not require the assumption of data normality. It creates a robust method for analyzing data that may not follow a normal distribution (Awang, Afthanorhan & Asri, 2015). It can also handle complex data structures. Bootstrapping can estimate the variability of statistics for complex data structures, such as clustered or longitudinal data, which can be challenging to analyze using traditional methods (Deen & de Rooij, 2020). It can also provide more accurate estimates. By generating numerous re-samples from the original data, bootstrapping can provide more accurate estimates of the variability of statistics compared to traditional methods (Awang et al., 2015). Bootstrapping is an alternative approach to null hypothesis testing that can be employed to assess the indirect effect and ascertain whether it differs significantly from zero (Hayes, 2013). When utilizing null hypothesis testing for an indirect effect, one assumption is that ab follows a normal distribution (i.e., if the study were repeated numerous times, resulting in the determination of ab for each iteration, the distribution of ab would be normal). However, since we lack knowledge about the true distribution of the indirect effect in the population, bootstrapping is

preferable as it doesn't rely on the assumption of normality for ab. Bootstrapping, as described by Hayes (2013), is a resampling technique. However, there are also some limitations to using bootstrapping in research. Firstly, generating numerous re-samples from the original data can be computationally intensive and may require large amounts of computing power and time (Bestgen & Vincze, 2012). Secondly, it may not work well for small sample sizes, as the re-samples may not adequately represent the population (Davidson & Flachaire, 2008). Accordingly, it could be susceptible to bias if the original data set is biased, as this bias may be propagated to the re-samples (Wehrens, Putter & Buydens, 2000).

Summary

This chapter provided a review of the methodology used to investigate the relationships between online learning readiness, self-efficacy to learn statistics, 2x2 achievement goal orientations, and learning outcomes such as course satisfaction and grade in the online statistics learning environment. The chapter covered the participants, data collection methods, instruments, and data analysis procedures. The population used in this study were students enrolled in a large research institution in the Southeastern U.S. during the Fall 2020 semester. The instruments used for data collection were a combination of AGQ-R, self-efficacy to learn statistics, and parts of the online learning readiness and course satisfaction scales. Bivariate correlation and path analysis were used to analyze quantitative data. The next chapter will discuss the findings and results of the study.

Chapter 4 - Findings

Overview

In this chapter, demographic findings, statistical assumptions, and preliminary analyses including reliability, descriptive findings, and correlations between measures and findings from the data analysis will be presented. The results and findings for each research question are described along with the tables and figures from the data analysis.

Problem Statement

The COVID-19 crisis has led to drastic changes in the education system. Students were forced to shift from physical to online classes in response to government restrictions on face-to-face interactions. These restrictions were considered critical to preventing the spread of respiratory disease. Aguilera-Hermida (2020) explains that while numerous challenges accompanied these changes, they created an awareness of the role and advantages of engaging in online learning. Therefore, even after the risk of spread was reduced, the popularity of online learning remained high. Bashir et al. (2021) reported that Aston University adopted a hybrid mode of course delivery (combining online and face-to-face courses) as a post-COVID solution that recognizes the popularity of online learning. A survey cited by Kelly (2021) established that 73% of students preferred studying fully online after the pandemic; however, only about 53% of faculty preferred teaching online.

Statistics courses have also been transferred online, with studies such as Ritzhaupt et al. (2020) revealing that students enrolled in the coursework experience less anxiety, making the module preferable to in-person learning. The popularity of online statistics courses postpandemic has created a demand for understanding this module of learning and how learners and educators can organize the course effectively. Further demand for understanding the learning module has been created by studies such as those by Figueroa-Cañas and Sancho-Vinuesa (2020), which have reported that the dropout rate of online statistics courses is higher than that of traditional statistics courses. This is a significant concern because, even though most learners prefer online learning, this same online model has seen higher dropout rates. The current study covers the existing gap in the literature regarding the effects of self-efficacy, self-directed learning readiness (SDLR), and goal orientation on learning outcomes for learning statistics online.

Research Questions

1. What is the extent of the relationship between online self-directed learning readiness and grades mediated by self-efficacy to learn statistics and each construct of achievement goal orientations in an online statistic learning environment?

1.1 What is the extent of the relationship between online self-directed learning readiness and grades in an online statistic learning environment?

1.2 What is the extent of the relationship between online self-directed learning readiness and grades mediated by self-efficacy to learn statistics in an online statistic learning environment?

1.3 What is the extent of the relationship between online self-directed learning readiness and grades mediated by each construct of achievement goal orientations (MAP, MAV, PAP, PAV) in an online statistic learning environment?

2. What is the extent of the relationship between online self-directed learning readiness and course satisfaction mediated by self-efficacy to learn statistics and each construct of achievement goal orientations in an online statistic learning environment?

2.1 What is the extent of the relationship between online self-directed learning readiness and course satisfaction in an online statistic learning environment?

2.2 What is the extent of the relationship between online self-directed learning readiness and course satisfaction mediated by self-efficacy to learn statistics in an online statistic learning environment?

2.3 What is the extent of the relationship between online self-directed learning readiness and course satisfaction mediated by each construct of achievement goal orientations in an online statistic learning environment?

Demographic Findings

A total of 168 students participated and completed the survey. Among the total participants, 47 students were removed because of the high level (above 10%) of missing data (Hair, Black, Babin, & Anderson, 2010). After excluding these respondents, the data analyzed in the present study was drawn from a valid sample of 121 participants. Table 2 presents the demographic characteristics of a valid sample of participants, including age, gender, ethnicity, international/domestic students, degree level, stem major, and final grade. Participants' ages ranged from 18 to 57 (M=25.31, SD=8.72). More than 66.9% of students (n=81) were reported as female, and 33.1% of participants (n=41) were reported as male. Seventy-four of the participants (61.2%) were undergraduate students, and 47 of the participants (36.8%) were graduate students.

Table 3

| Demographic Gender | Category | Frequency | Percentage |
|-----------------------|-----------------------|-----------|------------|
| | Male | 40 | 33.1% |
| | Female | 81 | 66.9% |
| Age | | | |
| | 17-24 | 81 | 66.9% |
| | 25-34 | 25 | 20.7% |
| | 35-44 | 7 | 5.8% |
| | 44~ | 8 | 6.6% |
| Ethnicity | | | |
| | White | 90 | 74.4% |
| | African American | 8 | 6.6% |
| | Asian | 11 | 9.1% |
| | Hispanic | 9 | 7.4% |
| | Other | 3 | 2.5% |
| International | | | |
| | Domestic Student | 103 | 85.1% |
| | International Student | 18 | 14.9% |
| Degree Level | | | |
| | Undergraduate | 74 | 61.2% |
| | Graduate | 47 | 38.8% |
| Stem Major | | | |
| | STEM | 93 | 76.9% |
| | Non-STEM | 28 | 23.1% |
| Final Grade | | | |
| | А | 68 | 56.2% |
| | В | 33 | 27.3% |
| | С | 17 | 14% |
| | D | 1 | 0.8% |
| | F | 2 | 1.7% |

Descriptive Statistics for Study Variables (N=121)

Measures of Reliability

Using the Cronbach Coefficient Alpha test, the results of the tests for online self-directed learning readiness, self-efficacy to learn statistics, 2x2 achievement goal orientations, and course satisfaction are presented in Table 3. A value of 0.70 or higher was considered evidence of very

good reliability, a value between 0.6 and 0.7 is acceptable, a value between 0.5 and 0.6 is considered poor reliability, and a value below 0.5 is unacceptable (Becker, 2000). The value of Cronbach's alpha for online self-directed learning readiness (OSLR) is 0.61, and for self-efficacy to learn statistics (SELS) is 0.79. The values of Cronbach's alpha for mastery-approach goal orientation (MAP), mastery-avoidance goal orientation (MAV), performance-approach goal orientation (PAP), and performance-avoidance goal orientation (PAV) were 0.65, 0.85, 0.83 and 0.77, respectively. The value of Cronbach's alpha for course satisfaction (CS) was 0.81. According to Becker (2000), all Cronbach's alpha values of variables in this study were acceptable to use.

Table 4

| | Ν | Cronbach's α | Previously reported Cronbach's α |
|-------------------------------|----|--------------|--|
| OSDLR | 5 | 0.61 | 0.72 |
| SELS | 14 | 0.80 | 0.97 |
| Achievement goal orientations | | | |
| MAP | 3 | 0.65 | 0.84 |
| MAV | 3 | 0.85 | 0.88 |
| PAP | 3 | 0.83 | 0.92 |
| PAV | 3 | 0.77 | 0.94 |
| Course satisfaction | 5 | 0.81 | 0.79 |

Reliability for Study Variables

Statistical Assumptions

Prior to performing an OLS regression-based path analysis, statistical assumptions including linearity, normality, independent errors, homoscedasticity, normally distributed errors, multicollinearity, and multivariate outliers were tested.

Linearity

The linearity assumption is one of the most important assumptions to be satisfied in multivariate analysis methods. In order to run path analysis, one of the assumptions is that any predictor should be linearly associated with the outcome variable (Field, 2013). The linearity assumption was checked with the matrix scatterplots between each variable. The result of scatterplots for each predictor variable in this study represents a linear pattern for the outcome variable. Therefore, we can conclude that the inspection of the scatterplots did not reveal any evidence of nonlinear relationships between the variables, indicating that the linearity assumption is reasonable (see Figure 6).

Figure 5

Matrix scatterplot of the relationships between self-directed learning readiness, each construct of 2x2 achievement goal orientations, self-efficacy to learn statistics, and reported grade.



Normality

Skewness and kurtosis were used to determine whether the data was normally distributed or not. An absolute skew value larger than 2 or smaller than -2 or an absolute kurtosis value larger than 7 or smaller than -7 may indicate non-normality (Bryne, 2010; Hair et al, 2010). According to Table 3, normality testing results represented the range for kurtosis between -.895 and .479 and the range for skewness between -1.154 and -.283. Both values of skewness and kurtosis demonstrated that the shape of the data distribution for each variable in the research is acceptable (Kline, 2011; Tabachnick & Fidell, 2013)

Table 5

| Variable | Skewness | Kurtosis |
|--------------|----------|----------|
| OSLDR | -0.60 | 0.09 |
| SELS | -0.33 | 0.54 |
| MAP | -1.15 | 0.48 |

-1.02

-0.87

-0.60

-0.28

Summary of Normality Statistics

Independence of Errors

MAV

PAP

PAV

Course Satisfaction

The assumption of independence of errors should be met when the errors of any two observations are not correlated to each other. The Durbin-Watson test was used to examine the assumption of independent errors. The Durbin-Watson value between 1.5 and 2.5 indicates that the independent error assumption is tenable (Neter et al., 1996). According to Field (2013), the conservative criterion of the assumption of independent errors is that Durbin-Watson values "less than 1 or greater than 3 should definitely raise alarm bells" (p. 337). The Durbin-Watson

0.18

0.38

-0.28 -0.90 statistics were computed to be 1.79 and 2.13, which is within the acceptable range. Therefore, the assumption of independent errors has been satisfied.

Homoscedasticity

The assumption of homoscedasticity states that the outcome variable should exhibit the same levels of variance at each level of each predictor (Hair et al., 2010). Violating this assumption indicates heteroscedasticity. Homoscedasticity was investigated graphically by plotting the standardized predicted values (ZPRED) against the standardized residuals (ZRESID). If the graph looks like a random array of scatter plots, this is indicative of a situation in which the homoscedasticity assumption has been met. Based on the scatterplot of ZPRED vs. ZRESID (see Figure 7), the assumption of homoscedasticity is not violated because it presents a random pattern. Consequently, the variance of the course satisfaction values is relatively equal across the range of predictor(s), thereby indicating homoscedasticity (Field, 2013; Hair et al., 2006).

Figure 6



Scatterplot of ZRESID vs. ZPRED

Normally distributed errors

The assumption of normally distributed errors examines whether the residuals in the model are normally distributed, which means that there are no differences between the observed data (little dots) and the model (diagonal line). To test the assumption, the P-P plot of the regression standardized residual plot was used (see Figure 8). Ideally, this plot should look like the little dots (observed data) and should follow the normality line. Thus, it can be concluded that the residual is normally distributed.

Figure 7





Multicollinearity

The assumption of multicollinearity is to be met by having no perfect multicollinearity between predictors (i.e., two predictors) or among predictors (i.e., more than two predictors). Ideally, the predictor variables in the model should be highly correlated with the outcome variable but have zero or little correlation between or among themselves. Multicollinearity indicates that the predictors actually have shared variance with each other, which challenges the ability to predict and explain the outcome as well as discover the relative importance of each predictor in the model. I used VIF to assess multicollinearity, applying the guidelines provided by Field (2013), where VIF values of up to 5 are acceptable levels of multicollinearity. To check this assumption, I ran a multiple linear regression with course satisfaction and grade as outcome variables and OSDLR, SELS, and 2x2 achievement goal orientations as predictors. Table 4 indicates the summary of multicollinearity statistics for this research. The VIF's values ranged from 1.12 (SELS) to 1.46 (PAV). Given that none of the VIF values have violated the cutoff value of 5, I can conclude that multicollinearity is not a concern for the research model.

Table 6

| Summary of | ^c Multicollinearity | Statistics | N = 1 | 121) |
|------------|--------------------------------|------------|-------|------|
| | | | 1 | |

| Variable | VIF |
|----------|------|
| OSDLR | 1.18 |
| SELS | 1.12 |
| MAP | 1.36 |
| MAV | 1.29 |
| PAP | 1.41 |
| PAV | 1.46 |

Multivariate Outliers

The OLS regression works best when the assumption of having no multivariate outliers is met. Outliers can be very influential in correlation and, therefore, regression. A thorough initial data analysis should be used to review the data and identify outliers (both univariate and multivariate). Mahalanobis distance, which describes the distance of a case from the centroid of the remaining cases, where the centroid is the point created at the intersection of the means of all the variables, was used to detect multivariate outliers. Chi-square tests of significance are usually used to determine outliers.

The Mahalanobis distance (MD) on p of variables is used as a multivariate outlier detection measure and is compared with a chi-square distribution on p degrees of freedom (Rousseeuw and Van Zomeren, 1990; Samson, 2014). For a p-dimensional vector, x(i), on observation i with a corresponding mean vector, mean, and a sample covariance matrix, C, we have (see equation 19):

$$MD(i) = Square Root of (x(i) - mean)TC - 1 (x(i) - mean)$$
(19)

If MD(i) is greater than $\chi^2(p, 0.999)$ then this suggests that observation i is an outlier. This high level of significance, which corresponds to p<0.001, is recommended (Tabachnich and Fidell, 1996; Hair et al, 1998).

Mahalonobis distance was calculated from the SPSS, and a chi-square distribution with the same degrees of freedom was compared. The degrees of freedom correspond to the number of variables that were grouped together to calculate the Mahalanobis distance. The Mahalonobis distance was represented by MAH_1, and the p-value of the right-tail of the chi-square distribution p-value was calculated by the expression 1 – CDF.CHISQ(X1, X2); X1 was substituted with the Mahalanobis distance variable that was created from the regression, and X2 was substituted with the degrees of freedom—which corresponds to the number of variables being examined. There was no observation with the values of the new probability variable less than 0.001; thus, I concluded that this data was usable without any multivariate outliers.

Preliminary Analyses

Descriptive Findings and Correlations among Measures

Table 5 displays the descriptive statistics of means and standard deviations and the Pearson correlation coefficient of the variables in this research except for grade. Grades were reported as an ordinal variable; thus, the correlation between grades and other variables in this study was measured with Kendall's coefficient of rank correlation, τ b, which is represented in the 8th row of the table below.

The results of the bivariate correlations showed that online self-directed learning readiness is positively related to self-efficacy (r = 0.19, p < 0.05), mastery-approach goal orientation (r = 0.31, p < 0.001), and course satisfaction (r = 0.35, p < 0.001). Moreover, mastery-approach goal orientation (r = 0.20, p < 0.05) and self-efficacy to learn statistics (r = 0.24, p < 0.001) are positively correlated with course satisfaction and master-approach goal orientation is positively associated with grade ($\tau b = 0.24 p < 0.001$). On the other hand, performance-avoidance goal orientation is negatively correlated with grade ($\tau b = -0.14$, p < 0.05).

Table 7

| | 1 | 2 | 3 | 4 | 5 | 6 | 7 | 8 |
|--|-------|-------|-------|-------|-------|------|-------|------|
| 1. Online self-directed learning readiness | | | | | | | | |
| 2. Self-efficacy to learn statistics | .19* | | | | | | | |
| 3. Mastery-approach | .31** | .21* | | | | | | |
| 4. Mastery-avoidance | 09 | .03 | .29** | | | | | |
| 5. Performance-approach | 07 | 12 | .19* | .26** | | | | |
| 6. Performance-avoidance | 14 | .05 | .001 | .35** | .46** | | | |
| 7. Course satisfaction | .36** | .24** | .20* | .02 | 11 | .02 | | |
| 8. Grade | .14 | .03 | .24** | 08 | .03 | 14* | .21** | |
| Mean | 3.89 | 3.99 | 6.34 | 5.78 | 5.11 | 4.86 | 3.43 | 4.35 |
| SD | .77 | 1.11 | .76 | 1.26 | 1.53 | 1.65 | 1.16 | .87 |

Intercorrelations for Study Variables

Results

This section reports the results of the assessment of the proposed mediation model in the current study. This hypothesized model was estimated using an ordinary least squares (OLS) regression-based path analysis with the PROCESS macro for SPSS (Hayes, 2013, 2018; Preacher & Hayes, 2008).

OLS Regression

The proposed mediation model in Figures 2, 3, 4 and 5 from Chapter 3 has four different models. Each model has one total effect, one direct effect, and one indirect effect for mediator SELS and four indirect effects for mediator 2x2 achievement goal orientations of OSDRS on cognitive and affective learning outcomes, such as grade and course satisfaction. The focus of this study is to specify the indirect effects of each model, passing through a mediator(s) of: (a) SELS between OSDLR and grade; (b) 2x2 achievement goal orientations between OSDLR and grade; (c) SELS between OSDLR and course satisfaction; and (d) 2x2 achievement goal orientations between questions. In addition to the estimation of the indirect effects, the parameters *c* and *c'* estimate the total effect and direct effect of OSDLR on cognitive and affective learning outcomes, respectively. The model parameters are known as the unstandardized regression coefficients that are quantified and estimated using PROCESS version 4.3 with the 95% bootstrap confidence interval as a resampling method implemented for the indexes of mediation inference (Hayes, 2013, 2018; Preacher & Hayes, 2008).

Research Questions 1.1 and 1.2

Research question 1.1 for the current study asked: What is the extent of the relationship between online self-directed learning readiness and cognitive learning outcomes (grades) in an online statistic learning environment? Research question 1.2 asked: What is the extent of the relationship between online self-directed learning readiness and cognitive learning outcomes (grades) mediated by self-efficacy to learn statistics in an online statistic learning environment? The primary intention of this question is to test whether self-efficacy was a potential mediator of the relationship between online self-directed learning readiness and cognitive learning outcomes. This was examined by measuring the indirect effect, which was quantified as the product of the effect of online self-directed learning readiness (a1) and the effect of self-efficacy to learn statistics (b1).

The results of the mediation model analysis for Research Questions 1.1 and 1.2 are presented in Table 8. As shown in Table 8, self-directed learning readiness was positively related to self-efficacy to learn statistics (B= 0.28, SE = 0.12, t (119) = 2.17, p = 0.03), but there is no evidence of an association between self-efficacy to learn statistics and cognitive learning outcome (grade) when controlling for SDL (B = 0.05, SE = 0.07, t (119) = 0.64, p = 0.52). As can be seen in Table 8, a_1 = 0.28, and b_1 = 0.05. Therefore, multiplying a_1 and b_1 does not yield any significant indirect effect, where $a_1 \times b_1$ = (0.28) (0.04) = 0.0129 (see Table 9). A resampling method using 5,000 bootstrap samples was applied to generate a 95% bias-corrected bootstrap confidence interval as the inferential test for indirect effect ($a_1 \times b_1$ = 0.01), and this was found to be around zero (-0.03 to 0.07). The first indirect effect is statistically non-significant because the confidence interval includes zero (Preacher & Hayes, 2008). Therefore, the results of this analysis do not support research questions 1.1 and 1.2, which assume that self-efficacy to learn statistics mediates the relation between online self-directed learning readiness and grade.

Table 8

Model Summary Information for the Proposed Mediation Model Portrayed in Figure 1 (N = 121)

| | | | | O | utput | | | |
|------------|-------------------|-----------------|-----------|------------------|-------------------|----------------|--------|----------------|
| Predictors | - | SELS | | | C | brade | | |
| | - | $B(\beta)$ | SE | р | | $B(\beta)$ | SE | р |
| OSDLR | $a_1 \rightarrow$ | 0.28 (.19) | 0.12 | 0.03 | <i>c'</i> 1 | 0.24 (0.21) | 0.10 | 0.04 |
| SELS | | | | | $b_1 \rightarrow$ | 0.05(0.06) | 0.07 | 0.52 |
| Constant | | 2.91 | 0.51 | < 0.001 | | 3.25 | 0.45 | < 0.001 |
| | | $R^2 = 0.04, F$ | F(1, 119) | = 4.72, <i>p</i> | | $R^2 = .05, H$ | 7(1,11 | 8), <i>p</i> = |
| | | | =0.03 | | | | 0.04 | |

Note. B = Unstandardized regression coefficient; β = Standardized regression coefficient.

Table 9

Summary of the Mediation Model 1-1 Analysis

| | | | | | 95% | ó CI |
|----------|----------|------------------|----------|------|-------|------|
| Effect | Mediator | Equation | Estimate | SE | LL | UL |
| Indirect | SELS | $a_1 \times b_1$ | 0.01 | 0.02 | -0.03 | 0.07 |
| Direct | | c_{I}' | 0.24 | 0.10 | 0.03 | 0.44 |
| Total | | c_{l} | 0.25 | 0.10 | 0.05 | 0.45 |

Note. CI = confidence interval; LL = lower limit; UL = upper limit. If the confidence interval does not include zero, it reveals a significant effect.

Research Question 1.3

Research question 1.3 for the present study was: what is the extent of the relationship between online self-directed learning readiness and cognitive learning outcome (grade) in an online statistics learning environment mediated by each construct of achievement goal orientations (MAP, MAV, PAP, PAV)? The primary intention was to test whether each construct of achievement Goal Orientations (MAP, MAV, PAP, PAV) was a potential mediator of the relation between online self-directed learning readiness and cognitive learning outcome. This was examined by an indirect effect, which was quantified as the product of the effect of online self-directed learning readiness (a_2) and the effect of MAP (b_2) , MAV (b_3) , PAP (b_4) , and PAV goal orientations (b_5) .

The results of the mediation model analysis are presented in Table 10. As shown in Table 10, self-directed learning readiness was positively related to MAP (B = 0.30, SE = 0.09, t (119) = 3.52, p < 0.001), and MAP was also positively associated with cognitive learning outcomes (grades) when controlling for SDLR (B = 0.45, SE = 0.11, t (119) = 4.05, p < 0.001). As can be seen in Table 10, $a_2 = 0.30$, and $b_2 = 0.45$. Therefore, multiplying a_2 and b_2 yielded the indirect effect of $a_2b2 = (0.30)$ (0.45) =0.13 (see Table 11). A resampling method using 5,000 bootstrap samples was applied to generate a 95% bias-corrected bootstrap confidence interval as the inferential test for indirect effect ($a_2b_2 = 0.13$), and this was found to be totally above zero (0.04 to 0.27). The indirect effect is statistically significant because the confidence interval does not include zero (Preacher & Hayes, 2008). Therefore, the result of this analysis supports research question 1.3, which assumes that MAP mediates the relationship between online self-directed learning readiness and cognitive learning outcomes (grades).

The results of the mediation model analysis for MAV are presented in Table 10. As shown in Table 8, self-directed learning readiness was not related to MAV (B = -0.14, SE = 0.15, t (119) = -0.96, p = 0.33, and there is evidence of an association of MAV with cognitive learning outcomes (grades) controlling for SDLR (B = -0.17, SE = 0.07, t (119) = -2.50, p = 0.01). As can be seen in Table 10, $a_3 = -0.14$, and $b_3 = -0.17$. Therefore, multiplying a_3 and b_3 does not yield any significant indirect effect; $a_3b_3 = (-0.14)$ (-0.17) = 0.02 (see Table 11). A resampling method using 5,000 bootstrap samples was applied to generate a 95% bias-corrected bootstrap confidence interval as the inferential test for indirect effect ($a_3b_3 = 0.02$), and this was found to involve zero (-0.02 to 0.08). The indirect effect is statistically non-significant because the confidence interval includes zero (Preacher & Hayes, 2008). Therefore, the results of this analysis do not support research question 1.3, which assumes that MAV mediates the relationship between online self-directed learning readiness and cognitive learning outcomes (grades).

The results of the mediation model analysis for PAP are presented in Table 10. As shown in Table 10, self-directed learning readiness was not related to PAP (B = -0.13, SE = 0.18, t (119) = -0.77, p = 0.44), and there is no evidence of an association of PAP with cognitive learning outcomes (grades) when controlling for SDLR (B = -0.00, SE = 0.06, t (119) = -0.01, p = 0.99). As can be seen in Table 10, $a_4 = -0.13$ and $b_4 = -0.00$. Therefore, multiplying a4 and b4 does not yield any significant indirect effect; $a_4b_4 = (-0.00)(-0.00) = 0.00$ (see Table 9). A resampling method using 5,000 bootstrap samples was applied to generate a 95% bias-corrected bootstrap confidence interval as the inferential test for indirect effect ($a_4b_4 = 0.00$), and this was found to involve zero (-0.02 to 0.03). The indirect effect is non-significant because the confidence interval includes zero (Preacher & Hayes, 2008). Therefore, the results of this analysis do not support research question 1.3, which assumes that PAP mediates the relationship between online self-directed learning readiness and cognitive learning outcomes (grades). The results of the mediation model analysis for PAV are presented in Table 8. As shown in Table 10, self-directed learning readiness was not significantly related to performance-avoidance (B = -0.30, SE = 0.19, t (119) = -1.55, p = 0.12), and there is also no evidence of an association between PAV and cognitive learning outcomes (grades) when controlling for SDLR (B = -0.03, SE = 0.05, t (119) = -0.58, p = 0.12). As can be seen in Table 10, $a_5 = -0.30$, and $b_5 = -0.03$. Therefore, multiplying a5 and b5 does not yield any significant indirect effect; $a_5b_5 = -0.30$ (-.017) = 0.01 (see Table 11). A resampling method using 5,000 bootstrap samples was applied to generate a 95% bias-corrected bootstrap confidence

interval as the inferential test for indirect effect ($a_5b_5 = 0.01$), and this was found to involve zero (-0.02 to 0.06). The indirect effect is non-significant because the confidence interval includes zero (Preacher & Hayes, 2008); therefore, the results of this analysis do not support research question 1.3, which assumes that PAV mediates the relationship between online self-directed learning readiness and cognitive learning outcomes (grades).

Models 1-2 were statistically significant, or the total indirect effect was significant, as can be seen from Tables 10 and 11. Adding the product of $a_2 \times b_2$, $a_3 \times b_3$, $a_4 \times b_4$, and $a_5 \times b_5$ yielded a statistically significant total indirect effect; $a_2 \times b_2 + a_3 \times b_3 + a_4 \times b_4 + a_5 \times b_5$; 0.17, and this was found not to involve zero (0.06 to 0.30). The overall indirect effect is statistically significant because the confidence interval does not include zero (Preacher & Hayes, 2008).

Table 10

Outcome Predictors MAP(M 1)MAV (M 2)PAP (M 3) PAV (M 4) Grade $B(\beta)$ SE SE SE SE $B(\beta)$ SE $B(\beta)$ $B(\beta)$ $B(\beta)$ р р р р p -0.30 -0.13 .30 0.08 *a*³ -. 14 a_4 a_5 OSDLR 0.09 <.001 0.15 0.33 0.18 0.44 0.19 0.12 $c_2 \rightarrow$ (-(-0.10 0.42 $a_2 \rightarrow$ (.31) → (-.09) (0.07) \rightarrow \rightarrow 0.07)0.14)0.45 < MAP $b_2 \rightarrow$ 0.11 (0.39)0.001 -0.17 MAV (-0.07 0.01 $b_3 \rightarrow$ 0.24)-0.00 b_4 0.06 PAP 0.99 (-.00) \rightarrow -0.03 b_5 PAV 0.05 0.56 (- \rightarrow 0.06)<Constant 5.18 0.34 < .001 6.34 0.59 <.001 5.65 0.71 <.001 6.02 0.76 <.0001 2.30 0.70 0.01 $R^2 = 0.19, F$ $R^2 = 0.09, F(1,$ $R^2 = 0.01, F(1,$ $R^2 = 0.01, F(1)$ $R^2 = 0.02, F(1,$ (1, 115)(119) = 12.42, p < 100(119) = 0.93, p =(119) = 0.59, p =(119) = 2.41, p ==5.51, *p* < 0.001 0.34 0.12 0.44 0.001

| | Model Summary | Information | for the Pro | posed Mediation Mod | lel Portrayed in | <i>Figure 2</i> ($N = 121$) |
|--|---------------|-------------|-------------|---------------------|------------------|-------------------------------|
|--|---------------|-------------|-------------|---------------------|------------------|-------------------------------|

Note. B = Unstandardized regression coefficient; β = Standardized regression coefficient.
Table 11

| | | | | | 95% CI | |
|----------------|--------------|--|----------|------|--------|------|
| Effect | Mediat or | Equation | Estimate | SE | LL | UL |
| Indirect | MAP | $a_2 \times b_2$ | 0.13 | 0.06 | 0.04 | 0.27 |
| Indirect | MAV | $a_3 \times b_3$ | 0.02 | 0.02 | -0.02 | 0.08 |
| Indirect | PAP | $a_4 \!$ | 0.00 | 0.01 | -0.02 | 0.03 |
| Indirect | PAV | $a_5 \times b_5$ | 0.01 | 0.02 | -0.02 | 0.06 |
| Total indirect | | $a_2 \times b_2 + a_3 \times b_3 + a_4 \times b_4 + a_5 \times b_5$ | 0.17 | 0.06 | 0.06 | 0.30 |
| Direct | | c_2 ' | 0.08 | 0.10 | -0.12 | 0.28 |
| Total | | C_2 | 0.25 | 0.10 | 0.05 | 0.45 |

Summary of the Mediation Model 1-2 Analysis

Note. CI = confidence interval; LL = lower limit; UL = upper limit. If the confidence interval does not include zero, it reveals a significant effect.

Research Questions 2.1 and 2.2

Research questions 2.1 and 2.2 asked: what is the extent of the relationship between online self-directed learning readiness and affective learning outcomes (course satisfaction) mediated by self-efficacy to learn statistics in an online statistic learning environment? The primary intention of the question was to test whether self-efficacy was a potential mediator of the relationship between online self-directed learning readiness and course satisfaction. This was examined by an indirect effect, which was quantified as the product of the effect of online self-directed learning readiness (a_6) and the effect of self-efficacy to learn statistics (b_6).

The results of the mediation model analysis are presented in Table 12, which shows that self-directed learning readiness is related positively to self-efficacy to learn statistics (B = 0.28, SE = 0.12, t (119) = 2.17, p = .035) and self-efficacy to learn statistics is associated with affective learning outcomes (course satisfaction) when controlling for SDLR (B = 0.18, SE = 0.09, t (119) = 2.03, p = 0.04). As can be seen in Table 12, $a_6 = 0.28$, and $b_6 = 0.18$. However, multiplying a_6 and b_6 does not yield any significant indirect effect; $a_6b_6 = (0.28)$ (0.18) = 0.05 (see Table 13). A

resampling method using 5,000 bootstrap samples was applied to generate a 95% bias-corrected bootstrap confidence interval as the inferential test for indirect effect ($a_6b_6 = 0.05$), and this was found to be around zero (-0.01 to 0.14). This indirect effect is non-significant because the confidence interval includes zero (Preacher & Hayes, 2008). Therefore, the results of this analysis do not support research questions 2.1 and 2.2, which assume that SELS mediates the relationship between online self-directed learning readiness and affective learning outcomes (course satisfaction).

Table 12

Model Summary Information for the Proposed Mediation Model Portrayed in Figure 3 (N =

121)

| | а.: | | |
|-------------------|-------------------------------|--|---|
| | Sati | sfaction | |
| | $B(\beta)$ | SE | Р |
| C3' | 0.48 (0.17) | 0.12 | < 0.001 |
| $b_6 \rightarrow$ | 0.18 (0.32) | 0.09 | 0.04 |
| | 0.83 | 0.57 | 0.14 |
| | $R^2 = .15, F(1, 0)$ | (118) = 10 |).84, <i>p</i> < |
| | $c_{3'}$ $b_6 \rightarrow$ | $B(\beta)$ $c_{3'} = 0.48 (0.17)$ $b_{6} \rightarrow = 0.18 (0.32)$ 0.83 $R^{2}=.15, F (1, 0)$ | $ \begin{array}{cccccccccccccccccccccccccccccccccccc$ |

Note. B = Unstandardized regression coefficient; β = Standardized regression coefficient.

Table 13

Summary of the Mediation Model 2-1 Analysis

| | | | | | 95% | ∕₀ CI |
|----------|----------|-----------------------|----------|------|-------|-------|
| Effect | Mediator | Equation | Estimate | SE | LL | UL |
| Indirect | SELS | $a_6 \! 	imes \! b_6$ | 0.05 | 0.04 | -0.01 | 0.14 |
| Direct | | С | 0.48 | 0.13 | 0.22 | 0.74 |
| Total | | C3 | 0.53 | 0.13 | 0.28 | 0.79 |

Note. CI = confidence interval; LL = lower limit; UL = upper limit. If the confidence interval does not include zero, it reveals a significant effect.

Research Question 2.3

Research question 2.3 for the present study asked: what is the extent of the relationship between online self-directed learning readiness and affective learning outcomes (course satisfaction) mediated by each construct of 2x2 achievement goal orientations (MAP, MAV, PAP, PAV) in an online statistic learning environment? The primary intention was to test whether each construct of 2x2 achievement goal orientations (MAP, MAV, PAP, PAV) is a potential mediator of the relation between online self-directed learning readiness and course satisfaction. This was examined by an indirect effect, which was quantified as the product of the effect of online selfdirected learning readiness (a_7) and MAP goal orientation (b_7), MAV goal orientation (b_8), PAP goal orientation (b_9), and PAV goal orientation (b_{10}).

The results of the mediation model analysis are shown in Table 14, which reveals that selfdirected learning readiness is positively related to MAP (B = 0.30, SE = 0.09, t (119) = 3.52, p < 0.001), and there is no evidence of an association between MAP and affective learning outcomes (course satisfaction) when controlling for SDLR (B = 0.21, SE = 0.15, t (119) = 1.38, p = 0.17). As can be seen in Table 14, $a_7 = 0.30$, and $b_7 = 0.21$. Therefore, multiplying a_7 and b_7 does not yield the indirect effect, $a_7b_7 = 0.30(.21) = 0.06$ (see Table 15). A resampling method using 5,000 bootstrap samples was applied to generate a 95% bias-corrected bootstrap confidence interval as the inferential test for indirect effect ($a_7b_7 = 0.06$), and this was found to involve zero (-0.02 to 0.18). The indirect effect is non-significant because the confidence interval includes zero (Preacher & Hayes, 2008). Therefore, the results of this analysis do not support research question 2.3, which assumes that MAP mediates the relationship between online self-directed learning readiness and affective learning outcomes (course satisfaction). As shown in Table 12, analysis did not provide evidence of a relationship between selfdirected learning readiness and MAV (B = -0.14, SE = 0.15, t (119) = -0.96, p = 0.33), and MAV is also not associated with affective learning outcomes (course satisfaction) when controlling for SDLR (B = 0.01, SE = 0.09, t (119) = 0.10, p = 0.92). As can be seen in Table 14, $a_8 = -0.14$ and b8=0.01. Therefore, multiplying a8 and b8 does not yield any significant indirect effect; a8b8 = -0.14 (.01) = -0.00 (see Table 15). A resampling method using 5,000 bootstrap samples was applied to generate a 95% bias-corrected bootstrap confidence interval as the inferential test for indirect effect (a8b8= -0.0013), and this was found to involve zero (-0.04 to 0.03). The indirect effect is nonsignificant because the confidence interval includes zero (Preacher & Hayes, 2008). Therefore, the results of this analysis do not support research question 2.3, which assumes that MAV mediates the relationship between online self-directed learning readiness and affective learning outcome (course satisfaction).

As shown in Table 14, self-directed learning readiness is not related to PAP (B = -0.13, SE = 0.18, t (119) = -0.77, p = 0.44), and there is no evidence of an association between PAP and affective learning outcomes (course satisfaction) when controlling for SDLR (B = -0.14, SE = 0.07, t (119) = -1.9, p = 0.06). As can be seen in Table 14, $a_9 = -0.13$ and $b_9 = -0.14$. Therefore, multiplying a9 and b9 does not yield any significant indirect effect; $a_9b_9 = -0.13$ (-0.14) = 0.02 (see Table 15). A resampling method using 5,000 bootstrap samples was applied to generate a 95% bias-corrected bootstrap confidence interval as the inferential test for indirect effect ($a_9b_9 = 0.02$), and this was found to involve zero (-0.02 to 0.09). The indirect effect is non-significant because the confidence interval includes zero (Preacher & Hayes, 2008). Therefore, the results of this analysis do not support research question 2.3, which assumes that PAP mediates the relationship

between online self-directed learning readiness and affective learning outcomes (course satisfaction).

As shown in Table 14, self-directed learning readiness is not related to PAV (B = -0.30, SE = 0.19, t (119) = -1.55, p = 0.12), and there is no evidence of an association between PAV and affective learning outcomes (course satisfaction) when controlling for SDLR (B = 0.11, SE = 0.07, t (119) = 1.47, p = 0.14). As can be seen in Table 12, $a_{10} = -0.30$, and $b_{10} = 0.11$. Therefore, multiplying a10 and b10 does not yield any significant indirect effect; $a_{10} b_{10} = -0.30$ (-.11) = -0.03 (see Table 15). A resampling method using 5,000 bootstrap samples was applied to generate a 95% bias-corrected bootstrap confidence interval as the inferential test for indirect effect ($a_{10} b_{10} = -0.03$), and this was found to involve zero (-0.10 to 0.02). The indirect effect is non-significant because the confidence interval includes zero (Preacher & Hayes, 2008). Therefore, the results of this analysis do not support research question 2.3, which assumes that PAV mediates the relationship between online self-directed learning readiness and affective learning outcomes (course satisfaction).

The results show that the model was non-significant, or the total indirect effect is nonsignificant, as can be seen from Tables 14 and 15. Adding the product of $a_7 \times b_7$, $a_8 \times b_8$, $a_9 \times b_9$, and $a_{10} \times b_{10}$ does not yield a statistically significant total indirect effect, where $a_7 \times b_7$, $a_8 \times b_8$, $a_9 \times b_9$, $\times b_9$, and $a_{10} \times b_{10}$; 0.05, and this was found to involve zero (-0.06 to 0.19). The overall indirect effect is non-significant because the confidence interval includes zero (Preacher & Hayes, 2008).

Table 14

Model Summary Information for the Proposed Mediation Model Portrayed in Figure 4 (N = 121)

| | | | Outcome | | | |
|------------------------|---|---|---|---|-------------------------------|---|
| Predictors | MAP (M 1) | MAV (M 2) | PAP (M 3) | PAV (M 4) | Course Satisf | action |
| CDI D | $B(\beta)$ SE p | $B(\beta)$ SE p | $B(\beta)$ SE p | $B(\beta)$ SE p | $B(\beta)$ | SE p |
| SDLR $a_7 \rightarrow$ | $(0.30 \ 0.09 \ < 0.001 \ a_8 - (0.31)$ | \rightarrow -0. 14 0.15 0.33 a_9 | $-0.13 \ 0.18 \ .44 \ a_{10} \rightarrow$ | 30 .19 .12 (- 14) | $c_4 \rightarrow .48$ | .14 <.001 |
| MAP | (0.31) | (.0) | (.07) | (.17) | $b_7 \rightarrow .21$ (13) | .15 .17 |
| MAV | | | | | $b_8 \rightarrow .01$ | .08 .92 |
| PAP | | | | | $b_9 \rightarrow14$ | .07 .06 |
| PAV | | | | | $b_{10} \rightarrow .11$ (15) | .07 .14 |
| Constant | 5.18 0.34 < .001 | 6.34 .59 < .001 | 5.65 .71 < .001 | 6.02 .76 < .00 | 01 .41 | 0.95 0.66 |
| | $R^2 = 0.09, F(1, 119)$ = 12.42, p < 0.001 | $R^2 = 0.01, F (1, 115)$ = 4.57, p < 0.001 | $R^2 = 0.01, F(1.119) = 0.59, p = 0.44$ | $R^2 = 0.02, F(1, 119)$ = 2.41, p = 0.12 | 1 | $R^2 = 0.16, F$ (1, 115) =4.57, $p < 0.001$ |

Note. B = Unstandardized regression coefficient; β = Standardized regression coefficient.

Table 15

Summary of the Mediation Model 2-2 Analysis

| | | | | | 95% | ω CI |
|----------------|----------|--|----------|------|-------|------|
| Effect | Mediator | Equation | Estimate | SE | LL | UL |
| Indirect | MAP | $a_7 	imes b_7$ | 0.06 | 0.05 | -0.02 | 0.18 |
| Indirect | MAV | $a_8 	imes b_8$ | -0.00 | 0.02 | -0.04 | 0.03 |
| Indirect | PAP | $a_9 	imes b_9$ | 0.02 | 0.03 | -0.03 | 0.09 |
| Indirect | PAV | $a_{10} 	imes b_{10}$ | -0.03 | 0.03 | -0.10 | 0.02 |
| Total indirect | | $a_7 	imes b_7 + a_8 	imes b_8 + a_9 	imes b_9 + a_{10} 	imes b_{10;}$ | 0.05 | 0.06 | -0.06 | 0.19 |
| Direct | | C4 ' | 0.48 | 0.14 | 0.21 | 0.76 |
| Total | | C4 | 0.53 | 0.13 | 0.28 | 0.79 |

Note. CI = confidence interval; LL = lower limit; UL = upper limit. If the confidence interval does not include zero, it reveals a significant effect.

Table 16

Explanation of Variables of Models and Research Questions

| Research Question | Model | Outcome | Mediator Pa | ath | Results |
|----------------------|-------|-----------------------|------------------------|---------|-------------|
| RQ 1.1 | 1-1 | Cognitive Learning | Direct SDL - effect | → Grade | Significant |

| RQ 1.2 | | Outcome | Indirect | $SDL \rightarrow SELS \rightarrow Grade$ | Not Significant |
|---------------|-------|---------------|----------|--|-----------------|
| | | (Grade) | Effect | $SDL \rightarrow MAP \rightarrow Grade$ | Significant |
| DO 1 2 | 1.2 | | | $SDL \rightarrow MAV \rightarrow Grade$ | Not Significant |
| KQ 1.5 | 1-2 | | | $SDL \rightarrow PAP \rightarrow Grade$ | Not Significant |
| | | | | $SDL \rightarrow PAV \rightarrow Grade$ | Not Significant |
| PO 2 1 | | Affective | Direct | $SDL \rightarrow Satisfaction$ | Significant |
| KQ 2.1 | _ 2 1 | Learning | effect | | |
| PO 22 | 2-1 | Outcome | Indirect | $\text{SDL} \rightarrow \text{SELS} \rightarrow$ | Not Significant |
| KQ 2.2 | | (Course | Effect | Satisfaction | |
| | | Satisfaction) | | $SDL \rightarrow MAP \rightarrow Grade$ | Not Significant |
| DO 2 2 | 2.2 | | | $SDL \rightarrow MAV \rightarrow Grade$ | Not Significant |
| KQ 2.3 | 2-2 | | | $SDL \rightarrow PAP \rightarrow Grade$ | Not Significant |
| | | | | $SDL \rightarrow PAV \rightarrow Grade$ | Not Significant |

Figure 8

Model summary information for the hypothesized mediation model portrayed in Figure 1



Figure 9

Model summary information for the hypothesized mediation model portrayed in Figure 2



Figure 10

Model summary information for the hypothesized mediation model portrayed in Figure 3



Figure 11

Model summary information for the hypothesized mediation model portrayed in Figure 4



Summary

The primary focus of this quantitative, cross-sectional correlation study was to investigate whether and to what extent OSDLR has an impact on academic outcomes measured by grade (cognitive learning outcome) and course satisfaction (affective learning outcome) through the potential mediators of SELS and each construct of 2x2 achievement goal orientation. The present study was conducted by analyzing the data collected from students enrolled in and studying at a large public university in the Southeast of the United States who had registered for at least one statistics course delivered online during the 2020 Fall semester. To answer the research questions of the current study, data was analyzed using OLS regression to estimate the hypothesized mediation model. In this model, the main interest was the indirect effects, which were representative of research questions that tested OSDLR as it relates to grade and course satisfaction through the potential mediators of SELS only ($a_1 \times b_1$; $a_6 \times b_6$) and each construct of 2x2 achievement goal orientation: MAP, MAV, PAP and PAV ($a_2 \times b_2$, $+ a_3 \times b_3 + a_4 \times b_4 + a_5 \times b_5$; $a_7 \times b_7$, $a_8 \times b_8$, $a_9 \times b_9$, and $a_{10} \times b_{10}$).

This chapter provided the findings of the mediation model analyses. The results partially supported the research questions, thus suggesting that there was evidence of mediating pathways from OSDLR to academic outcomes measured by grade (cognitive learning outcomes) and course satisfaction (affective learning outcomes) through MAP.

In chapter 5, I will provide a comprehensive summary of the entire study, its limitations and conclusions, and its implications based on the findings of the mediation analysis.

Chapter 5 - Summary, Conclusions, Implications, Limitations, and Recommendations for Future Research

Overview

This chapter presents the study summary, conclusions based on the data analysis, implications of the findings, limitations, and results. Recommendations for future research are also described.

Problem Statement

The COVID-19 global pandemic has led to drastic changes in educational systems around the world. Students were forced to shift from physical to online classes in response to government restrictions on face-to-face instruction. These restrictions were considered critical to preventing the spread of COVID-19. Aguilera-Hermida (2020) explained that while numerous challenges accompanied these changes, these changes also created an awareness of the role and advantages of engaging in online learning. Therefore, even after the risk of virus spread was reduced, the popularity of online learning remained high. Bashir et al. (2021) reported that Aston University adopted a hybrid mode of course delivery (combining online and face-to-face courses) as a post-COVID solution that highlighted the popularity of online learning. A survey cited by Kelly (2021) established that 73% of students preferred studying fully online after the pandemic; however, only about 53% of faculty preferred teaching online.

In recent years, many statistics courses have also been transferred online, with studies such as that conducted by Ritzhaupt et al. (2020) revealing that students enrolled in the coursework experience less anxiety, making online instruction preferable to in-person learning. The popularity of online statistics courses post-pandemic has created a demand for understanding this modality of learning and how learners and educators can organize their courses effectively.

Further demand for understanding the learning modality has been created by studies such as those by Figueroa-Cañas and Sancho-Vinuesa (2020), which have reported that the dropout rate of online statistics courses is higher than that of traditional statistics courses. This is a significant concern because, even though most learners prefer online learning, this same online model has seen higher dropout rates. However, Knowles' self-directed learning theory has encouraged scholars to focus on self-efficacy to improve online statistics courses (Manning, 2007). Thus, the present study sought to explore the existing gap in the literature regarding the effects of selfefficacy, self-directed learning readiness (SDLR), and goal orientation on learning outcomes for learning statistics online.

Research Questions

1. What is the extent of the relationship between online self-directed learning readiness and grades mediated by self-efficacy to learn statistics and each construct of achievement goal orientations in an online statistic learning environment?

1.1 What is the extent of the relationship between online self-directed learning readiness and grades in an online statistic learning environment?

1.2 What is the extent of the relationship between online self-directed learning readiness and grades mediated by self-efficacy to learn statistics in an online statistic learning environment?

1.3 What is the extent of the relationship between online self-directed learning readiness and grades mediated by each construct of achievement goal orientations (MAP, MAV, PAP, PAV) in an online statistic learning environment?

2. What is the extent of the relationship between online self-directed learning readiness and course satisfaction mediated by self-efficacy to learn statistics and each construct of achievement goal orientations in an online statistic learning environment?

2.1 What is the extent of the relationship between online self-directed learning readiness and course satisfaction in an online statistic learning environment?

2.2 What is the extent of the relationship between online self-directed learning readiness and course satisfaction mediated by self-efficacy to learn statistics in an online statistic learning environment?

2.3 What is the extent of the relationship between online self-directed learning readiness and course satisfaction mediated by each construct of achievement goal orientations in an online statistic learning environment?

Study Overview

The increased demand for online learning in today's education system has caused scholars and educators to question how statistics coursework will be affected by the features of online teaching (Akabayashi et al., 2023). Thus, developing an understanding of self-directed, selfefficacy, and goal orientation theories is crucial to understanding such effects. Undoubtedly, online learning has drastically expanded learners' horizons by enabling them to not only save time but also access learning materials and resources without geographical barriers. As a result, the completion rate has increased as the number of online learning graduates increases. Nevertheless, the concept of the efficiency of online courses has laid the foundation for scholars to explore whether they are more effective than face-to-face learning (Kemp & Grieve, 2014).

Considering the growing demand for online learning, understanding the extent to which the outcomes of learning statistics online have been affected by self-directed learning readiness,

self-efficacy, and goal orientation is crucial to promoting success among such classes (Chu & Tsai, 2009). The key focus of this study was to explore the literature gaps regarding the effects of self-directed learning readiness, self-efficacy, and goal orientation on learning outcomes for learning statistics online. While many studies on online learning struggle to examine the effectiveness of self-directed, goal-oriented, and self-efficacy theories in improving academic performance, this study examined the gap regarding learning outcomes for learning statistics online. To examine these gaps, this study utilized 2x2 achievement goal orientation, self-efficacy, and self-directed learning theory, which served as a road map for developing the arguments in this study.

Self-directed learning theory was used to demonstrate its effectiveness in enhancing selfefficacy in learners. This theory argues that individuals take initiative, with the help of others or on their own, to formulate goals, identify learning resources, and implement learning strategies (Charokar, 2022). Tekkol and Demirel's article argues that self-directed learning theory shifts the learning responsibility from educators or any external source to individuals (Tekkol & Demirel, 2018). Throughout the study, self-directed learning was used to investigate the extent to which self-directed learners take responsibility for their education and accept the independence that comes with focusing on meaningful topics. The theory was crucial in understanding knowledge acquisition, learner preparedness, and the extent to which this theoretical learning approach appears to be a pedagogical delivery that learners can benefit from in the changing world (Tekkol & Demirel, 2018). Moreover, it examined the extent to which learning can affect learners' academic performance, since the theory is regarded as a positive predictor of academic performance (Li et al., 2022). Hence, as a pedagogical strategy, self-directed learning facilitates understanding learners' needs in a rapidly globalizing, diversifying, and technologizing world

(Charokar, 2022). Furthermore, it was crucial to understand the relationship between mastery of concepts, motivation, and online statistics learners' performance.

As a primary element of Bandura's social-cognitive theory, self-efficacy tends to focus more on the role of motivation in academic achievement. This theory reviewed individuals' confidence and abilities to control their motivation, behavior, and external environments (Domenech-Betoret et al., 2017). The theory provided information on mastery experience and predicted students' academic achievements and performances. It allowed the researcher to recognize the need to achieve academic goals and superior grades and embrace new experiences. Furthermore, the theory was used to facilitate the study of mastery experiences, social modeling, social persuasion, and psychological responses (Bandura, 2012). By understanding the interaction between students' thoughts and the nature of the assigned tasks, the researcher was able to understand the relationship between self-efficacy and higher achievement levels (Iroegbu, 2015). Apart from being a predictor of academic success, the theory facilitates an understanding of how self-efficacy influences various learning areas. It allowed the researcher to explore the relationship between motivation and student performance and how self-efficacy improved learning statistics online. In addition to providing information on how to achieve successful statistical education and information, self-efficacy theory helped explore students' abilities to use symmetric variables. This theory was significant in understanding how statistical self-efficacy can contribute to improved performance.

Goal orientation theory is a social cognitive theory of achievement motivation. Unlike other motivational theories, goal orientation focuses on what stimulates learners to be more concerned with and engaged in learning (Wang et al., 2021). At the same time, the theory focuses on demonstrating why this theory is important to learners. The theory related to the current study

facilitated an understanding of how students can improve their skills by concentrating on the learning process attained through effort. Furthermore, it facilitated an understanding of the role of mastery orientation in attaining academic goals (Sidebar & Shankar, 2022) and helped the researchers explore the role of performance-oriented learning in promoting student performance. Using goal orientation theory, the study managed to relate motivation to academic performance and the extent to which it supports a positive attitude toward performance-oriented goals. Generally, these theories were crucial in understanding how academic performance can be impacted. The relationship between this theory and the current study is that the theory determines the probability of success and the probability of attaining academic and professional goals. The theory provided the platform through which educators evaluated learners' initial core orientation (ability to master skills) and avoided failure or sought success. This was a critical area explored in this study, and the theory uncovered the extent to which learners became more concerned and engaged in learning.

The current study sought to create awareness and inform learners and tutors engaging in online statistics courses about how they can rely on self-directed learning readiness. The main objective of the study was to understand the effects of SDLR on grades and course satisfaction for online learning statistics through the mediating variables of self-efficacy and 2x2 achievement goal orientation. By understanding the potential mediator roles of SELS and AGO in learning statistics and OSDLR, the current study will have comprehensively investigated the relationship between OSDLR and grade outcomes and course satisfaction.

A quantitative research design explored the relationships between online self-directed learning readiness, self-efficacy to learn statistics, 2x2 achievement goal orientations, course satisfaction, and grades in online statistics course settings. To address the research questions, the

study included a total of 168 students who participated in the survey. However, 47 participants were removed because of their high level of missing data. Therefore, only 121 participants were involved in this study. The Achievement Goal Questionnaire instrument evaluated AGOs computerized in the 2x2 goal framework. In addition, self-efficacy in learning statistics was a reliable instrument for measuring students' confidence and ability to learn statistics. Hung et al.'s (2010) article highlighted another instrument used in this study—an online SDLR used in the survey to evaluate online learning readiness. Such instruments were crucial in data analysis. Participants were those who enrolled as students at the university, were 19 years old or older and had registered for at least one statistics course delivered online during the 2020 fall semester. A cross-sectional and correlational study was used to collect data on relevant variables from different people, subjects, and phenomena.

Findings

The study's findings indicated that online self-directed learning readiness was positively associated with efficacy in learning statistics and grades. However, self-efficacy in learning statistics was not associated with grades. Moreover, the findings indicated that online self-directed learning readiness was positively related to MAP and that MAP was positively related to cognitive learning outcomes (grades). In addition, self-efficacy in learning statistics was positively related to satisfaction. Nonetheless, the indirect effect of self-efficacy in learning statistics and satisfaction was insignificant. It must be noted that only PAP was negatively related to course satisfaction, even if there was no significant indirect effect of self-efficacy.

Research Questions 1.1, 1.2 and 1.3 sought to investigate the effects of OSDLR on grades, cognitive learning outcomes, and the mediating effects of the association between OSDLR and grades through the mediators: SELS and each construct of 2x2 achievement goal

orientations (MAP, MAV, PAP, and PAV from model 1-1 and 1-2.). The analysis revealed that OSDLR was directly associated with grades; however, there is no evidence of an association between SELS and grades. The result of the mediation model analysis did not support that SELS mediates the association between OSDLR and grades. Otherwise, in one of the 2x2 achievement goal orientations, only the MAP orientation was found to be a significant mediator of the association between OSDLR and grades.

Research Questions 2.1, 2.2 and 2.3 investigated the effects of OSDLR on course satisfaction, affective learning outcomes, and the mediating effects of the association between OSDLR and grades through the mediators of SELS and each construct of 2x2 achievement goal orientations (MAP, MAV, PAP, and PAV from models 2-1 and 2-2). The analysis revealed that OSDLR was related positively to SELS, and SELS was associated with affective learning outcomes (course satisfaction) when controlling for SDLR. However, this did not yield any significant indirect effect from the resampling method using 5,000 bootstrap samples applied to generate a 95% bias-corrected bootstrap confidence interval, which was found to be around zero. Therefore, this does not support the idea that SELS mediates the relationship between online self-directed learning readiness and affective learning outcomes (course satisfaction). The result analysis from model 2-2 showed that there is no evidence of a significant association between any of the constructs of 2x2 achievement goal orientations and course satisfaction.

Conclusions

The results of this study are divided into different models used in the analysis. The results indicated that some studies were consistent with the results of the current study, while others contradicted them. As demonstrated in the subsequent sections, some studies, such as Uus et al. (2022), support the findings of the current study that technology (such as the Internet) helps

learners engage in self-directed learning. Previous research on the effects of self-directed learning readiness, self-efficacy, and goal orientation on learning outcomes for online statistics courses is limited. However, previously conducted studies such as that by Loeng (2020) demonstrated that self-efficacy, self-directed learning readiness, and goal orientation theories play a role in explaining online learners' performances in a different manner. As a pedagogical strategy, self-directed learning readiness is an excellent way of teaching students how to manage knowledge. The current study examined the extent to which self-directed learning readiness was related to self-efficacy in learning. It found a positive relationship between self-directed learning readiness and self-efficacy to learn statistics and satisfaction, respectively. Hence, it revealed that self-directed learners could identify learning requirements, create learning plans, and choose practical learning plans. Like previous studies, the current study revealed that technology empowered learners' abilities to engage in self-directed learning. In the following section, the conclusion regarding all variables in each model is first introduced, followed by the studies that support these findings and what these findings mean.

Research Questions 1.1 and 1.2

The results from this model indicate that online self-directed learning readiness (OSDLR) is positively related to grade, which is a cognitive learning outcome. The results also indicate that OSDLR is positively related to SELS. At the same time, the results did not provide evidence of the association between SELS and cognitive learning outcomes (grades). In other words, SELS is not a significant mediator between OSDLR and grade.

Existing research that was consistent with the findings of this study with regards to the relationship between OSDLR and grade as a learning outcome includes a study by Wei and Chou (2020), who found a positive relationship between OSDLR and student learning outcomes, such

as grade. Tang et al. (2021) also established that the Internet, an important technology in facilitating OSDLR, plays a crucial role in strengthening OSDLR by positively influencing cognitive learning outcomes or grades.

Previous research on OSDLR and its relationship with learning outcomes was generally limited. Still, there was a plethora of literature on the relationship between SDLR and grades. For example, studies such as that by Örs (2018) argued that SDLR has a positive relationship with grades and that SDLR puts learners in a better position to comprehend their academic material and, thus, perform better. Shokar et al. (2002) also found a positive correlation between students' scores on the SDLR scale and their course grades.

In terms of the relationship between OSDLR and SELS, Tang et al. (2021) established a positive relationship between OSDLR and SELS that is consistent with the current study's findings. Tang et al. (2021) argued that motivation is key in OSDLR since it determines students' attitudes while learning online. Other studies, such as that by Yilmaz (2017), indicated that self-efficacy is a critical factor when discussing OSDLR since it enables the interaction between internal and external factors that enable students to achieve a certain degree of confidence to learn about a given concept, such as statistics. These results mean that students with higher levels of OSDLR achieve higher levels of self-efficacy, thereby creating pathways for them to interact with their teachers and peers. Other studies, such as that by Turan and Koç (2018) and Dong et al. (2021), investigated the relationship between self-efficacy and self-directed learning readiness and self-efficacy.

Some of the studies that contradict the findings of this study include those by Domenech-Betoret et al. (2017) and Betoret et al. (2014), which found a positive relationship between self-

efficacy and grade. The sample comprises 797 Spanish secondary education students from 36 educational settings and three schools. The researchers argued that self-efficacy could be associated with motivation, better attitudes toward learning, good relationships and interactions with teachers and peers, and more confidence.

Studies such as the one conducted by Doménech-Betoret et al. (2017), postulate a relationship between self-efficacy, course satisfaction, performance, and expectancy-value beliefs. According to their findings, process expectations are connected to students' emotions during interactions with their teachers and play a significant role in determining satisfaction levels. However, this process can be influenced by students' self-efficacy and their expected performance in terms of grades. In other words, students with strong self-efficacy beliefs are more likely to envision successful outcomes, which guide their performance and provide them with the necessary support. As a result, these students tend to experience higher satisfaction with the learning process and achieve better performance.

The inconsistency between the findings of previous research and the current study regarding the relationship between self-efficacy, course satisfaction, and grades may be attributed to differences in sample characteristics, such as sample size, research context, study population, and the validity and reliability of the research instruments used in both cases.

Regarding the relationship between online self-directed learning readiness (OSDLR) and grades, both previous literature and the current study demonstrate a positive association. This implies that learners who possess the necessary attitudes, capabilities, and personality traits for online self-directed learning are more likely to achieve high academic performance. Furthermore, the results of this study suggest that such learners, who have the appropriate attitudes, capabilities, and personality characteristics for online self-directed learning, are also more likely

to be confident in their ability to effectively learn statistical skills required for their statistics course.

Research Question 1.3

The results indicated that OSDLR is positively related to grades, as established in model 1-1. The results also established that OSDLR is positively related to MAP but shows no positive relationship with any other orientations (MAV, PAP and PAV). Moreover, MAP is positively related to grades; therefore, MAP is a significant mediator between the relationship between OSDLR and cognitive learning outcomes. It was also established that MAV is negatively related to grades and that neither PAP or PAV is significantly associated with grades.

The findings of the current study regarding the variables mentioned above are consistent with previous research. Studies cited in this study, such as those by Simamora and Mutiarawati (2021) and Keklik and Keklik (2013), indicate that the mastery approach (MAP) is the only significant mediator in the relationship between online self-directed learning readiness (OSDLR) and cognitive learning outcomes (grades). These studies show that MAP has a positive relationship with both OSDLR and grades, while the relationship between a mastery-avoidance approach (MAV) and grades is negative.

The results of this study suggest that learners who focus on developing competence-based learning, expanding their understanding, and improving their performance (or simply those who display a MAP orientation) are likely to achieve high academic performance. Furthermore, these findings imply that such students are more likely to possess the attitudes, capabilities, and personality characteristics necessary for online self-directed learning. Regarding the negative relationship between MAV and grades, the results suggest that learners who focus on avoiding the loss of knowledge or skills are likely to perform poorly academically. Additionally, whether

learners strive to outperform others or demonstrate their competence does not impact their academic performance or their attitudes, capabilities, and personality characteristics necessary for online self-directed learning. Finally, whether learners focus on avoiding looking incompetent, making errors, or being outperformed by others does not affect their academic performance or their attitudes, capabilities, and personality characteristics necessary for online self-directed learning.

Research Questions 2.1 and 2.2

The results indicate a positive association between OSDLR and course satisfaction. They also show a positive association between self-efficacy for learning statistics (SELS) and course satisfaction. However, it was found that SELS did not serve as a significant mediator. These findings align with several previous studies, including those by Wei and Chou (2020) and Ariffin et al. (2020), who argue that OSDLR positively influences course satisfaction because it includes qualities such as self-confidence in using e-communication, student perception of learning delivery, and autonomy in learning participation, which are key components of course satisfaction. Ariffin et al. (2020) further note that OSDLR enhances student engagement in online courses, thereby increasing the likelihood of course satisfaction. Hoogerheide et al. (2018) and Hsia et al. (2016) also suggest that self-efficacy positively impacts overall course satisfaction as it improves learners' perception of the teaching model, particularly in terms of seeking high instructional value.

The results of this study indicate that OSDLR, as well as a learner's attitudes, capabilities, and personality characteristics necessary for online self-directed learning, play a significant role in determining students' satisfaction with their online statistics courses. Additionally, the confidence a student has in their ability to effectively learn statistical skills influences course

satisfaction, but it does not affect their capabilities and personality characteristics necessary for self-directed learning in relation to course satisfaction.

Research Question 2.3

The results in this model show that none of the 2x2 achievement orientations were associated with course satisfaction. These results contradict most of the previous research. For example, studies such as that by Wolters (2004) argue that mastery structure and mastery orientation are directly associated with affective outcomes and course satisfaction. Butler (1987) found that achievement orientation can play a very important role in strengthening the ability of learners to master their coursework, which positively affects course satisfaction. Butler (1987) added that a mastery-goal orientation helps learners tackle complex tasks due to their intrinsic interest in learning and orientation toward the ultimate goal of learning, improving their skills, and increasing the chances of course satisfaction. Mega et al. (2014) argued that students with a mastery approach utilize cognitive strategies that enable them to comprehend their current subjects better and have the capacity to attain maximum skills and knowledge in a given course. According to Mega et al. (2014), course satisfaction among such students is often higher because learners are more likely to be satisfied in courses where they understand the content and have gained enough skills.

The results of this study imply that the patterns affecting a student's cognitive or emotional behavior regarding events do not affect their course satisfaction. There are some fundamental reasons why the current study's findings differ from previous research. They are, among others, the differences in sample characteristics and sample sizes, the differences in study populations, the fact that some previous studies investigated different subjects apart from statistics, and the fact that some previous studies investigated more than one subject.

Implications

Online learning has gained significant recognition and appreciation in educational institutions. However, studies specifically focused on learning statistics online are limited, as most research tends to concentrate on general online teaching and learning processes. Consequently, there is a lack of literature addressing the needs of online learners and the enhancement of learning outcomes in the context of online statistics learning (Salah Dogham et al., 2022). Given the challenges faced by online statistics learners, it is crucial to develop a deeper understanding of the impact of self-directed learning readiness, self-efficacy, and goal orientation on learning outcomes to enhance performance among these learners (Salah Dogham et al., 2022).

Mastery avoidance goal Orientation refers to an individual's focus on avoiding mistakes, failure, or the appearance of incompetence, rather than actively seeking to develop competence or master tasks (Lin et al., 2019). It suggests that the person's primary concern is to avoid looking incompetent or making errors, rather than pursuing growth or learning. The findings from the current study indicate that having a mastery avoidance goal orientation is associated with lower academic performance or poorer grades. Students who are primarily focused on avoiding mistakes or failure may not fully engage in the learning process, take risks, or invest the necessary effort to achieve high levels of understanding and performance. Given that mastery avoidance goal orientation is linked to lower grades, it is important for the instructor to promote mastery-approach goals (Benita, 2021). Encouraging students to focus on their personal growth, learning from mistakes, and seeking a deep understanding of statistical concepts can help counteract the negative effects of a mastery avoidance mindset. The instructor can provide examples of how mistakes and setbacks are valuable learning opportunities and emphasize the

importance of effort and perseverance. Students with a mastery avoidance goal orientation may be more sensitive to criticism or fear of failure. Therefore, the instructor should create a supportive online learning environment that encourages risk-taking, emphasizes learning from mistakes, and provides constructive feedback. Offering opportunities for students to practice and apply statistical concepts in a safe and non-judgmental setting can help build their confidence and reduce anxiety related to making errors. Students with a mastery avoidance goal orientation may benefit from individualized support and motivation (Tuominen et al., 2012). The instructor can provide personalized feedback, guidance, and encouragement to help students overcome their fear of failure and develop a growth mindset. Recognizing and rewarding students' efforts, progress, and improvements can also help shift their focus from avoiding mistakes to embracing challenges and pursuing mastery.

Also, the instructor can assess students' goal orientations, including any signs of mastery avoidance, through surveys, self-reflection activities, or informal discussions. Identifying students who exhibit a mastery avoidance goal orientation can allow for targeted interventions and support. The instructor can engage in open and honest conversations about goal orientations, emphasizing the importance of a growth mindset and helping students reframe their perspectives on mistakes and failure.

From the results of the current study, encouraging students to adopt a mastery-approach goal orientation can be beneficial. By focusing on their personal growth, students are more likely to engage actively in the learning process, persist in the face of challenges, and strive for a deeper understanding of statistical concepts. The instructor can promote mastery goals by highlighting the importance of learning for its own sake, providing constructive feedback that supports improvement, and setting high but attainable expectations (Tuominen et al., 2012).

To recognize the significance of self-directed learning readiness, the instructor can design the course to foster independent learning skills. This can include incorporating self-paced learning modules, encouraging students to set their own learning goals, and providing resources and guidance for self-directed study. Promoting a supportive online learning environment that encourages active participation and collaboration can also enhance students' readiness for selfdirected learning. Also, the instructor can monitor students' goal orientations throughout the course by using surveys or informal assessments to gauge their level of mastery-approach goal orientation. This information can help identify students who may need additional support in developing a mastery-focused mindset. Moreover, considering the connection between masteryapproach goal orientation and grades, the instructor can use this knowledge to provide targeted guidance and interventions to enhance students' academic performance.

Acknowledging the positive association between online self-directed learning readiness and self-efficacy to learn statistics, the instructor should foster an online learning environment that promotes and supports self-directed learning. This can include providing resources and tools for independent study, offering opportunities for students to set learning goals and monitor their progress, and promoting active engagement and collaboration among students (Liang et al., 2023). The instructor can also guide students in developing effective learning strategies and time management skills suited for online learning.

While self-efficacy to learn statistics is positively associated with course satisfaction, it does not mediate the relationship between course satisfaction and online self-directed learning readiness. This suggests that factors other than self-efficacy contribute to students' satisfaction with the course. The instructor should consider various aspects of the course experience, such as instructional design, course materials, instructor support, and engagement strategies, to ensure a

positive and enriching learning environment. Collecting regular feedback from students and making necessary improvements based on their suggestions can also enhance course satisfaction.

Understanding the complex relationships between self-efficacy, course satisfaction, and online self-directed learning readiness, the instructor should adopt a holistic approach to support students' learning experiences. This involves addressing multiple dimensions, including building self-efficacy beliefs, fostering self-directed learning skills, and creating a satisfying course environment. By attending to these interrelated factors, the instructor can contribute to students' overall satisfaction, engagement, and success in the online statistics course.

Limitations

There were some limitations regarding the sample size for this study. The study relied on 168 participants, but 47 were removed because of high levels of missing data. Overall, only a valid sample of 121 participants was analyzed. This sample size is not large enough; hence, the researcher had to change the study's original design. Although Delice (2010) indicates that while the minimum sample size for quantitative research is 100, using a minimum sample in a study with such a large study population (by 2020, 7 million undergraduates were enrolled exclusively in online college courses, according to Reimers (2022)) is not sufficient. Additionally, Kock and Hadaya (2018) recommend that the best sample size for path analysis should be at least 20 times the number of parameters, and since this study had at least eight parameters, the best sample size would have been 160 participants.

Initially, it was planned to use SEM to demonstrate how latent variables were related and the dynamics of relations among every variable. However, due to the limited sample size, the study had to be divided into four different models. The reliability of the scale could have been improved if the study had a larger sample size. Additionally, it is important to note that this study

did not include control variables due to the sample size and complexity of the models. Control variables are essential in any study as they help establish the correlation between dependent and independent variables. Therefore, this study was unable to establish a causal relationship between the variables of interest (York, 2018). Furthermore, the inclusion of both undergraduate and graduate students in the study may be seen as a limitation since the differences in grade and/or degree among students could have had a stronger impact on their self-efficacy skills. For instance, junior or senior students might have higher self-efficacy in learning statistics compared to freshmen. It is also important to consider potential factors such as gender and ethnicity in this study. Regarding the measurement of the mediation model, the initial plan was to measure the models in sequence. However, due to the nature of correlational research, which does not demonstrate causal relationships, testing the mediation models in sequence proved difficult. Additionally, some participants did not answer most parts of the survey, leading to the deletion of unanswered responses. This created a gap in explaining missing data, as deletion is not the best way to address this issue and may affect the reliability and validity of the study. Furthermore, the study relied on convenience sampling, which poses limitations as convenient samples may not fully represent the target population, thereby hindering generalization (Akabayashi et al., 2023). Therefore, the choice of sampling method affected the ability to generalize from the sample to the population of interest. Moreover, the study relied on self-reported questionnaires, increasing the likelihood of participant bias. The variables measured in the study required participants to express their opinions regarding goal orientation, self-efficacy, and self-directed learning theories. Since the study focused on online statistics learning settings, the generalizability of the results to other subjects and settings is questionable. The inclusion of numerous variables in the study also increased the potential for confounding variables. Additionally, some students reported

their grades from multiple online statistics courses, which may have affected the reliability and validity of the data. Lastly, students may not have fully understood their achievement goal orientation.

Recommendations

Future studies should explore the effectiveness of self-directed learning, self-efficacy, and 2x2 achievement goal orientations (MAV, MAP, PAP, PAV) on outcomes for learning statistics online. In future studies, it would be valuable to explore influential factors such as learners' intrinsic motivation, learning goals, and attitudes toward the subject to further investigate their impact on the effectiveness of these theories in promoting positive affective and cognitive learning outcomes among online learners. To measure cognitive and affective learning outcomes, future studies should consider using an experimental research design. Additionally, it would be beneficial to compare the outcomes of online and face-to-face learning approaches to obtain reliable results. It should be noted that the current study lacked a control group, which is a limitation. Categorizing intervention and control groups in future studies would enhance the accuracy and validity of the data.

The study found a positive association between online self-directed learning readiness, self-efficacy in learning statistics, and grades. Additionally, online self-directed learning readiness was positively related to MAP, which, in turn, was positively related to grades, while MAV was negatively related to grades. Furthermore, self-directed learning readiness was positively related to self-efficacy in learning statistics and satisfaction, and self-efficacy in learning statistics was positively related to satisfaction. Many factors can influence academic performance among learners, including stress, satisfaction, and motivation. Therefore, future studies should explore how these influences affect the extent to which the mentioned theories

impact performance among online learners. Such studies could provide a deeper understanding of the factors affecting academic performance and offer potential solutions to issues affecting the effectiveness of online learning.

Moreover, the current study relied on a survey method to collect data. In future studies, it would be valuable to utilize mixed methods research design that involves combining quantitative research design approaches and qualitative studies, such as observation, interviews, focus groups, and group discussions. These approaches would help gain a deeper understanding of the effects of self-directed learning, self-efficacy, and goal orientation on learning outcomes for online statistics learning. Additionally, follow-up studies should aim to clarify the factors that influence learners' performance and how they impact the success of self-directed learning, self-efficacy, and goal orientation in promoting the effectiveness of online statistics learning.

These recommendations have the potential to benefit not only online statistics learning but also online learning in general. With an increasing number of educational institutions adopting online learning, these recommendations may contribute to the improvement and effectiveness of the online learning approach. Furthermore, as policies regarding online learning continue to evolve, the recommendations presented in this study can serve as a foundation for shaping future online learning policies. They can inform and guide decision-making processes within educational institutions' administrations.

Lastly, online learning is a unique approach. This study takes an important step in addressing gaps in understanding the effects of self-directed learning readiness, self-efficacy, and goal orientation on learning outcomes for online statistics learning. Moreover, it highlights the significance of improving academic performance among online learners. Therefore, this study contributes to the existing literature in this field and extends the application of these theories to

shaping academic performance. Additionally, it suggests practical strategies that higher education institutions, faculty members, and other education stakeholders can use to support student achievements.

References

- Abdi, H. M., Bageri, S., Shoghi, S., Goodarzi, S., & Hosseinzadeh, A. (2012). The role of metacognitive and self-efficacy beliefs in students' test anxiety and academic achievement *Australian Journal of Basic and Applied Sciences*, 6(12), 418–422.
- Acharya, A. S., Prakash, A., Saxena, P., & Nigam, A. (2013). Sampling: Why and how of it. *Indian Journal of Medical Specialties*, *4*(2), 330-333.
- Aguilera-Hermida, A. P. (2020). College students use and accept emergency online learning due to COVID-19. *International Journal of Educational Research Open*, 1(1), 100011. https://doi.org/10.1016/j.ijedro.2020.100011
- Akabayashi, H., Taguchi, S., & Zvedelikova, M. (2023). Access to and demand for online school education during the COVID-19 pandemic in Japan International Journal of Educational Development, 96, 1-12. https://doi.org/10.1016/j.ijedudev.2022.102687
- Akbar, S., Claramita, M., & Kristina, T. (2017). Intrinsic motivation and self-directed learning relationships: Strive for adult learning character formation. *South-East Asian Journal of Medical Education*, 11(1), 26. https://doi.org/10.4038/seajme.v11i1.5
- Aldemir, T., Borge, M., & Soto, J. (2022). Shared meaning-making in online intergroup discussions around sensitive topics. *International Journal of Computer-Supported Collaborative Learning*, 17(3), 361-396. https://doi.org/10.1007/s11412-022-09375-9

Alfaifi, M. S. (2016). Self-directed learning readiness among undergraduate students at Saudi

Electronic University in Saudi Arabia [Doctoral dissertation, University of South

Florida]. Digital Commons. https://digitalcommons.usf.edu/etd/6449

- Alhazzani, H., AlAmmari, G., AlRajhi, N., Sales, I., Jamal, A., Almigbal, T. H., Batais, M. A., Asiri, Y. A., & AlRuthia, Y. (2021). Validation of an Arabic version of the self-efficacy for appropriate medication use scale *International Journal of Environmental Research and Public Health*, 18(22), 11983. https://doi.org/10.3390/ijerph182211983
- Allen, I. E., & Seaman, J. (2016). *Opening the textbook: Educational resources in US higher education, 2015-16.* Babson Survey Research Group.
- Alqurashi, E. (2016). Self-efficacy in online learning environments: A literature review. *Contemporary Issues in Education Research (CIER)*, 9(1), 45–52.

https://doi.org/10.19030/cier.v9i1.9549

- Alrakaf, S., Sainsbury, E., Rose, G., & Smith, L. (2014). Identifying achievement goals and their relationship to academic achievement in undergraduate pharmacy students *American Journal of Pharmaceutical Education*, 78(7), 133. <u>https://doi.org/10.5688/ajpe787133</u>
- Al-Asfour, A. (2012). Examining Student Satisfaction with Online Statistics Courses Journal of College Teaching and Learning, 9(1), 33-38. <u>https://doi.org/10.19030/tlc.v9i1.6764</u>
- Althauser, K. (2015). Job-embedded professional development: Its impact on teacher selfefficacy and student performance. *Teacher Development*, *19*(2), 210–225.

https://doi.org/10.1080/13664530.2015.1011346

Ames, C. (1992). Classrooms: Goals, structures, and student motivation. *Journal of Educational Psychology*, *84*(3), 261–271. https://doi.org/10.1037/0022-0663.84.3.261

- Ames, C., & Archer, J. (1988). Achievement goals in the classroom: Students' learning strategies and motivation processes. *Journal of Educational Psychology*, 80, 260-267. doi:10.1037/0022-0663.80.3.260
- Anderman, E. M., Eccles, J. S., Yoon, K. S., Roeser, R., Wigfield, A., & Blumenfeld, P. (2001). Learning to value mathematics and reading: relations to mastery and performanceoriented instructional practices *Contemporary Educational Psychology*, 26(1), 76–95. https://doi.org/10.1006/ceps.1999.1043
- Ariffin, A., Wan Hassan, W. A. S., Ahmad, F., Hamzah, N., Rubani, S. N. K., & Zakaria, N. (2020). Students self-directed learning readiness towards using the "SolveMe" Web in Technical and Vocational Education. *International Journal*, 9(3).
- Artino, A. R. (2012). Academic self-efficacy: from educational theory to instructional practice

Perspectives on Medical Education, 1(2), 76-85. https://doi.org/10.1007/s40037-0120012-5

- Arquero, J.L., Fernández-Polvillo, C., and Hassall, T. and Joyce, J. (2015), "Vocation, motivation and approaches to learning: a comparative study", *Education + Training*, Vol. 57 No. 1, pp. 13-30. https://doi.org/10.1108/ET-02-2013-0014
- Atkinson, J. W. (1957). Motivational determinants of risk-taking behavior. *Psychological Review*, 64(6, Pt.1), 359–372. https://doi.org/10.1037/h0043445
- Awang, Z., Afthanorhan, A., & Asri, M. A. M. (2015). Parametric and non parametric approach in structural equation modeling (SEM): The application of bootstrapping. *Modern Applied Science*, 9(9), 58.

Bandura, A. (1977). Social learning theory Prentice-Hall.

Bandura, A. (1986). *Social foundations of thought and action: a social cognitive theory* Prentice-Hall.

Bandura, A. (1993). Perceived self-efficacy in cognitive development and functioning.

Educational Psychologist, 28(2), 117–148. https://doi.org/10.1207/s15326985ep2802_3 Bandura, A. (1997). *Self-efficacy: the exercise of control* Freeman.

Bandura, A. (2012). On the Functional Properties of Perceived Self-Efficacy Revisited Journal of Management, 38(1), 9-44. https://doi.org/10.1177/0149206311410606

Bandura, A. Ross, D., & Ross, S. A. (1961). Transmission of aggression through the imitation of aggressive models *Journal of Abnormal and Social Psychology*, *63*(3), 575–582.

https://doi.org/10.1037/h0045925

Barbeau, K., Boileau, K., Sarr, F., & Smith, K. (2019). Path analysis in Mplus: A Tutorial using a Conceptual Model of Psychological and Behavioral Antecedents of Bulimic Symptoms in Young Adults *The Quantitative Methods for Psychology*, 15(1), 38–53.

https://doi.org/10.20982/tqmp.15.1.p038

Bárkányi, Z. (2021). Motivation, self-efficacy beliefs, and speaking anxiety in language

MOOCs. ReCALL, 33(2), 143-160. https://doi.org/10.1017/S0958344021000033

- Barron, K. E., & Harackiewicz, J. M. (2003). Revisiting the benefits of performance-approach goals in the college classroom: Exploring the role of goals in Advanced College courses. *International Journal of Educational Research*, 39(4–5), 357–374. https://doi.org/10.1016/j.ijer.2004.06.004
- Barrows, J., Dunn, S., & Lloyd, C. A. (2013). Anxiety, self-efficacy, and college exam grades. Universal Journal of Educational Research, 1(3), 204–208.

https://doi.org/10.13189/ujer.2013.010310

- Basar, Z. M., Mansor, A. N., Jamaludin, K. A., & Alias, B. S. (2018). The effectiveness and challenges of online learning for secondary school students: a case study *Asian Journal of University Education*, 17(3), 1–11. https://doi.org/10.36892/ijlls.v2i4.404
- Bashir, A., Bashir, S., Rana, K., Lambert, P., & Vernallis, A. (2021). Post-COVID-19 adaptations; the shifts towards online learning and hybrid course delivery, and the implications for

biosciences courses in the higher education setting *Frontiers in Education*, 6(7), 711619. https://doi.org/10.3389/feduc.2021.711619

- Batez, M. (2021). ICT skills of university students from the faculty of sport and physical education during the COVID-19 pandemic *Sustainability*, *13*(4), 1711.
- https://doi.org/10.3390/su13041711
- Becker, G. (2000). Creating comparability among reliability coefficients: The case of Cronbach alpha and Cohen kappa. *Psychological Reports*, 87(3, Pt 2), 1171– 1182. https://doi.org/10.2466/PR0.87.7.1171-1182
- Bell, A. A. (2022). Social desirability and other predictors of statistics anxiety at the graduate level (Publication No. 28969554) [Doctoral dissertation, Capella University]. ProQuest Dissertations & Theses Global https://www.proquest.com/dissertationstheses/socialdesirability-other-predictors-statistics/docview/2644001158/se-2
- Benita, M., & Matos, L. (2021). Internalization of mastery goals: The differential effect of teachers' autonomy support and Control. *Frontiers in Psychology*, 11. https://doi.org/10.3389/fpsyg.2020.599303
- Bestgen, Y., & Vincze, N. (2012). Checking and bootstrapping lexical norms by means of word similarity indexes. *Behavior research methods*, *44*, 998-1006.
- Beyth-Marom, R., Fidler, F., & Cumming, G. (2008). Statistical cognition: Towards evidencebased practice in statistics and statistics education *Statistics Education Research Journal*, 7(2), 20–39. https://doi.org/10.52041/serj.v7i2.468
- Bhardwaj, P. (2019). Types of sampling in research. *Journal of the Practice of Cardiovascular Sciences*, 5(3), 157.
- Bjørnebekk, G., Diseth, Å., & Ulriksen, R. (2013). Achievement motives, self-efficacy, achievement goals, and academic achievement at multiple stages of education: A longitudinal analysis. *Psychological reports*, *112*(3), 771-787.

https://doi.org/10.2466/14.09.PR0.112.3.771-787

- Blackburn, G. (2015). Effectiveness of eLearning in statistics: Pictures and stories. E-Learning and Digital Media, 12(5–6), 459–480. <u>https://doi.org/10.1177/2042753016653704</u>
- Bong, M., Cho, C., Ahn, H. S., & Kim, H. J. (2012). Comparison of self-beliefs for predicting student motivation and achievement *The Journal of Educational Research*, *105*(5), 336-
352. https://psycnet.apa.org/doi/10.1080/00220671.2011.627401

- Bonk, C. J., & Lee, M. M. (2017). Motivations, achievements, and challenges of self-directed informal learners in open educational environments and moocs. *Journal of Learning for Development*, 4(1). <u>https://doi.org/10.56059/jl4d.v4i1.195</u>
- Boyadjieva, P., & Ilieva-Trichkova, P. (2017). Between inclusion and fairness: Social justice perspective to participation in adult education. *Adult Education Quarterly*, 67(2), 97–117. https://doi.org/10.1177/0741713616685398
- Boyer, S. L., Edmondson, D. R., Artis, A. B., & Fleming, D. (2014). Self-directed learning: A tool for lifelong learning. *Journal of Marketing Education*, 36(1), 20–32.
- https://doi.org/10.1177/0273475313494010
- Brookfield, S. (1993). Self-directed learning, political clarity, and the critical practice of adult education *Adult Education Quarterly*, *43*, 227-242.
- Brown, A. L., & Palincsar, A. S. (1989). Guided, cooperative learning, and individual knowledge acquisition. In L. B. Resnick (Ed.), Knowing, Learning, and Instruction: Essays in Honor of Robert Glaser (pp. 393–451). Lawrence Erlbaum Associates, Inc.
- Brophy, J. (2004). (2nd ed.). Lawrence Erlbaum Associates Publishers.
- Bruning, R., Dempsey, M., Kauffman, D. F., McKim, C., & Zumbrunn, S. (2013). Examining dimensions of self-efficacy for writing. Journal of Educational Psychology, 105(1), 25.
- Bruning, R., & Horn, C. (2000). Developing motivation to write Educational psychologist, 35, 25-37. https://doi.org/10.1207/S15326985EP3501_4
- Bryant, Salina K., "Self-Efficacy Sources and Academic Motivation: A Qualitative Study of 10th Graders" (2017). Electronic Theses and Dissertations Paper 3231.

https://dc.etsu.edu/etd/3231

Buhr, E. E., Daniels, L. M., & Goegan, L. D. (2019). Cognitive appraisals mediate relationships between two basic psychological needs and emotions in a massive open online course. *Computers in Human Behavior*, 96(1), 85–94.

https://doi.org/10.1016/j.chb.2019.02.009

Butler, R. (1987). Task-involving and ego-involving properties of evaluation: effects of different feedback conditions on motivational perceptions, interest, and performance *Journal of Educational Psychology*, 79(4), 474–482. https://doi.org/10.1037/0022-0663.79.4.474

- Caffarella, R. S. (1993). Self-directed learning. New Directions for Adult and Continuing Education, 1993(57), 25–35. https://doi.org/10.1002/ace.36719935705
- Cai, L., Chung, S. W., & Lee, T. (2021). Incremental model fit assessment in the case of categorical data: Tucker–Lewis index for item response theory modeling *Prevention Science*, 1–12. https://doi.org/10.1007/s11121-021-01253-4
- Canals, L. (2017). Instruments for gathering data In E. Moore & M. Dooly (Eds.), *Qualitative approaches to research on plurilingual education* (pp. 390–401). Researchpublishing.net. https://doi.org/10.14705/ rpnet.2017.emmd2016.637
- Candy, P. C. (1991). Self-direction for lifelong learning: a comprehensive guide to theory and practice. Jossey-Bass.
- Carpenter, S. L. (2007). A comparison of the relationships between students' self-efficacy, goal orientation, and achievement across grade levels: A meta-analysis [master's thesis, Simon Fraser University]. Summit Research Repository. https://summit.sfu.ca/item/2661
- Cazan, A. M. (2014, July). Self-regulated learning and academic achievement in the context of online learning environments In I. Roceanu (Ed.), *10th international conference*
- "eLearning and software for education": Vol. 3. Let's build the future through learning innovation! (pp. 90–95). Editura Universitatii Nationale de Aparare "Carol I". https://doi.org/10.12753/2066-026X-14-153
- Cecilia Titiek Murniati*, Heny Hartono, and Agus Cahyo Nugroho, (2022), "Self-directed Learning, Self-efficacy, and Technology Readiness in E-learning Among University
- Students" in 4th International Conference on Education and Social Science Research
- (ICESRE), KnE Social Sciences, pages 213-224. DOI 10.18502/kss.v7i14.11970
- Charokar, C. K. (2022). Self-directed Learning Theory to Practice: A Footstep towards the Path of Being a Life-long Learner. *Journal of Advances in Medical Education & Professionalism*, 10(3), 135–144. https://doi.org/10.30476/JAMP.2022.94833.1609
- Chazan, D. J., Pelletier, G. N., & Daniels, L. M. (2022). Achievement goal theory review: an application to school psychology *Canadian Journal of School Psychology*, 37(1), 40–56. <u>https://doi.org/10.1177/08295735211058319</u>
- Chemers, M. M., Hu, L.-tze, & Garcia, B. F. (2001). Academic self-efficacy and first year college student performance and adjustment. *Journal of Educational Psychology*, *93*(1),

55-64. https://doi.org/10.1037/0022-0663.93.1.55

- Chen, C. H., Chen, K. Z., & Tsai, H. F. (2022). Did Self-Directed Learning Curriculum Guidelines Change Taiwanese High-School Students' Self-Directed Learning Readiness? *The Asia-Pacific Education Researcher*, 31(4), 409-426.
- Cho, M.-K.; Kim, M.Y. Factors Affecting Learning Satisfaction in Face-to-Face and Non-Faceto-Face Flipped Learning among Nursing Students. Int. J. Environ. Res. Public Health 2021, 18, 8641. https://doi.org/10.3390/ ijerph18168641

Cho, M. H., & Shen, D. (2013). Self-regulation in online learning *Distance Education*, *34*(3), 290–301. https://doi.org/10.1080/01587919.2013.835770

- Chou, P. (2012). The relationship between engineering students self-directed learning abilities and online learning performances: A pilot study. *Contemporary Issues in Education Research (CIER)*, 5(1), 33. https://doi.org/10.19030/cier.v5i1.6784
- Chu, R. J., & Tsai, C. (2009). Self-directed learning readiness, internet self-efficacy and preferences towards constructivist internet-based learning environments among higheraged adults. *Journal of Computer Assisted Learning*, 25(5), 489-501.
- Chu, R. J., & Tsai, C. (2009). Self-directed learning readiness, internet self-efficacy and preferences towards constructivist internet-based learning environments among higheraged adults. *Journal of Computer Assisted Learning*, 25(5), 489-501. https://doi.org/10.1111/j.1365-2729.2009.00324.x

Cleary, T. J., & Chen, P. P. (2009). Self-Regulation, Motivation, and Math Achievement in

- Middle School: Variations across Grade Level and Math Context. *Journal of School Psychology*, 47, 291-314. https://doi.org/10.1016/j.jsp.2009.04.002
- Clerge, J. (2019). An Investigation into Self-Efficacy and Academically Successful Minority Students Honors Thesis *BSU Honors Program Theses and Projects* https://vc.bridgew.edu/honors_proj/360
- Code, J. (2020). Agency for learning: Intention, motivation, self-efficacy, and selfregulation. *Frontiers in Genetics*, *5*, 19. https://doi.org/10.3389/feduc.2020.00019 Confessore, G. J., & Confessore, S. J. (1992). *Guideposts to self-directed learning: Expert commentary on essential concepts*. Organization Design and Development.

- Coros, J. D., & Madrigal, D. V. (2021). Self-directed learning, self-efficacy in learning, and academic motivation of public senior high school students *Asian Journal of Education and Social Studies*, 19–34. https://doi.org/10.9734/ajess/2021/v21i230503
- Cottrell, C. (2018). Self-efficacy, implicit theories of ability, and 2 x 2 achievement goal orientation: A mediation analysis in collegiate athletics [Master's thesis, Georgia Southern University]. Digital Commons. https://digitalcommons.georgiasouthern.edu/etd/1786
- Crippen, K. J., Biesinger, K. D., Muis, K. R., & Orgill, M. (2009). The role of goal orientation and self-efficacy in learning from web-based work examples *Journal of Interactive Learning Research*, 20(4), 385–403.
- Dangol, R., & Shrestha, M. (2019). Learning readiness and educational achievement among school students *The International Journal of Indian Psychology*, 7(2), 467–476.
- Davidson, R., & Flachaire, E. (2008). The wild bootstrap, tamed at last. *Journal of Econometrics*, *146*(1), 162-169.
- Deci, E. L., & Ryan, R. M. (1985). Intrinsic Motivation and Self-Determination in Human Behavior. Berlin: Springer Science & Business Media. https://doi.org/10.1007/978-1-4899-2271-7
- Deen, M., & de Rooij, M. (2020). ClusterBootstrap: An R package for the analysis of hierarchical data using generalized linear models with the cluster bootstrap. *Behavior Research Methods*, 52, 572-590.
- Delice, A. (2010). The Sampling Issues in Quantitative Research. *Educational Sciences: Theory and Practice*, *10*(4), 2001-2018.
- DeVaney, T. A. (2010) Anxiety and Attitude of Graduate Students in On-Campus vs. Online Statistics Courses, *Journal of Statistics Education*, 18:1, DOI: 10.1080/10691898.2010.11889472
- Diaconu-Gherasim, L. R., Măirean, C., & Brumariu, L. E. (2019). Quality of teachers' and peers' behaviors and achievement goals: the mediating role of self-efficacy *Learning and Individual Differences*, 73(1), 147–156. https://doi.org/10.1016/j.lindif.2019.06.001

- Dikmen, M., & Bahadır, F. (2021). University students' views on the effectiveness of learning through homework *International Online Journal of Educational Sciences*, 13(3), 689– 704.
- Douglas, E. (2010). *A study of barriers to adult self-directed learning* (Order No. 3425722). Available from ProQuest Dissertations & Theses Global. (757725252).
- Doménech-Betoret, F., Abellán-Roselló, L., & Gómez-Artiga, A. (2017). Self-efficacy, satisfaction, and academic achievement: the mediator role of students' expectancy-value beliefs *Frontiers in Psychology*, 8(1), 1193. http://dx.doi.org/10.3389/fpsyg.2017.01193
- Doménech-Betoret, F., Gómez-Artiga, A., & Lloret-Segura, S. (2014). Personal variables, motivation and avoidance learning strategies in undergraduate students. *Learning and Individual Differences*, *35*(1), 122–129. https://doi.org/10.1016/j.lindif.2014.06.007
- Dresler, T., Schecklmann, M., Ernst, L. H., Pohla, C., Warrings, B., Fischer, M., Polak, T. & Fallgatter, A. J. (2012). Recovery of cortical functioning in abstinent alcohol-dependent patients: prefrontal brain oxygenation during verbal fluency at different phases during withdrawal *The World Journal of Biological Psychiatry*, *13*(2), 135–145. https://doi.org/10.3109/15622975.2011.564654
- Du, F. (2013). Student perspectives on self-directed language learning: implications for teaching and research *International Journal for the Scholarship of Teaching and Learning*, 7(2),
- 24. https://doi.org/10.20429/ijsotl.2013.070224
- du Toit-Brits, C. (2019). Unleashing the power of self-directed learning: Criteria for structuring self-directed learning within the learning environments of higher education institutions. *Africa Education Review*, *17*(2), 20–32. https://doi.org/10.1080/18146627.2018.1494507
- Dubey, A. (2000). The effect of goal orientation on life satisfaction: a study of students in higher education In N. K. Garg, R. K. Gupta, S. K. Mangla, & N. Jundal (Eds.), *Global Economic Order in the Post-COVID-19 Era: Challenges, Opportunities, and Strategies* (pp. 350–355). Maharaja Agrasen University Publication
- Dunn, K. (2014). Why Wait? The Influence of Academic Self-Regulation, Intrinsic Motivation, and Statistics Anxiety on Procrastination in Online Statistics Innovative Higher Education, 39, 33–44. <u>https://doi.org/10.1007/s10755-013-9256-1</u>

Dweck, C. S. (1986). Motivational processes affecting learning. American Psychologist, 41(10),

1040-1048. https://doi.org/10.1037/0003-066X.41.10.1040

Eccles, J. S., & Wigfield, A. (2002). Motivational beliefs, values, and goals Annual Review of

Psychology, 53(1), 109–132. https://doi.org/10.1146/annurev.psych.53.100901.135153

- Eccles, J., & Wigfield, A. (2001). Development of academic achievement motivation In J. D.
 Wright (Ed.), *International Encyclopedia of the Social & Behavioral Sciences* (2nd ed., pp. 14–20). Elsevier.
- Elliot, A. J., & Harackiewicz, J. M. (1996). Approach and avoidance of achievement goals and intrinsic motivation: a mediational analysis *Journal of Personality and Social Psychology*, 70(3), 461–475. https://doi.org/10.1037/0022-3514.70.3.461

Elliot, A. J., & McGregor, H. A. (2001). A 2×2 achievement goal framework. *Journal of Personality and Social Psychology*, *80*(3), 501–519. https://doi.org/10.1037/0022-3514.80.3.501

Erturan, G., McBride, R., & Agbuga, B. (2020). Self-regulation and self-efficacy as mediators of achievement goals and leisure-time physical activity: a proposed model *Pedagogy of*

Physical Culture and Sports, 24(1), 12–20. https://doi.org/10.15561/26649837.2020.0102

- Eschenbacher, S., & Fleming, T. (2020). Transformative dimensions of lifelong learning: Mezirow, Rorty, and COVID-19. *International Review of Education*, 66(5), 657–672. https://doi.org/10.1007/s11159-020-09859-6
- Etikan, I., Musa, S. A., & Alkassim, R. S. (2016). Comparison of convenience sampling and purposive sampling. *American journal of theoretical and applied statistics*, 5(1), 1-4.
- Fan, J. Y., Tseng, Y. J., Chao, L. F., Chen, S. L., & Jane, S. W. (2020). Learning outcomes of a flipped classroom teaching approach in an adult-health nursing course: A quasiexperimental study. *BMC Medical Education*, 20(1), 1–11.

https://doi.org/10.1186/s12909-020-02240-z

- Field, A. (2013). Discovering statistics using IBM SPSS Statistics (4th ed.). Thousand Oaks, CA: Sage.
- Feldhammer-Kahr, M., Arendasy, M., & Paechter, M. (2022). Students perceived academic stress, a sense of belonging, adaptability, sports, and depression in the second year of the pandemic. In C. Pracana & M. Wang (Eds.), *Psychological Applications and Trends* (pp.

84-88). inScience Press. https://doi.org/10.36315/2022inpact018

- Field, A. (2013). Discovering statistics using IBM SPSS Statistics (4th ed.). Thousand Oaks, CA: Sage.
- Figueroa-Cañas, J., & Sancho-Vinuesa, T. (2020). Early prediction of dropout and final exam performance in an online statistics course. *IEEE Revista Iberoamericana de Tecnologias del Aprendizaje*, 15(2), 86–94. https://doi.org/10.1109/RITA.2020.2987727

Finney, S. J., & Schraw, G. (2003). Self-efficacy beliefs in college statistics courses.

Contemporary Educational Psychology, 28(2), 161–186.

https://doi.org/10.1016/S0361476X(02)00015-2

- Finnegan, M. (2022). A curriculum for the new era of progressive, self-directed learning. *Childhood Education*, 98(3), 6-15. https://doi.org/10.1080/00094056.2022.2083381
- Fisher, M., King, J., & Tague, G. (2001). Development of a self-directed learning readiness scale for nursing education *Nurse Education Today*, *21*(7), 516–525.

https://doi.org/10.1054/nedt.2001.0589

- Fleming, J., & Ward, D. (2017). Self-directed Groupwork social justice through social action and empowerment. *Critical and Radical Social Work*, 5(1), 75–91. https://doi.org/10.1332/204986016x14822509544479
- Frambach, J.M., Driessen, E.W., Chan, L.C., & van der Vleuten, C.P. (2012). Rethinking the globalization of problem-based learning: how culture challenges self-directed learning Medical Education, 46: 738-47.
- Gagnon, M. P., Gagnon, J., Desmartis, M., & Njoya, M. (2013). The impact of blended teaching on knowledge, satisfaction, and self-directed learning in nursing undergraduates: a randomized, controlled trial *Nursing Education Perspectives*, 34(6), 377–382.

https://doi.org/10.5480/10-459

Galán, C. A., Feldman, J. S., & McClaine, R. N. (2022). Using the social information processing model to understand gender differences in the manifestation and frequency of aggression *Aggression and Violent Behavior*, 66, 101766.

https://doi.org/10.1016/j.avb.2022.101766

Gan Z, An Z and Liu F (2021) Teacher Feedback Practices, Student Feedback Motivation, and Feedback Behavior: How Are They Associated With Learning Outcomes? Front. Psychol. 12:697045. doi: 10.3389/fpsyg.2021.697045

Gannouni, K., & Ramboarison-Lalao, L. (2018). Leadership and students' academic success:

Mediating effects of self-efficacy and self-determination. International Journal of

Leadership in Education, 21(1), 66-79. https://doi.org/10.1080/13603124.2015.1123300

- Geitz, G., Joosten-ten Brinke, D., & Kirschner, P. A. (2016). Changing learning behavior: selfefficacy and goal orientation in PBL groups in higher education *International Journal of Educational Research*, 75, 146–158. https://doi.org/10.1016/j.ijer.2015.11.001
- Gerard, L., Bradford, A., & Linn, M. C. (2022). Supporting Teachers to Customize Curriculum for Self-Directed Learning. *Journal of Science Education and Technology*, 31(5), 660– 679. https://doi.org/10.1007/s10956-022-09985-w
- Gomez, F. C., Trespalacios, J., Hsu, Y. C., & Yang, D. (2022). Exploring teachers' technology integration self-efficacy through the 2017 ISTE standards *TechTrends*, 66(2), 159–171. https://doi.org/10.1007/s11528-021-00639-z
- Gopal, K., Salim, N. R., & Ayub, A. F. M. (2018, November). Influence of self-efficacy and attitudes towards statistics on undergraduates' statistics engagement in a Malaysian public university. In C. Y. Chen, L. S. Lee, A. Kilicman, F. Ismail, H. Midi, I. Garfurjan, S. K. S. Husain, & S. N. I. Ibrahim (Eds.), *Journal of physics: Conference series: Vol.*

1132 (Nr. 012042). IOP Publishing. https://doi.org/10.1088/1742-6596/1132/1/012042

Gore, P. A., Jr. (2006). Academic Self-Efficacy as a Predictor of College Outcomes: Two

Incremental Validity Studies. Journal of Career Assessment, 14(1), 92-

115. https://doi.org/10.1177/1069072705281367

- Gray, D. (2018). Determining the role self-directed learning readiness and internet self-efficacy have on online learning self-efficacy for the purpose of increasing persistence in online courses for adult learners [Doctoral dissertation, Boise State University]. Scholar Works. https://doi.org/10.18122/td/1473/boisestate
- Gray, J. A., & DiLoreto, M. (2016). The effects of student engagement, student satisfaction, and perceived learning in online learning environments *International Journal of Educational Leadership Preparation*, 11(1), 98–119.

Gregorich, M., Strohmaier, S., Dunkler, D., & Heinze, G. (2021). Regression with highly correlated predictors: variable omission is not the solution. *International Journal of Environmental Research and Public Health*, 18(8), 4259.

https://doi.org/10.3390/ijerph18084259

Gresham, J. D. (2019). Self-directed learning: Empowering authentic learner autonomy through self-agency in the secondary school learning environment (Publication No. 13882012)
[Doctoral dissertation, California Institute of Integral Studies]. ProQuest Dissertations & Theses Global https://www.proquest.com/dissertations-theses/self-directed-learningempowering-authentic/docview/2235775839/se-2

Grow, G. O. (1991). Teaching learners to be self-directed. *Adult Education Quarterly*, *41*(3), 125–149. https://doi.org/10.1177/0001848191041003001

- Grow, G. O. (1994). In defense of the staged self-directed learning model. Adult Education Quarterly, 44: 109-114.
- Gruber, S., Rosca, R. I., Chazan, D., Fleming, E., Balady, S., VanNetta, C., & Okoudjou, K. A. (2021). Active learning in an undergraduate precalculus course: Insights from a course redesign. *PRIMUS*, 31(3-5), 358–370. <u>https://doi.org/10.1080/10511970.2020.1772920</u>
- Guglielmino, L.M. (2013). The case for promoting self-directed learning in formal educational institutions *SA-Educ Journal*, 10: 1-18.
- Gutiérrez, M., & Tomás, J. M. (2019). The role of perceived autonomy support in predicting university students' academic success is mediated by academic self-efficacy and school engagement. *Educational Psychology*, 39(6), 729-748.

https://doi.org/10.1080/01443410.2019.1566519

- Hatcher, T. G. (1997). The ins and outs of self-directed learning. *Training & Development*, 51(2), 34-36+.
- Hair, J. F., Black, W. C., Babin, B. J., Anderson, R. E., & Tatham, R. L. (2006). Multivariate data analysis (6th ed.). Upper Saddle River, NJ: Pearson/Prentice Hall.
- Hair, J., Black, W., Babin, B., & Anderson, R. (2009). Multivariate data analysis (7th ed.). Upper Saddle River, NJ: Prentice Hall.
- Hair, J. F., Black, W. C., Babin, B. J., & Anderson, R. E. (2010). Multivariate data analysis (7th ed.). Upper Saddle River, NJ: Pearson/Prentice Hall

- Hayat, A. A., Shateri, K., Amini, M., & Shokrpour, N. (2020). Relationships between academic self-efficacy, learning-related emotions, and metacognitive learning strategies and academic performance in medical students: a structural equation model *BMC Medical Education*, 20(1), 1-11. https://doi.org/10.1186/s12909-020-01995-9
- Hayes, A. F. (2013). Introduction to mediation, moderation, and conditional process analysis: a regression-based perspective New York: The Guilford Press
- Hiemstra, R. (1994). Self-directed learning. T. Husen & T. N. Postlethwaite (Eds.), The International Encyclopedia of Education (second edition), Oxford: Pergamon Press

Hsieh, P. (Pei-H., Sullivan, J. R., & Guerra, N. S. (2007). A Closer Look at College Students:

Self-efficacy and goal orientation. Journal of Advanced Academics, 18(3), 454-

476. https://doi.org/10.4219/jaa-2007-500

Hoegler, S., & Nelson, M. (2018). The influence of anxiety and self-efficacy on statistical performance: a path analysis *Psi Chi Journal of Psychological Research*, *23*(5), 364–

399. https://doi.org/10.24839/2325-7342.JN23.5.364

Hommel, B. (2022). GOALIATH: A theory of goal-directed behavior Psychological

Research, 86(4), 1054–1077. https://doi.org/10.1007/s00426-021-01563-w

Hoogerheide, V., van Wermeskerken, M., van Nassau, H., & van Gog, T. (2018). Modelobserver similarity and task-appropriateness in learning from video modelling examples: Do model and student gender affect test performance, self-efficacy, and perceived competence? *Computers in Human Behavior*, 89(1), 457–464.

https://doi.org/10.1016/j.chb.2017.11.012

- Hoque, A. S., Awang, Z., Jusoff, K., Salleh, F., & Muda, H. (2017). Social Business Efficiency: Instrument Development and Validation Procedure Using Structural Equation Modeling. *International Business Management*, 11, 222-231.
- Houle, C. O. (1961). *The inquiring mind: A study of the adult who continues to learn* University of Wisconsin Press
- Hoy, W. K., Tarter, C. J., & Hoy, A. W. (2006). Academic optimism of schools: A force for student achievement. *American Educational Research Journal*, 43(3), 425–446. https://doi.org/10.3102/00028312043003425

Hsia, L. H., Huang, I., & Hwang, G. J. (2016). Effects of different online peer-feedback approaches on students' performance skills, motivation, and self-efficacy in a dance course. *Computers & Education*, 96(1), 55–71.

https://doi.org/10.1016/j.compedu.2016.02.004

- Hsieh, P., Sullivan, J. R., & Guerra, N. S. (2007). A closer look at college students: Self-efficacy and goal orientation. *Journal of Advanced Academics*, 18(3), 454–476. https://doi.org/10.4219/jaa-2007-500
- Huang, C. (2016). Achievement goals and self-efficacy: a meta-analysis *Educational Research Review*, *19*(1), 119–137. https://doi.org/10.1016/j.edurev.2016.07.002
- Hung, M.-L., Chou, C., Chen, C.-H., & Own, Z.-Y. (2010). Learner readiness for online learning: Scale development and student perceptions. *Computers & Education*, 55(3),
- 1080-1090. https://doi.org/10.1016/j.compedu.2010.05.004
- International Business Machines Corporation. (2022). *Downloading IBM SPSS Statistics 26*. https://www.ibm.com/support/pages/downloading-ibm-spss-statistics-26
- Iroegbu, M. N. (2015). Self-efficacy and work performance: a theoretical framework of Albert Bandura & Apos's model, review of findings, implications, and directions for future research *Psychology and Behavioral Sciences*, 4(4), 170. https://doi.org/10.11648/j.pbs.20150404.15
- Iwuanyanwu, P. N. (2022). Facilitating problem-solving in a university undergraduate physics classroom: the case of students' self-efficacy *Interdisciplinary Journal of Environmental* and Science Education, 18(2), e2270.

https://doi.org/10.21601/ijese/11802

- Jager, J., Putnick, D. L., & Bornstein, M. H. (2017). II. More than just convenient: The scientific merits of homogeneous convenience samples. *Monographs of the Society for Research in Child Development*, 82(2), 13-30.
- Jan, S. K. (2015). The relationships between academic self-efficacy, computer self-efficacy, prior experience, and satisfaction with online learning *American Journal of Distance*
- Education, 29(1), 30-40. https://doi.org/10.1080/08923647.2015.994366
- Jang, H., & Pak, S. (2017). Perfectionism and high school adjustment: Self-directed learning strategies as a mediator. *Journal of Asia Pacific Counseling*, 7(1), 63–78.

https://doi.org/10.18401/2017.7.1.6

- Jawale, K. V. (2012). Methods of sampling design in the legal research: Advantages and disadvantages. *Online International Interdisciplinary Research Journal*, *2*(6), 183-190.
- Jones, T. L., Baxter, M. A. J., & Khanduja, V. (2013). A quick guide to survey research *The Annals of the Royal College of Surgeons of England*, 95(1), 5–7. https://doi.org/10.1308/003588413X13511609956372
- Joo, B. K., Park, S., & Oh, J. R. (2013). The effects of learning goal orientation, developmental needs awareness, and self-directed learning on career satisfaction in the Korean public sector *Human Resource Development International*, 16(3), 313–329. https://doi.org/10.1080/13678868.2013.782993
- Jordan, K. (2015). Massive open online course completion rates revisited: assessment, length, and attrition *International Review of Research in Open and Distance Learning*, *16*(3),
- 341-358. https://doi.org/10.19173/irrodl.v16i3.2112
- Jung, Y., & Lee, J. (2018). Learning engagement and persistence in massive open online courses (MOOCS). Computers & Education, 122(1), 9–22. https://doi.org/10.1016/j.compedu.2018.02.013
- Kärchner, H., Schöne, C., & Schwinger, M. (2021). Beyond the level of self-esteem: exploring the interplay of level, stability, and contingency of self-esteem, mediating factors, and academic achievement *Social Psychology of Education*, 24(2), 319–341. https://doi.org/10.1007/s11218-021-09610-5
- Keklik, D. E., & Keklik, İ. (2013). Motivation and learning strategies as predictors of high school students' math achievement *Cukurova University Faculty of Education Journal*, 42(1), 96–109.
- Kelly, R. (2021). 73 percent of students prefer some courses be fully online post-pandemic. Campus Technology. https://campustechnology.com/articles/2021/05/13/73-percentofstudents-prefer-some-courses-be-fully-online-post-
- pandemic.aspx#:~:text=73%20Percent%20of%20Students%20Prefer%20Some%20Cour ses%20Be%20Fully%20Online%20Post%2DPandemic,-
- By%20Rhea%20Kelly&text=In%20a%20recent%20survey%2C%20nearly,courses%20f ully%20online%20post%2Dpandemic

- Kemp, N., & Grieve, R. (2014). Face-to-face or face-to-screen? Undergraduates' opinions and test performance in classroom vs. online learning. *Frontiers in Psychology*, 5, 1-11. https://doi.org/10.3389/fpsyg.2014.01278
- Keskin, H. K. (2014). A path analysis of metacognitive strategies in reading, self-efficacy, and task value *International Journal of Social Science & Education*, *4*(4), 798–808.
- Khalid, M., Bashir, S., & Amin, H. (2020). Relationship between self-directed learning (SDL) and academic achievement of university students: A case of online distance learning and traditional universities *Bulletin of Education and Research*, *42*(2), 131–148.
- Khiat, H. (2017). Academic performance and the practice of self-directed learning: the adult student perspective *Journal of Further and Higher Education*, 41(1), 44–59. https://doi.org/10.1080/0309877X.2015.1062849
- Khodaei, S., Hasanvand, S., Gholami, M., Mokhayeri, Y., & Amini, M. (2022). The effect of the online flipped classroom on self-directed learning readiness and metacognitive awareness in nursing students during the COVID-19 pandemic *BMC Nursing*, 21(1), 1–10.

https://doi.org/10.1186/s12912-022-00804-6

Kim, T. K., & Park, J. H. (2019). More about the basic assumptions of the t-test: normality and sample size *Korean Journal of Anesthesiology*, 72(4), 331–335.

https://doi.org/10.4097/kja.d.18.00292

- Kizilgunes, B., Tekkaya, C., & Sungur, S. (2009a). Modeling the relations among students' epistemological beliefs, motivation, learning approach, and achievement. *The Journal of Educational Research*, 102(4), 243–256. https://doi.org/10.3200/joer.102.4.243-256
- Klassen, R. M., & Lynch, S. L. (2007). Self-efficacy from the perspective of adolescents with learning disabilities and their specialist teachers *Journal of Learning Disabilities*, 40, 494-507.
- Klein, H. J., Noe, R. A., & Wang, C. (2006). Motivation to learn and course outcomes: the impact of delivery mode, learning goal orientation, and perceived barriers and enablers *Personnel Psychology*, 59(3), 665–702. https://doi.org/10.1111/j.17446570.2006.00050.x
- Kline, R. B. (1998). Software review: Software programs for structural equation modelling: Amos, EQS, and LISREL. *Journal of Psychoeducational Assessment*, 16(4), 343–364.
 https://doi.org/10.1177/073428299801600407

- Kline, R. (2011). Principles and practice of structural equation modeling (3rd ed.). New York, NY: The Guilford Press.
- Knowles, M. (1975). Self-directed learning: A guide for learners and teachers New York, NY: Association Press.
- Koç, S. E. (2019). The relationship between emotional intelligence, self-directed learning readiness, and achievement *International Online Journal of Education and Teaching*, 6(3), 672–688.
- KOÇ, K., & TURAN, M. B. (2018). The impact of cultural intelligence on social skills among university students *Journal of Education and Learning*, 7(6), 241. https://doi.org/10.5539/jel.v7n6p241
- Kock, N., & Hadaya, P. (2018). Minimum sample size estimation in PLS-SEM: The inverse square root and gamma-exponential methods. *Information systems journal*, 28(1), 227-261.
- Koehler, N., Correia, A.-P., Alpay, N., & LeVally, C. (2018). Determining the cognitive value of online interactive multimedia in statistics education. *International Conference on Computer-Supported Education*, 865, 138-153.
- Kohan, Noushin, et al. "Self- Directed Learning Barriers in a Virtual Environment: A Qualitative Study." *Journal of Advances in Medical Education & Professionalism*, vol. 5, no. 3, 2017, pp. 116–123, www.ncbi.nlm.nih.gov/pmc/articles/PMC5522903/. Accessed 12 Oct. 2022.
- Kock, N., & Hadaya, P. (2018). Minimum sample size estimation in PLS-SEM: The inverse square root and gamma-exponential methods. *Information Systems Journal*, 28(1), 227-261.
- Korn, R. M., & Elliot, A. J. (2016). The 2 × 2 standpoints model of achievement goals *Frontiers in Psychology*, 7, 1–12. https://doi.org/10.3389/fpsyg.2016.00742
- Kuo, Y. C., Walker, A. E., Belland, B. R., & Schroder, K. E. (2013). A predictive study of student satisfaction in online education programs *International Review of Research in Open and Distributed Learning*, 14(1), 16–39. https://doi.org/10.19173/irrodl.v14i1.1338
- Labonté, C., & Smith, V. R. (2022). Learning through technology in middle school classrooms: Students' perceptions of their self-directed and collaborative learning with and without

technology. *Education and Information Technologies*, 27(5), 6317-6332. https://doi.org/10.1007/s10639-021-10885-6

- Lai, Y., Saab, N., & Admiraal, W. (2022). Learning Strategies in Self-directed Language
 Learning Using Mobile Technology in Higher Education: A Systematic Scoping Review.
 Education and Information Technologies, 27, 7749–7780.
 https://link.springer.com/content/pdf/10.1007/s10639-022-10945-5.pdf
- Landry, C. C. (2003). Self-efficacy, motivation, and outcome expectation correlates of college students' intention certainty. *LSU Doctoral Dissertations*. 1254.

https://digitalcommons.lsu.edu/gradschool_dissertations/1254

- Larwin, K. & Larwin, D. (2011). A Meta-Analysis Examining the Impact of Computer-Assisted Instruction on Postsecondary Statistics Education: 40 Years of Research Journal of Research on Technology in Education, 43(3), 253-278.
- Lauermann, F., & König, J. (2016). Teachers' professional competence and wellbeing: understanding the links between general pedagogical knowledge, self-efficacy, and burnout *Learning and Instruction*, 45(1), 9–19.

https://doi.org/10.1016/j.learninstruc.2016.06.006

Leach, L. (2000). Self-directed learning: theory and practice

Lemmetty, S. (2021). Employee opportunities for self-directed learning at technology organisations : features and frames of self-directed learning projects. *Studies in Continuing Education*, 43(2), 139-155. https://doi.org/10.1080/0158037X.2020.1765758

Li, H., Majumdar, R., Chen, M. R. A., & Ogata, H. (2021). Goal-oriented active learning (GOAL) system to promote reading engagement, self-directed learning behaviour, and motivation through extensive reading. *Computers & Education*, *171*, 104239.

https://doi.org/10.1016/j.compedu.2021.104239

- Liaw, S.-S., Huang, H.-M., & Chen, G.-D. (2007). Surveying instructor and learner attitudes toward e-learning. *Computers & Education*, 49(4), 1066–1080. https://doi.org/10.1016/j.compedu.2006.01.001
- Lin, Y., Cheng, Y., and Zhang, Y. (2021) A Study of the Relationship between Achievement Goal Orientation on Online Academic Procrastination among Junior High School Students

Multiple mediation analysis of task value and motivational regulation, ACM Digital Library, https://dl.acm.org/doi/fullHtml/10.1145/3474995.3475032#bib16

- Liu, H., Wang, T., Lin, H. K., Lai, C., & Huang, Y. (2022). The influence of affective feedback adaptive learning system on learning engagement and self-directed learning. *Frontiers in Psychology*, 13, 858411. https://doi.org/10.3389/fpsyg.2022.8584114
- Loehlin, J. (2004). Latent variable models: An introduction to factor, path, and structural equation analysis (4th ed). Mahwah, NJ: Lawrence Erlbaum Associates. Psychology, 5, 1-11. https://doi.org/10.3389/fpsyg.2014.01278
- Li, J., Yang, D., & Hu, Z. (2022). Wuhan college students' self-directed learning and academic performance: Chain-mediating roles of optimism and mental health. *Frontiers in Psychology*, 12, 1-12. <u>https://doi.org/10.3389/fpsyg.2021.757496</u>
- Lin, X., Su, S., & McElwain, A. (2019). Academic stressors as predictors of achievement goal orientations of American and ESL international students. *Journal of International Students*, 9(4), 1134–1154. https://doi.org/10.32674/jis.v9i4.752
- Lin, T. J. (2021). Exploring the differences in Taiwanese university students' online learning task value, goal orientation, and self-efficacy before and after the COVID-19 outbreak *The Asia-Pacific Education Researcher*, 30(3), 191–203. <u>https://doi.org/10.1007/s40299021-00553-1</u>
- Linnenbrink, E. A., & Pintrich, P. R. (2002). Motivation as an enabler for academic success. *School Psychology Review*, *31*(3), 313–327.
- Liu, E. Z. F., Lin, C. H., & Chang, C. S. (2010). Student satisfaction and self-efficacy in a cooperative robotics course *Social Behavior and Personality: An International Journal*, 38(8), 1135–1146. http://dx.doi.org/10.2224/sbp.2010.38.8.1135

Liu, J. C. (2019). Evaluating online learning orientation design with a readiness scale. Online

- Learning, 23(4), 42-61. http://dx.doi.org/10.24059/olj.v23i4.2078
- Loeng, S. (2020). Self-directed learning: a core concept in adult education *Education Research International*, 2020, 1-12. https://doi.org/10.1155/2020/3816132
- Long, H. B., & Agyekum, S. K. (1983). Guglielmino's self-directed learning readiness scale: A validation study. *Higher education*, 12(1), 77–87. https://doi.org/10.1007/BF00140273

- Luu, T. M. V. (2022). Readiness for online learning: learners' comfort and self-directed learning ability *International Journal of TESOL & Education*, 2(1), 213–224. <u>https://doi.org/10.54855/ijte.222113</u>
- Macher, D., Papousek, I., Ruggeri, K., & Paechter, M. (2015). Statistics anxiety and performance: Blessings in disguise. *Frontiers in Psychology*, *6*,
- 1116. https://doi.org/10.3389/fpsyg.2015.01116
- Maldonado-Mahauad, J., Pérez-Sanagustín, M., Kizilcec, R. F., Morales, N., & Munoz-Gama, J. (2018). Mining theory-based patterns from big data: Identifying self-regulated learning strategies in massive open online courses. *Computers in Human Behavior*, 80(1), 179– 196. https://doi.org/10.1016/j.chb.2017.11.011
- Manning, G. (2007). Self-directed learning: is a key component of adult learning theory. *Journal of the Washington Institute of China Studies*, *2*(2), 104–115.
- Martin, A. J., Collie, R. J., Durksen, T. L., Burns, E. C., Bostwick, K. C., & Tarbetsky, A. L.
- (2019). Growth goals and growth mindset from a methodological-synergistic perspective: Lessons learned from a quantitative correlational research program. *International Journal of Research & Method in Education*, 42(2), 204–219.

https://doi.org/10.1080/1743727X.2018.1481938

Martin, F., Stamper, B., & Flowers, C. (2020). Examining student perception of readiness for online learning: Importance and confidence. *Online Learning*, 24(2), 38–58.

https://doi.org/10.24059/olj.v24i2.2053

Mascret, N., Elliot, A. J., & Cury, F. (2017). The 3×2 achievement goal questionnaire for teachers *Educational Psychology*, *37*(3), 346–361.

https://doi.org/10.1080/01443410.2015.1096324

Masier, D. J. (2013). An exploratory study of the relationship between self-directed learning and Senge's five disciplines necessary to become a learning organization: in a high-tech company [Doctoral dissertation, North Carolina State University]. NC State Repository. https://repository.lib.ncsu.edu/bitstream/handle/1840.16/8461/etd.pdf?sequence=1&isAll owed=y

- McCollum, D. L., & Kajs, L. T. (2007, July). Examining the relationship between school administrators 'efficacy and goal orientations Allied Academies International Conference: Academy of Educational Leadership Proceedings, USA, 12(2), 31–36.
- McCollum, D., & Kajs, L. (2007, July). A confirmatory factor analytic study of the goal orientation theory of motivation in educational leadership *Allied Academies*

International Conference: Academy of Educational Leadership Proceedings, USA, 12(2), 37–41.

- McGrath, V. (2009). Reviewing the evidence on how adult students learn: an examination of Knowles' model of andragogy *Adult Learner: The Irish Journal of Adult and Community Education*, 99–110.
- Mead, M. S. (2011). The effect of self-directed learning readiness and online course quality ratings on student satisfaction and academic performance in undergraduate eLearning
- (Publication No. 3487458) [Doctoral dissertation, University of Missouri-Kansas City].
- ProQuest Dissertations & Theses Global <u>https://www.proquest.com/dissertationstheses/effect-</u> self-directed-learning-readiness-online/docview/913076924/se-2
- Medaille, A., Beisler, M., Tokarz, R., & Bucy, R. (2022). The Role of Self-Efficacy in the
- Thesis-Writing Experiences of Undergraduate Honors Students *Teaching and Learning Inquiry*, 10. https://doi.org/10.20343/teachlearninqu.10.2
- Meece, J. L., Blumenfeld, P. C., & Hoyle, R. H. (1988). Students' goal orientations and cognitive engagement in classroom activities. Journal of Educational Psychology, 80, 514-523.
- Mega, C., Ronconi, L., & De Beni, R. (2014). What makes a good student? How emotions, selfregulated learning, and motivation contribute to academic achievement. *Journal of Educational Psychology*, *106*(1), 121–131. https://doi.org/10.1037/a0033546
- Meyers, L., Gamst, G., & Guarino, A. (2013). Applied multivariate research: Design and interpretation. Los Angeles, CA: Sage Publications, Inc
- Mezirow, J. (2018). Transformative learning theory In K. Illeris (Ed.), *Contemporary Theories of Learning* (pp. 114–128). Routledge.
- Mezirow, J., & Marsick, V. (1978). *Education for perspective transformation: Women's re-entry programs in community colleges*. Center for Adult Education.
- Midgley, C. (2014). Goals, goal structures, and patterns of adaptive learning Routledge.

- Miller, A. L., Fassett, K. T., & Palmer, D. L. (2021). Achievement goal orientation: A predictor of student engagement in higher education. *Motivation and Emotion*, 45(3), 327–344. https://doi.org/10.1007/s11031-021-09881-7
- Mills, J. D. Raju, D. (2011). Teaching Statistics Online: A Decade's Review of the Literature About What Works *Journal of Statistics Education*, 19(2).

https://doi.org/10.1080/10691898.2011.11889613

Miron-Spektor, E., Vashdi, D. R., & Gopher, H. (2022). Bright sparks and enquiring minds:

Differential effects of goal orientation on the creativity trajectory Journal of Applied

Psychology, 107(2), 310-318. https://doi.org/10.1037/ap10000888

Moeller, A. J., Theiler, J. M., & Wu, C. (2012). Goal setting and student achievement: a longitudinal study *The Modern Language Journal*, *96*(2), 153–169.

https://doi.org/10.1111/j.1540-4781.2011.01231.x

- Mohajan, H. K. (2020). Quantitative research: A successful investigation in natural and social sciences. *Journal of Economic Development, Environment, and People*, 9(4), 50–79. https://doi.org/10.26458/jedep.v9i4.679
- Mohamad Nasri, N., Husnin, H., Mahmud, S. N., & Halim, L. (2020). Mitigating the COVID-19 pandemic: A Snapshot from Malaysia into the Coping Strategies for Pre-Service Teachers' Education *Journal of Education for Teaching*, *46*(4), 546–553. https://doi.org/10.1080/02607476.2020.1802582
- Monroe, K. S. (2016). The relationship between assessment methods and self-directed learning readiness in medical education *International Journal of Medical Education*, 7, 75–80. https://doi.org/10.5116/ijme.56bd.b282
- Morina, N. (2021). Comparisons inform me who I am: A general comparative-processing model of self-perception. *Perspectives on Psychological Science*, 16(6), 1281–1299. https://doi.org/10.1177/1745691620966788
- Morris, D. B., Usher, E. L., & Chen, J. A. (2017). Reconceptualizing the sources of teaching selfefficacy: A critical review of emerging literature *Educational Psychology Review*, 29(4), 795–833. <u>https://doi.org/10.1007/s10648-016-9378-y</u>
- Neter, J., Kutner, M. H., Nachtsheim, C. J., & Wasserman, W. (1996). Applied linear statistical models. Chicago: Irwin.

Nicholls, J. G. (1984). Achievement motivation: Conceptions of ability, subjective experience, task choice, and performance. *Psychological Review*, *91*(3), 328–

346. https://doi.org/10.1037/0033-295X.91.3.328

- Norton, S. M. (2013). *A phenomenological investigation into the self-efficacy beliefs of teachers who have persisted in the teaching profession* Liberty University.
- Nugraha, D. S., Lustyantie, N., & Chaeruman, U. A. (2022). Self-determined learning in EFL classroom: A trajectory for the future research. *Journal on English as a Foreign Language*, *12*(2), 339–361. https://doi.org/10.23971/jefl.v12i2.4068 Ormrod, J. E. (1999). *Human learning*. Merrill.
- Örs, M. (2018). The self-directed learning readiness level of the undergraduate students of midwifery and nursing in terms of sustainability in nursing and midwifery education *Sustainability*, *10*(10), 3574. https://doi.org/10.3390/su10103574
- Ottu, I. F. (2017). Cooperative stakeholding: optimizing students' educational practice through need-centered self-determination, connectedness with the learning environment, and passion *Journal of Education and Practice*, *8*(4), 21–33.
- Pajares, F., & Schunk, D. H. (2001). Self-beliefs and school success: Self-efficacy, self-concept, and school achievement. In R. J. Riding & S. G. Rayner (Eds.), *Self perception* (pp. 239– 265). Ablex Publishing.
- Pajares, F. (2003). Self-Efficacy Beliefs, Motivation, and Achievement in Writing: A Review of the Literature *Reading and Writing Quarterly*, 19, 139-158.

https://doi.org/10.1080/10573560308222

- Pajares, F., Johnson, M., & Usher, E. L. (2007, August). Sources of Writing Self-Efficacy Beliefs of Elementary, Middle, and High School Students *Research in the Teaching of English*, 42(1), 104–120.
- Panisoara IO, Lazar I, Panisoara G, Chirca R, and Ursu AS. Motivation and continuance intention towards online instruction among teachers during the COVID-19 pandemic: the mediating effect of burnout and technostress. Int J Environ Res Public Health. 2020;17(21).

https://doi.org/10.3390/ijerph17218002.

- Pappas, C. (2013, May 9). The Adult Learning Theory Andragogy of Malcolm Knowles. ELearning Industry; eLearning Industry. https://elearningindustry.com/the-adult-learningtheory-andragogy-of-malcolm-knowles
- Paradis, J., Tulpar, Y., & Arppe, A. (2016). Chinese L1 children's English L2 verb morphology over time: Individual variation in long-term outcomes. *Journal of Child Language*, 43(3), 552-500 June 101 (10.1017/20205000015000550)
- 553–580. https://doi.org/10.1017/S0305000915000562
- Partridge, J. A., Knapp, B. A., & Massengale, B. D. (2014). An investigation of motivational variables in CrossFit facilities *The Journal of Strength & Conditioning Research*, 28(6),

1714–1721. https://doi.org/10.1519/JSC.00000000000288

- Perepiczka, M., Chandler, N., & Becerra, M. (2011). Relationship between graduate students' statistics self-efficacy, statistics anxiety, attitude toward statistics, and social support. *The Professional Counselor*, 1(2), 99–108. https://doi.org/10.15241/mpa.1.2.99
- Pham, T. T., Le, H. A., & Do, D. T. (2021). The factors affecting students' online learning outcomes during the COVID-19 pandemic: a Bayesian exploratory factor analysis

Education Research International, 1(1), 2669098. https://doi.org/10.1155/2021/2669098

Pilling-Cormick, J., & Garrison, D. R. (2007). Self-directed and self-regulated learning: conceptual links *Canadian Journal of University Continuing Education*, 33(2), 13–33.

https://doi.org/10.21225/D5S01M

Pintrich, P. R., & Schunk, D. H. (2002). Motivation in Education. Englewood Cliffs, NJ: Prentice Hall.

Polm, J. A., Jr. (2016). *Principal self-efficacy and the teacher principal evaluation project* [Doctoral dissertation, Seattle Pacific University]. Digital Commons.

https://digitalcommons.spu.edu/soe_etd/11

- Preacher, K. J., & Hayes, A. F. (2008). Asymptotic and resampling strategies for assessing and comparing indirect effects in multiple mediator models. Behavior Research Methods, 40, 879–891. doi:10.3758/BRM.40.3.879
- Pulkka, A. T., & Niemivirta, M. (2013). Adult students' achievement goal orientations and evaluations of the learning environment: a person-centered longitudinal analysis *Educational Research and Evaluation*, 19(4), 297–322. https://doi.org/10.1080/13803611.2013.767741

Pumptow, M., & Brahm, T. (2021). Students' digital media self-efficacy and its importance for higher education institutions: development and validation of a survey instrument *Technology, Knowledge, and Learning*, 26(3), 555–575.

https://doi.org/10.1008/s10758-020-09463-5

- Razzaq, S., Shujahat, M., Hussain, S., Nawaz, F., Wang, M., Murad, A., & Tehseen, S. (2019).
 Knowledge management, organizational commitment, and knowledge-worker
 performance: the neglected role of knowledge management in the public sector
 [Knowledge management] *Business Process Management Journal, 25*(5), 923-947.
 https://doi.org/10.1108/BPMJ-03-2018-0079
- Reh, S., Van Quaquebeke, N., Tröster, C., & Giessner, S. R. (2022). When and why does status threat at work bring out the best and the worst in us? A temporal social comparison theory *Organizational Psychology Review*, 12(3), 241–267. https://doi.org/10.1177/20413866221100200
- Reimers, F. M. (2022). Learning from a pandemic. The impact of COVID-19 on education around the world. *Primary and secondary education during Covid-19: Disruptions to educational opportunity during a pandemic*, 1-37.
- Richardson, J. C., & Newby, T. (2006). The role of students' cognitive engagement in online learning. *American Journal of Distance Education*, 20(1), 23-37.
- Ritzhaupt, A. D., Valle, N., & Sommer, M. (2020). Design, development, and evaluation of an online statistics course for educational technology doctoral students: A design and development case. *Journal of Formative Design in Learning*, 4(1), 119–135. https://doi.org/10.1007/s41686-020-00051-5
- Roth, L. (2023). Pathway analysis, causal mediation, and the identification of causal mechanisms *Texts in Quantitative Political Analysis*, 123-151. https://doi.org/10.1007/978-3-031-12982-7 6
- Rousseeuw, P. J., & van Zomeren, B. C. (1990). Unmasking multivariate outliers and leverage points: Rejoinder *Journal of the American Statistical Association*, 85(411), 648. https://doi.org/10.2307/2289999
- Salah Dogham, R., Elcokany, N. M., Saber Ghaly, A., Dawood, T. M., Aldakheel, F. M., Llaguno, M. B., & Mohsen, D. M. (2022). Self-directed learning readiness and online

learning self-efficacy among undergraduate nursing students. *International Journal of Africa Nursing Sciences*, 17, 1-17. https://doi.org/10.1016/j.ijans.2022.100490

- Saeid, N., & Eslaminejad, T. (2017). Relationship between student's self-directed-learning readiness and academic self-efficacy and achievement motivation in students. *International Education Studies*, 10(1), 225–232. http://dx.doi.org/10.5539/ies.v10n1p225
- Sahoo, S. (2016). Finding self-directed learning readiness and fostering self-directed learning through weekly assessment of self-directed learning topics during undergraduate clinical training in ophthalmology. *International Journal of Applied and Basic Medical Research*, 6(3) https://doi.org/10.4103/2229-516X.18695
- Samson Abiodun, T. (2014). The impact of strategic learning orientation, entrepreneurial orientation, and reconfiguring capabilities on the export performance of SMES in Nigeria *International Journal of Management Science and Business Administration*, 2(3), 33–42. https://doi.org/10.18775/ijmsba.1849-5664-5419.2014.23.1004
- Samuel, T. S., & Warner, J. (2021). "I can do math!"Reducing math anxiety and increasing math self-efficacy using a mindfulness and growth mindset-based intervention in first-year students" *Community College Journal of Research and Practice*, 45(3), 205–222. https://doi.org/10.1080/10668926.2019.1666063

Sánchez Rosas, J. (2015). Validation of the achievement goal questionnaire - revised in

Argentinean university students (A-AGQ-R). International Journal of Psychological

Research, 8(1), 10-23. https://doi.org/10.21500/20112084.641

Schumacker, R., and Lomax, R. (2010). A beginner's guide to structural equation modeling (3rd ed.). New York: Taylor and Francis.

Schunk, D. H. (1991). Self-efficacy and academic motivation. *Educational Psychologist, 26*(3-4), 207–231. <u>https://doi.org/10.1207/s15326985ep2603&4_2</u>

Schunk, D. H. (1994). Self-regulation of self-efficacy and attributions in academic settings In D.
H. Schunk & B. J. Zimmerman (Eds.), *Self-regulation of Learning and Performance* (pp. 75–99). Erlbaum.

- Schweder, S., & Raufelder, D. (2021). Examining positive emotions, autonomy support and learning strategies: Self-directed versus teacher-directed learning environments. *Learning Environments Research*. https://doi.org/10.1007/s10984-021-09378-7
- Schweizer, K. (2010). Some guidelines concerning the modeling of traits and abilities in test construction *European Journal of Psychological Assessment*, 26(1), 1–2. https://doi.org/10.1027/1015-5759/a000001

Seaman, J. (2021). The digital learning pulse survey results Cengage.

- https://www.cengagegroup.com/news/press-releases/2022/despite-early-struggles-withdigitallearning-new-research-shows-the-majority-of-community-college-students-wantonlinecourses-going-forward/
- Sekano, K. G. (2020). Enhancing mathematics teachers' self-directed learning through technology-supported cooperative learning [Master's thesis, North-West University] Boloka Institutional Repository. http://hdl.handle.net/10394/35078
- Sharin, A. N. (2021). E-learning during Covid-19: A review of literature. *Jurnal Pengajian Media Malaysia*, 23(1), 15–28. https://doi.org/10.22452/jpmm.vol23no1.2
- Shokar, G. S., Shokar, N. K., Romero, C. M., & Bulik, R. J. (2002). Self-directed learning:
- Looking at outcomes with medical students. Family Medicine, 34(3), 197-200.
- Siddiqui, F. S., Nerali, J. T., & Telang, L. A. (2021). Relationship between the sense of coherence, self-directed learning readiness, and academic performance in Malaysian undergraduate dental students. *Journal of Education and Health Promotion*, 10(1), 105.
- Sidebar, A., & Shankar, s. P. (2022). Goal Orientation As Determinant Of Academic Performance Of Under-Graduate Students. *Journal of Positive School Psychology*, 6(8), 8802-8810. https://journalppw.com/index.php/jpsp/article/view/11369/7349

Siedlecki, S. L. (2020). Understanding descriptive research designs and methods. Clinical Nurse

- Specialist, 34(1), 8-12. https://doi.org/10.1097/NUR.00000000000493
- Simamora, B., & Mutiarawati, E. V. (2021). Achievement goals model validation: Is the 2x2 better than the trichotomous? *International Journal of Evaluation and Research in Education (IJERE)*, 10(1), 142. https://doi.org/10.11591/ijere.v10i1.20869

Siregar, M., Sinar, T. S., Saragih, A., & Lubis, S. (2018). Need analysis for developing translation's textbook based on TEFL pedagogical purpose in Indonesia: English teachers' perspectives. Advances in Language and Literary Studies, 9(3), 81–86.

https://doi.org/10.7575/aiac.alls.v.9n.3p.81

- Skaalvik, E. (1997). Self-enhancing and self-defeating ego orientations: relations with task and avoidance orientation, achievement, self-perceptions, and anxiety. Journal of Educational Psychology, 89, 71-81.
- Sogunro, O. A. (2015). Motivating factors for adult learners in higher education *International Journal of Higher Education*, *4*(1), 22–37.

https://doi.org/https://doi.org/10.5430/ijhe.v4n1p22

- Spector, P. E. (2019). Do not cross me: optimizing the use of cross-sectional designs Journal of
- Business and Psychology, 34(2), 125-137. https://doi.org/10.1007/s10869-018-09613-8
- Standage, M., & Treasure, D. C. (2002). Relationship between achievement goal orientations and multidimensional situational motivation in physical education. *British Journal of*

Educational Psychology, 72(1), 87–103. https://doi.org/10.1348/000709902158784

- Story, D. A., & Tait, A. R. (2019). Survey research. *Anesthesiology*, *130*(2), 192–202. https://doi.org/10.1097/ALN.00000000002436
- Suartama, I. K., Triwahyuni, E., Abbas, S., Hastuti, W. D., Subiyantoro, S., & Salehudin, M. (2020). Development of e-learning oriented inquiry learning based on character education in a multimedia course. *European Journal of Educational Research*, 9(4), 1591–1603.

https://doi.org/10.12973/EU-JER.9.4.1591

- Sumuer, E. (2018). Factors related to college students' self-directed learning with technology. *Australasian Journal of Educational Technology*, *34*(4), 29–43. https://doi.org/10.14742/ajet.3142
- Sun, C., & Clarke-Midura, J. (2022). Testing the efficacy of a near-peer mentoring model for recruiting youth into computer science. *Mentoring & Tutoring: Partnership in Learning*, 30(2), 184–201. https://doi.org/10.1080/13611267.2022.2057101
- Supena, A. (2017). Model pendidikan inklusif untuk siswa tunagrahita di sekolah dasar. *Parameter*, *29*(2), 145–155. https://doi.org/10.21009/parameter.292.03

- Tabachnick, B. G., & Fidell, L. S. (1996). Using Multivariate Statistics (3rd ed.). New York: Harper Collins.
- Tabachnick, B. G., & Fidell, L. S. (2013). Using multivariate statistics (6th ed.). Boston, MA: Pearson.
- Tan, L., & Miksza, P. (2019). Motivational orientations of college band students: A cross-cultural examination of a collective 2 x 2 achievement goal model *Psychology of Music*,

47(1), 33-50. https://doi.org/10.1177/0305735617734628

- Tang, Y. M., Chen, P. C., Law, K. M. Y., Wu, C. H., Lau, Y.-Y., Guan, J., He, D., & Ho, G. T.
- S. (2021). Comparative analysis of student's live online learning readiness during the coronavirus (COVID-19) pandemic in the higher education sector. *Computers & Education*, 168(1), 104–211. http://dx.doi.org/10.1016/j.compedu.2021.104211
- Tarhini, A., Teo, T., & Tarhini, T. (2016). A cross-cultural validity of the e-learning Acceptance measure (EIAM) in Lebanon and England: A confirmatory factor analysis. *Education and Information Technologies*, 21(5), 1269–1282. https://www.learntechlib.org/p/175898/
- Tekkol, I. A., & Demirel, M. (2018). An investigation of self-directed learning skills of undergraduate students. *Frontiers in Psychology*, 9, 1-14. https://doi.org/10.3389/fpsyg.2018.02324
- Teraoka, E., Jancer Ferreira, H., Kirk, D., & Bardid, F. (2021). Affective learning in physical education: A systematic review. *Journal of Teaching in Physical Education*, 40(3), 460– 473. https://doi.org/10.1123/jtpe.2019-0164
- Terry, M. (2006). Self-directed learning by undereducated adults. *Educational Research Quarterly, 29*(4), 28-38.
- Thompson, K. V., & Verdino, J. (2018). An exploratory study of self-efficacy in community college students. *Community College Journal of Research and Practice*, 43(6), 476–479. https://doi.org/10.1080/10668926.2018.1504701
- Toh, W., & Kirschner, D. (2020). Self-directed learning in video games: affordances and pedagogical implications for teaching and learning *Computers & Education*, 154(1),
- 103912. https://doi.org/10.1016/j.compedu.2020.103912

- Torun, E. D. (2019). Online distance learning in higher education: E-learning readiness as a predictor of academic achievement. *Open Praxis*, 12(2), 191. <u>https://doi.org/10.5944/openpraxis.12.2.1092</u>
- Triwahyuni, E., Suartama, I. K., & Barata, M. A. S. (2021, August). The effect of the STEAM strategy on the cognitive and affective learning outcomes of primary school *Seminar Nasional Teknologi Pembelajaran*, 1(1), 457–468.
- Tuominen-Soini, H., Salmela-Aro, K., & Niemivirta, M. (2012). Achievement goal orientations and academic well-being across the transition to upper secondary education. *Learning and Individual Differences*, 22(3), 290–305. https://doi.org/10.1016/j.lindif.2012.01.002
- Tuong, N. V., & Trường, P. N. (2021). Students' self directed learning at University of Social
- Science and Humanities, National University of Ho Chi Minh City Journal of Studies in Education, 11(3), 73. https://doi.org/10.5296/jse.v11i3.18727
- Uchida, A., Michael, R. B., & Mori, K. (2018). An induced successful performance enhances student self-efficacy and boosts academic achievement. *Aera Open*, *4*(4),

2332858418806198. https://doi.org/10.1177/2332858418806198

- Uus, O., Mettis, K., & Väljataga, T. (2022). Self-directed learning: A case study of school students scientific knowledge construction outdoors. *Cogent Education*, 9(1), 1-11. https://doi.org/10.1080/2331186x.2022.2074342
- Uztemur, S. (2020). Achievement goals and learning approaches in the context of social studies teaching: a comparative analysis of 3x2 and 2x2 models *Participatory Educational Research*, 7(2), 1–18. <u>https://doi.org/10.17275/per.20.16.7.2</u>
- Vanthournout G., Coertjens, L. Gijbels. D, V. Donche, P. Van PetegemAssessing students' development in learning approaches according to initial learning profiles: a personoriented perspective Studies in Educational Evaluation, 39 (1) (2013), pp. 33-
- 40, <u>10.1016/j.stueduc.2012.08.002</u>
- Verhulst, B., Eaves, L. J., & Hatemi, P. K. (2012). Correlation, not causation: the relationship between personality traits and political ideologies *American Journal of Political Science*, 56(1), 34–51. https://doi.org/10.1111/j.1540-5907.2011.00568.x

- Walker, C. O., Greene, B. A., & Mansell, R. A. (2006). Identification with academics, intrinsic/extrinsic motivation, and self-efficacy as predictors of cognitive engagement. *Learning and Individual Differences*, 16(1), 1– 12. https://doi.org/10.1016/j.lindif.2005.06.00
- Walker, G. K., Jalukar, S., & Brake, J. (2017). Effect of refined functional carbohydrates from enzymatically hydrolyzed yeast on the presence of Salmonella spp. in the ceca of broiler breeder females. *Poultry Science*, 96(8), 2684–2690. https://doi.org/10.3382/ps/pex054
- Wang, C. H., Shannon, D. M., & Ross, M. E. (2013). Students' characteristics, self-regulated learning, technology self-efficacy, and course outcomes in online learning. *Distance Education*, 34(3), 302–323. https://doi.org/10.1080/01587919.2013.835779 \
- Wang, W., Song, S., Chen, X., & Yuan, W. (2021). When learning goal orientation leads to learning from failure: the roles of negative emotion coping orientation and positive grieving *Frontiers in Psychology*, 12, 1-17. https://doi.org/10.3389/fpsyg.2021.608256
- Wang, H., Xu, M., Xie, X., Dong, Y., & Wang, W. (2021). Relationships between achievement goal orientations, learning engagement, and academic adjustment in first-year students:
 Variable-centered and person-centered approaches. *Frontiers in Psychology*, *12*, 767886. https://doi.org/10.3389/fpsyg.2021.767886
- Watson, S. L., Watson, W. R., Yu, J. H., Alamri, H., & Mueller, C. (2017). Learner profiles of attitudinal learning in a MOOC: An explanatory sequential mixed methods study. *Computers & Education*, 114(1), 274–285. https://doi.org/10.1016/j.compedu.2017.07
- Wehrens, R., Putter, H., & Buydens, L. M. (2000). The bootstrap: a tutorial. *Chemometrics and intelligent laboratory systems*, 54(1), 35-52.
- Wei, H. C., & Chou, C. (2020). Online learning performance and satisfaction: Do perceptions and readiness matter? *Distance Education*, 41(1), 48–69. https://doi.org/10.1080/01587919.2020.1724768
- Wei, X., Saab, N., & Admiraal, W. (2021). Assessment of cognitive, behavioral, and affective learning outcomes in massive open online courses: a systematic literature review *Computers & Education*, 163, 104097.
- https://doi.org/10.1016/j.compedu.2020.104097

- Welter, V. D. E., Becker, L. B., & Großschedl, J. (2022). Helping learners become their own teachers: The beneficial impact of trained concept-mapping-strategy use on metacognitive regulation in learning *Education Sciences*, 12(5), 325. https://doi.org/10.3390/educsci12050325
- Widyastuti, M., Simanjuntak, A. G. F., Hartama, D., Windarto, A. P., & Wanto, A. (2019, August). Classification model C. 45 on determining the quality of customer service in bank BTN Pematangsiantar branch. In A. R. Putra, D. Hartama, A. P. Windarto, A. Wanto, S. Sembiring, & T. Herawan (Eds.), *Journal of Physics: Conference Series: Vol.*
- 1255 (Nr. 012002). IOP Publishing. <u>https://doi.org/10.1088/1742-6596/1255/1/012002</u> Wiley K. (1983). Effects of a self-directed learning project and preference for structure on self-directed learning readiness *Nursing Research*, *32*(3), 181–185.
- Williams, T., & Williams, K. (2010). Self-efficacy and performance in mathematics: Reciprocal determinism in 33 nations. *Journal of Educational Psychology*, *102*(2), 453–466. https://doi.org/10.1037/a0017271
- Wolters, C. A. (2004). Advancing achievement goal theory: using goal structures and goal orientations to predict students' motivation, cognition, and achievement *Journal of Educational Psychology*, 96(2), 236–250. https://doi.org/10.1037/0022-0663.96.2.236
- Wong, J., Baars, M., He, M., de Koning, B. B., & Paas, F. (2021). Facilitating goal setting and planning to enhance online self-regulation of learning. *Computers in Human Behavior*, 124, 106913.
- Wood, J. (2022). These 3 charts show the global growth of online learning. World Economic Forum. https://www.weforum.org/agenda/2022/01/online-learning-courses-reskillskillsgap/
- Wu, C., Jing, B., Gong, X., Mou, Y., & Li, J. (2021). Students' learning strategies and academic emotions: their influence on learning satisfaction during the COVID-19 pandemic *Frontiers in Psychology*, 12. https://doi.org/10.3389/fpsyg.2021.717683
- Xu, W., Shen, Z.-Y., Lin, S.-J., & Chen, J.-C. (2022). Improving the behavioral intention of continuous online learning among learners in higher education during COVID-19. *Frontiers in Psychology*, 13. https://doi.org/10.3389/fpsyg.2022.857709

- Yan, S. (2012). Teachers' roles in autonomous learning. *Journal of Sociological Research*, 3(2), 557–562. <u>https://doi.org/10.5296/jsr.v3i2.2860</u>
- Yang G. F. (2016). Factors affecting the continued use of MOOC user behavior *Open Educ. Res.* 1 100–111. 10.13966/j.cnki.kfjyyj.2016.01.012
- Yao, J. J. (2021). The significance of self-directed learning readiness, academic self-efficacy, and problem-solving ability among Filipino nursing students *International Journal of Learning, Teaching, and Educational Research*, 20(10), 83–94.
- https://doi.org/10.26803/ijlter.20.10.5
- Yeh, Y. C., Kwok, O. M., Chien, H. Y., Sweany, N. W., Baek, E., & McIntosh, W. A. (2019). How college students' achievement goal orientations predict their expected online learning outcome: the mediation roles of self-regulated learning strategies and supportive online learning behaviors *Online Learning*, 23(4), 23–41. https://doi.org/10.24059/olj.v23i4.2076
- Yilmaz, R. (2017). Exploring the role of e-learning readiness on student satisfaction and motivation in a flipped classroom *Computers in Human Behavior*, *70*(1), 251–260.
- https://doi.org/10.1016/j.chb.2016.12.085
- Yokoyama, S. (2019). Academic self-efficacy and academic performance in online learning: A mini review. *Frontiers in Psychology*, 9, 2794. <u>https://doi.org/10.3389/fpsyg.2018.02794</u>
- Yoo, J.E. Structural relationship among self-directed learning ability, learner-instructor interaction, learner-learner interaction, and class satisfaction in online learning environments. J. Christ. Educ. Korea 2020, 63, 255–281.
- York, R. (2018). Control variables and causal inference: A question of balance. *International Journal of Social Research Methodology*, 21(6), 675-684. https://doi.org/10.1080/13645579.2018.1468730
- Young, T. J. (2015). Questionnaires and surveys In. Z. Hua (Ed.), Research methods in intercultural communication: A practical guide (pp. 163–180). https://doi.org/10.1002/9781119166283.ch11
- Yousafzai, B. K., Hayat, M., & Afzal, S. (2020). Application of machine learning and data mining in predicting the performance of intermediate and secondary education students.

Education and Information Technologies, *25*(6), 4677–4697. https://doi.org/10.1007/s10639-020-10189-1

- Zeldin, A. L., & Pajares, F. (2000). Against the odds: Self-efficacy beliefs of women in mathematical, scientific, and technological careers. *American Educational Research Journal*, 37(1), 215–246. <u>https://doi.org/10.2307/1163477</u>
- Zimmerman, B. J., Bandura, A., & Martinez-Pons, M. (1992). Self-motivation for academic attainment: The role of self-efficacy beliefs and personal goal setting. *American Educational Research Journal*, 29(3), 663–676. https://doi.org/10.2307/1163261
- Zimmerman, B. J. (2000). Attaining self-regulation: A social cognitive perspective. In M. Boekaerts, P. R. Pintrich, & M. Zeidner (Eds.), *Handbook of self-regulation* (pp.

13-39).Academic Press. https://doi.org/10.1016/B978-012109890-2/50031-7

Appendix A

The Online Survey

11/10/2020

Qualtrics Survey Software

The Auburn University Institutional Review Board has approved this Document for use from 11/05/2020 to _____ Protocol # _____20-535 EX 2011

Default Question Block

INFORMATION LETTER for a Research Study entitled

"Relations of Online readiness, Academic procrastination, Achievement goal theory, learning interaction and online statistics learning outcomes"

You are invited to participate in a research study to investigate relations of online readiness, academic procrastination, statistical self-efficacy, goal-approach orientation, learning interaction and learning outcomes in online statistics learning settings. The study is being conducted by Sangah (Sunny) Lee, and Hyeon Jean Yoo, Ph.D student in Department of Educational Foundations, Leadership and Technology at Auburn University under the direction of Dr. Chihhsuan Wang, a professor of the Department of Educational Foundations, Leadership, and Technology at Auburn University. You are invited to participate because you over 18 years old and who have been taking statistics course(s) offered from Department of Educational Foundations, Leadership and/or Technology and Department of Mathematics and Statistics at Auburn University in Fall semester 2020.

What will be involved if you participate? Your participation is completely voluntary. If you decide to participate in this research study, you will be asked to click the "NEXT" button if you consent to participate in the survey. Your total time commitment will be approximately 20 minutes.

Are there any risks or discomforts? There is no known risk or discomfort associated with this survey. The survey questions focus on your learning experiences in taking online statistics courses. 11/10/2020

Qualtrics Survey Software

Will you receive compensation for participating? To thank you for your time you will be entered into a random drawing to win one of the ten \$30 Amazon e-Gift Card at the end of the survey. You will be asked to provide your information if you wish to be entered in a drawing to receive a gift certificate. Your information will not be associated with your survey responses.

If you change your mind about participating, you can withdraw at any time by closing your browser. If you choose to withdraw, your data can be withdrawn as long as it is identifiable. Once you've submitted anonymous data, it cannot be withdrawn since it will be unidentifiable. Your decision about whether or not to participate or to stop participating will not jeopardize your future relations with Auburn University, the Department of Educational Foundations, Leadership, and Technology.

Any data obtained in connection with this study will remain anonymous. We will protect your privacy and the data you provide by password protected only via Horizon Auburn VPN software. Information collected through your participation may be published in a professional journal, and/or presented at a professional conference.

If you have questions about this study, please contact Sangah (Sunny) Lee at szl0146@auburn.edu, Hyeon Jean Yoo at hjy0002@auburn.edu or Dr. Chih-hsuan Wang at wangchi@auburn.edu.

If you have questions about your rights as a research participant, you may contact the Auburn University Office of Research Compliance or the Institutional Review Board by phone (334) 844-5966 or e-mail at IRBadmin@auburn.edu or <u>IRBChair@auburn.edu</u>.

Demographics

What is your gender?

The Auburn University Institutional Review Board has approved this Document for use from 11/05/2020 to Protocol # 20-535 EX 2011

O Male

https://auburn.ca1.qualtrics.com/Q/EditSection/Blocks/Ajax/GetSurveyPrintPreview?ContextSurveyID=SV_eRV9rIOHqmcsDS8&ContextLibraryID=U... 2/22

| /10/2020 | Qualtrics Survey Software |
|-------------------------------------|---|
| O Female | The Auburn University Institutional Review Board has approved this |
| O Prefer not to say | Document for use from 11/05/2020 to Protocol # 20-535 EX 2011 |
| What is your age? | |
| | |
| What is your ethnicity? | |
| O Asian | |
| O African-American | |
| O White/Caucasian | |
| O Hispanic/Latino | |
| O Hawaiian/Pacific Islander | |
| O Other. Please specify | |
| Are you an international student or | r domestic student? |
| O International student | |
| O Domestic student | |
| Are you currently taking the online | course in the States or in your home country? |
| O in the United States | |
| O in my home country | |
| Have you taken online courses be | fore? |

O Yes O No

What is your major area? (STEM-related major indicates that the majors in science, technology, engineering and mathematics, including agriculture, medical and health majors)

 $https://auburn.ca1.qualtrics.com/Q/EditSection/Blocks/Ajax/GetSurveyPrintPreview?ContextSurveyID=SV_cRV9rfOHqmcsDSB&ContextLibraryID=U... 3/22$

| unazuzu quainos survey somware | | | | |
|--------------------------------|------------------------|-------------------------|------------------------------|---------------------------|
| O STEM | | | | |
| O Non-STEM | | | | |
| What is your major | ? | | | |
| | | | | |
| Which degree leve | I of school are you | registered in Fall 2 | 2020? | |
| O Undergraduate | | | | |
| O Master's | | | | |
| O PhD | | | | |
| Which Department | t did you take statis | stics course from in | Fall 2020? | |
| O EFLT (Departme | nt of Educational Fou | indations, Leadership | and Technology; e.g | . ERMAXXXX) |
| O MATH/STAT (De | partment of Mathema | tics and Statistics; e. | g. STATXXXX) | |
| Think about ONE of | of the statistical cou | urses you are takin | g <mark>in</mark> Fall 2020. | |
| What is the Course | e Number (e.g., ST | AT XXXX or ERMA | XXXX)? | |
| | | | | |
| What is your expe | cted final grade of t | the course you spe | cified above? (Bas | ed on the grade |
| from class activitie | s, participation, Mid | dterm or homework | , and so on.) | |
| Ô | BO | c | Ō | FO |
| Online Readiness | | | | |
| The following state | ements will help us | understand your o | nline learning rea | diness . Usina the |
| scale below (from | strongly disagree to | o strongly agree), p | lease indicate to w | hat extent each o |
| | a a man da ta un | | | |

https://auburn.ca1.qualtrics.com/Q/EditSection/Blocks/Ajax/GetSurveyPrintPreview?ContextSurveyID=SV_cRV9rfOHqmcsDS8&ContextLibraryID=U... 4/22
11/10/2020

Qualtrics Survey Software

All of your responses will be kept anonymous and confidential. There are no right or wrong responses, so please be open and honest.

| | Strongly disagree | Somewhat disagree | Neither agree nor disagree | Somewhat agree | Strongly agree |
|---|----------------------|----------------------|-------------------------------|-------------------|-------------------|
| I feel confident in performing basic functions of Microsoft Office programs (MSWord, MS Excel, MS Power Point) | 0 | 0 | 0 | 0 | 0 |
| I feel confident in my knowledge and skills of how to manage software for online learning | 0 | 0 | 0 | 0 | 0 |
| I feel confident in using the Internet to find information | 0 | 0 | 0 | 0 | 0 |
| l carry out my own study plan while learning online | 0 | 0 | 0 | 0 | 0 |
| I seek assistance when facing learning problems | 0 | 0 | 0 | 0 | 0 |
| I manage my time well while learning online | 0 | 0 | 0 | 0 | 0 |
| l set up my online learning goals | 0 | 0 | 0 | 0 | 0 |
| I have a high expectation for my learning performance | 0 | 0 | 0 | 0 | 0 |
| I can direct my own learning progress while learning online | 0 | 0 | 0 | 0 | 0 |
| I am not distracted by other online activities (WhatsApp, Insta, FB) while learning online | 0 | 0 | 0 | 0 | 0 |
| I repeated the online learning materials based on my needs | 0 | 0 | 0 | 0 | 0 |
| I am open to new ideas when learning online | 0 | 0 | 0 | 0 | 0 |
| I have the motivation to do online learning | 0 | 0 | 0 | 0 | 0 |

 $https://auburn.ca1.qualtrics.com/Q/EditSection/Blocks/Ajax/GetSurveyPrintPreview?ContextSurveyID=SV_cRV9rfOHqmcsDSB&ContextLibraryID=U... 5/22$

| 11/10/2020 | Qualtrics Survey Software | | | | | | | |
|---|---------------------------|----------------------|-------------------------------|----------------|----------------|--|--|--|
| | Strongly disagree | Somewhat disagree | Neither agree nor disagree | Somewhat agree | Strongly agree | | | |
| While learning online, I improve from my previous mistakes | 0 | 0 | 0 | 0 | 0 | | | |
| I like to share my ideas with ideas others while learning online | 0 | 0 | 0 | 0 | 0 | | | |
| I feel confident in using online tools to communicate with others | 0 | 0 | 0 | 0 | 0 | | | |
| I express my thoughts through online text messages/ posting comments | 0 | 0 | 0 | 0 | 0 | | | |
| I post questions in online discussion | 0 | 0 | 0 | 0 | 0 | | | |

Procrastination

The following questions assess your **habits and routines** as a student. Please answer the following as they apply to yourself. how much do you, yourself agree to the following statement?

There are no right or wrong responses, so please be open and honest.

| | Strongly agree | Somewhat agree | Neither agree nor disagree | Somewhat disagree | Strongly disagree |
|--|-------------------|-------------------|-------------------------------|----------------------|----------------------|
| I usually allocate time to review and proofread my work. | 0 | 0 | 0 | 0 | 0 |
| I put off projects until the last minute. | 0 | 0 | 0 | 0 | 0 |
| I have found myself waiting until the day before to start a big project. | 0 | 0 | 0 | 0 | 0 |
| I know I should work on schoolwork, but I just don't do it. | 0 | 0 | 0 | 0 | 0 |

https://auburn.ca1.qualtrics.com/Q/EditSection/Blocks/Ajax/GetSurveyPrintPreview?ContextSurveyID=SV_cRV9rfOHqmcsDSB&ContextLibraryID=U...6/22

| 11/10/2020 | Qualtrics Survey Software | | | | | | | | |
|---|---------------------------|----------------|-------------------------------|----------------------|----------------------|--|--|--|--|
| | Strongly agree | Somewhat agree | Neither agree nor disagree | Somewhat disagree | Strongly disagree | | | | |
| When working on schoolwork, I usually get distracted by other things. | 0 | 0 | 0 | 0 | 0 | | | | |
| I waste a lot of time on unimportant things. | 0 | 0 | 0 | 0 | 0 | | | | |
| I get distracted by other, more fun, things when I am supposed to work on schoolwork. | 0 | 0 | 0 | 0 | 0 | | | | |
| I concentrate on schoolwork instead of other distractions. | 0 | 0 | 0 | 0 | 0 | | | | |
| I can't focus on schoolwork or projects for more than an hour until I get distracted. | 0 | 0 | 0 | 0 | 0 | | | | |
| My attention span for schoolwork is very short. | 0 | 0 | 0 | 0 | 0 | | | | |
| Tests are meant to be studied for just the night before. | 0 | 0 | 0 | 0 | 0 | | | | |
| 12. I feel prepared well in advance for most tests. | 0 | 0 | 0 | 0 | 0 | | | | |
| 13. "Cramming" and last-minute studying is the best way that I study for a big test. | 0 | 0 | 0 | 0 | 0 | | | | |
| 14. I allocate time so I don't have to "cram" at the end of the semester. | 0 | 0 | 0 | 0 | 0 | | | | |
| I only study the night before exams. | 0 | 0 | 0 | 0 | 0 | | | | |
| 16. If an assignment is due at midnight, I will work on it until 11:59. | 0 | 0 | 0 | 0 | 0 | | | | |
| 17. When given an assignment, I usually put it away and forget about it until it is almost due. | 0 | 0 | 0 | 0 | 0 | | | | |

https://auburn.ca1.qualtrics.com/Q/EditSection/Blocks/Ajax/GetSurveyPrintPreview?ContextSurveyID=SV_cRV9rfOHqmcsDSB&ContextLibraryID=U... 7/22

| 11/10/2020 | Qualtrics Survey Software | | | | | | | | |
|--|---------------------------|----------------|-------------------------------|----------------------|----------------------|--|--|--|--|
| | Strongly agree | Somewhat agree | Neither agree nor disagree | Somewhat disagree | Strongly disagree | | | | |
| Friends usually distract me from schoolwork. | 0 | 0 | 0 | 0 | 0 | | | | |
| 19. I find myself talking to friends or family instead of working on school work. | 0 | 0 | 0 | 0 | 0 | | | | |
| 20. On the weekends, I make plans to do homework and projects, but I get distracted and hang out with friends. | 0 | 0 | 0 | 0 | 0 | | | | |
| 21. I tend to put off things for the next day. | 0 | 0 | 0 | 0 | 0 | | | | |
| 22. I don't spend much time studying school material until the end of the semester. | 0 | 0 | 0 | 0 | 0 | | | | |
| 23. I frequently find myself putting important deadlines off. | 0 | 0 | 0 | 0 | 0 | | | | |
| 24. If I don't understand something, I'll usually wait until the night before a test to figure it out | 0 | 0 | 0 | 0 | 0 | | | | |
| 25. I read the textbook and look over notes before coming to class and listening to a lecture or teacher | 0 | 0 | 0 | 0 | 0 | | | | |

Achievement Goal

The following statements represent **types of goals** that you may or may not have for this class.

Using the scale to the right of each statement, select the answer that best describes you. All of your responses will be kept anonymous and confidential. There are no right or wrong responses, so please be open and honest.

If you think the statement is very true of you, select 7; if a statement is not at all true of you, select 1. If the statement is more or less true of you, find the number between 1 and 7 that

https://auburn.ca1.qualtrics.com/Q/EditSection/Blocks/Ajax/GetSurveyPrintPreview?ContextSurveyID=SV_cRV9rfOHqmcsDSB&ContextLibraryID=U... 8/22

11/10/2020

Qualtrics Survey Software

best describes you.

| | Not at All TRUE of Me 1 | 2 | 3 | 4 | 5 | 6 | Very TRUE of Me 7 |
|--|-------------------------------|---|---|---|---|---|-------------------------|
| It is important to me to perform as well as I possibly can. | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| I want to perform as well as it is possible for me to perform. | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| It is important for me to master all aspects of my performance. | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| I worry that I may not perform as well as I possibly can. | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| Sometimes I'm afraid that I may not perform as well as I'd like. | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| I'm often concerned that I may not perform as well as I can perform. | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| It is important to me to do well compared to others. | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| It is important for me to perform better than others. | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| My goal is to do better than most other performers. | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| I just want to avoid performing worse than others. | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| My goal is to avoid performing worse than everyone else. | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| It is important for me to avoid being one of the worst performers in the group. | 0 | 0 | 0 | 0 | 0 | 0 | 0 |

Stat SE

https://auburn.ca1.qualtrics.com/Q/EditSection/Blocks/Ajax/GetSurveyPrintPreview?ContextSurveyID=SV_cRV9rfOHqmcsDSB&ContextLibraryID=U... 9/22

Qualtrics Survey Software

11/10/2020

The following statement relate to your **confidence** in completing the following tasks successfully. Please answer the following as they apply to yourself.

The item scale has 6 possible responses: $(1) = no \text{ confidence et all, } (2) = a \text{ little confidence, } (3) = a fair amount of confidence, } (4) much confidence, } (5) = very much confidence, } (6) = complete confidence.$

| | No confidence et all (1) | A little confidence (2) | A fair amount of confidence (3) | Much confidence (4) | Very much confidence (5) | Complete confidence (6) |
|---|--------------------------------|-------------------------------|--|---------------------------|--------------------------------|-------------------------------|
| 1.Identify the scale of measurement for a variable. | 0 | 0 | 0 | 0 | 0 | 0 |
| 2.Interpret the probability value(p- value) from a statistical procedure. | 0 | 0 | 0 | 0 | 0 | 0 |
| 3.Identify if a distribution is skewed when given the values of three measures of central tendency. | 0 | 0 | 0 | 0 | 0 | 0 |
| 4.Select the correct statistical procedure to be used to answer a research question. | 0 | 0 | 0 | 0 | 0 | 0 |
| 5.Interpret the results of a statistical procedure in terms of the research question. | 0 | 0 | 0 | 0 | 0 | 0 |
| 6.Identify the factors that influence power. | 0 | 0 | 0 | 0 | 0 | 0 |
| 7.Explain what the value of the standard deviation means in terms of the variable being measured. | 0 | 0 | 0 | 0 | 0 | 0 |

The following statement relate to your **confidence** in completing the following tasks successfully. Please answer the following as they apply to yourself.

https://auburn.ca1.qualtrics.com/Q/EditSection/Blocks/Ajax/GetSurveyPrintPreview?ContextSurveyID=SV_cRV9rfOHqmcsDSB&ContextLibraryID=... 10/22

11/10/2020

Qualtrics Survey Software

The item scale has 6 possible responses: (1) = no confidence et all, (2) = a little confidence, (3) = a fair amount of confidence, (4) much confidence, (5) = very much confidence, (6) = complete confidence.

| | No confidence et all (1) | A little confidence (2) | A fair amount of confidence (3) | Much confidence (4) | Very much confidence (5) | Complete confidence (6) |
|---|--------------------------------|-------------------------------|--|---------------------------|--------------------------------|-------------------------------|
| 8.Distinguish between a Type 1 error and a Type 2 error in hypothesis testing. | 0 | 0 | 0 | 0 | 0 | 0 |
| 9.Explain what the numeric value of the standard error is measuring. | 0 | 0 | 0 | 0 | 0 | 0 |
| 10.Distinguish between the objectives of descriptive versus inferential statistical procedures. | 0 | 0 | 0 | 0 | 0 | 0 |
| 11.Distinguish between the information given by the three measures of central tendency. | 0 | 0 | 0 | 0 | 0 | 0 |
| 12.Distinguish between a population parameter and a sample statistic. | 0 | 0 | 0 | 0 | 0 | 0 |
| 13.Identify when the mean, median, and mode should be used as a measure of central tendency. | 0 | 0 | 0 | 0 | 0 | 0 |
| 14.Explain the difference between a sampling distribution and a population distribution. | 0 | 0 | 0 | 0 | 0 | 0 |

Interaction

The following statements will help us understand the **type of interaction** you had in your class.

Using the scale below, indicate to what extent each of the following items presently corresponds to one of the interaction you had in your class.

https://auburn.ca1.qualtrics.com/Q/EditSection/Blocks/Ajax/GetSurveyPrintPreview?ContextSurveyID=SV_cRV9rfOHqmcsDSB&ContextLibraryID=... 11/22

| 11/10/2020 | Qualtrics Survey Software | | | | | | | | |
|--|---------------------------|---|---|---|---|---|---|---|--------------------------|
| | No confidence 1 | 2 | 3 | 4 | 5 | 6 | 7 | 8 | Total Confidence 9 |
| 30. Selecting reliable and valid instruments | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| 31. Writing statistical computer programs | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| 32. Getting money to help pay for research | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 |
| Operationalizing variables of interest | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 |

Satisfaction

The following questions relates to your **satisfaction** with the course you had. Using the scale to the right of each statement, select the answer that best describes you.

There is no right or wrong answer, just respond as best you can.

| | Strongly disagree | Somewhat disagree | Neither agree nor disagree | Somewhat agree | Strongly agree |
|---|----------------------|----------------------|-------------------------------|-------------------|-------------------|
| Overall, I am satisfied with this class. | 0 | 0 | 0 | 0 | 0 |
| This course contributed to my educational development. | 0 | 0 | 0 | 0 | 0 |
| This course contributed to my professional development. | 0 | 0 | 0 | 0 | 0 |
| I am satisfied with the level of interaction that happened in this course. | 0 | 0 | 0 | 0 | 0 |
| In the future, I would be willing to take a fully online course again. | 0 | 0 | 0 | 0 | 0 |

Gift

Do you want to be entered in a drawing to receive a gift certificate?

https://auburn.ca1.qualtrics.com/Q/EditSection/Biocks/Ajax/GetSurveyPrintPreview?ContextSurveyID=SV_cRV9rfOHqmcsDSB&ContextLibraryID=... 21/22

Appendix B

Information Letter of the Online Survey for the Study



EDUCATIONAL FOUNDATIONS, LEADERSHIP AND TECHNOLOGY

(NOTE: DO NOT AGREE TO PARTICIPATE UNLESS IRB APPROVAL INFORMATION WITH CURRENT DATES HAS BEEN ADDED TO THIS DOCUMENT.)

INFORMATION LETTER for a Research Study entitled

"Relations of Online readiness, Academic procrastination, Achievement goal theory, learning interaction and online statistics learning outcomes"

You are invited to participate in a research study to investigate relations of online readiness, procrastination, learning factors and learning outcomes in online statistics learning settings. The study is being conducted by Sangah (Sunny) Lee, and Hyeon Jean Yoo, Ph.D students in Department of Educational Foundations, Leadership and Technology at Auburn University under the direction of Dr. Chih-hsuan Wang, a professor of the Department of Educational Foundations, Leadership, and Technology at Auburn University. You are invited to participate because you over 18 years old and who have been taking statistics course(s) offered from Department of Educational Foundations, Leadership and Department of Mathematics and Statistics at Auburn University in Fall semester 2020.

What will be involved if you participate? Your participation is completely voluntary. If you decide to participate in this research study, you will be asked to take an anonymous online survey through Qualtrics and click "NEXT" button if you consent to participate in the survey. Your total time commitment will be approximately 20 minutes.

Are there any risks or discomforts? There is no known risk or discomfort associated with this survey. The survey questions focus on your learning experiences in taking online statistics courses.

Will you receive compensation for participating? To thank you for your time you will be entered into a random drawing to win one of the ten \$30 Amazon e-Gift Card at the end of the survey. You will be asked to provide your information if you wish to be entered in a drawing to receive a gift certificate. Your information will not be associated with your survey responses.

1 of

Version Date (date document created): 11/09/2020

The Auburn University Institutional Review Board has approved this Document for use from 11/05/2020 to Protocol # 20-535 EX 2011 If you change your mind about participating, you can withdraw at any time by closing your browser. Once you've submitted anonymous data, it cannot be withdrawn since it will be unidentifiable. Your decision about whether or not to participate or to stop participating will not jeopardize your future relations with Auburn University, the Department of Educational Foundations, Leadership, and Technology.

Any data obtained in connection with this study will remain anonymous. We will protect your privacy and the data you provide by password protected only via Horizon Auburn VPN software. Information collected through your participation may be published in a professional journal, and/or presented at a professional conference.

If you have questions about this study, please contact Sangah (Sunny) Lee at szl0146@auburn.edu, Hyeon Jean Yoo at <u>hjy0002@auburn.edu</u> or Dr. Chihhsuan Wang at <u>wangchi@auburn.edu</u>.

If you have questions about your rights as a research participant, you may contact the Auburn University Office of Research Compliance or the Institutional Review Board by phone (334) 844-5966 or e-mail at IRBadmin@auburn.edu or IRBChair@auburn.edu.

HAVING READ THE INFORMATION ABOVE, YOU MUST DECIDE IF YOU WANT TO PARTICIPATE IN THIS RESEARCH PROJECT. IF YOU DECIDE TO PARTICIPATE, PLEASE CLICK ON THE LINK BELOW. YOU MAY PRINT A COPY OF THIS LETTER TO KEEP.

Sangah (Sunny) Lee 11/16/2020 Investigator Date

Chih-hsuan Wang 11/16/2020 Faculty Advisor Date

Hyeon Jean Yoo 11/16/2020 Co-Investigator Date The Auburn University Institutional Review Board has approved this Document for use from <u>11/05/2020</u> to ______ Protocol # _____20-535 EX 2011

The Auburn University Institutional Review Board has approved this document for use from ______ to _____. Protocol #_____

https://auburn.gualtrics.com/jfe/form/SV_cRV9rfOHgmcsDSB

Version Date (date document created): ___11/09/2020__

2 of 2

Appendix C

Initial E-mail Invitation

E-MAIL INVITATION FOR ON-LINE SURVEY

Dear valued students,

My name is Sangh (Sunny) Lee and I am a graduate student in the Department of Educational Foundations, Leadership and Technology and Department of Mathematics and Statistics at Auburn University. I would like to invite you to participate in my research study to examine the "Relations of Online readiness, Academic procrastination, Achievement goal theory, learning interaction and online statistics learning outcomes." Because you are over 18 years old and who have been taking statistics course(s) offered from Department of Educational Foundations, Leadership and Technology and Department of Mathematics and Statistics at Auburn University in Fall semester 2020, I am inviting you to participate in this research study by completing the attached surveys.

In this study, you will be asked to complete an electronic survey. Your participation in this study is completely voluntary and you are free to withdraw your participation from this study at any time. The following questionnaire will require approximately 20 minutes to complete. Participants may win \$30 Amazon Gift Card and those who entirely complete the survey would be part of the draw to win the gift card.

This survey has been approved by the Institutional Review Board of Auburn University. There are no risks associated with participating in this study. The survey collects no identifying information of any respondent. All of the response in the survey will be recorded anonymously.

If you decide to participate in this research study, please click the link below to go the survey and then you will be asked to click "NEXT" button if you consent to participate in the survey.

https://auburn.gualtrics.com/jfe/form/SV_cRV9rfOHqmcsDSB

Thank you for taking the time to assist me in my educational endeavors. The data collected will provide useful information regarding. If you would like a summary copy of this study, please complete and detach the Request for Information Form and return it to me in a separate envelope. Completion and return of the questionnaire will indicate your willingness to participate in this study. If you require additional information or have questions, please contact me at the email address listed below.

If you would like to know more information about this study, an information letter is attached to this email. If you decide to participate after reading the letter, you can access the survey from a link below and

If you have any questions, please contact me at <u>szl0146@aburun.edu</u> or my advisor, Dr. Chih-hsuan Wang, at <u>wangchi@auburn.edu</u>.

Thank you for your consideration, /sign/

> The Auburn University Institutional Review Board has approved this Document for use from <u>11/05/2020</u> to ______ Protocol # 20-535 EX 2011

Appendix D

Approved Email from Office of Research Compliance of Auburn University

5/10/23, 3:10 PM

Mail - Sunny Lee - Outlook

Lee Approval Exempt Protocol #20-535 EX 2011, "Relations of Online Readiness, Academic Procrastination, Achievement Goal Theory, Learning Interaction, Self-Efficacy and Online Statistics Learning Outcomes"

IRB Administration <irbadmin@auburn.edu>

Wed 12/2/2020 8:13 AM

To: Sunny Lee <sz 0146@auburn.edu>

Cc: Chih-hsuan Wang <wangchi@auburn.edu>;James Satterfield <jws0089@auburn.edu>

2 attachments (4 MB)

Investigators Responsibilities rev 1-2011.docx; Lee 20-535 EX 2011 Revisions 2.pdf;

Use <u>IRBsubmit@auburn.edu</u> for protocol related submissions and <u>IRBadmin@auburn.edu</u> for questions and information. The IRB only accepts forms posted at <u>https://cws.auburn.edu/ypr/compliance/humansubjects/?Forms</u> and submitted electronically.

Dear Sangah,

Your protocol titled "Relations of Online Readiness, Academic Procrastination, Achievement Goal Theory, Learning Interaction, Self-Efficacy and Online Statistics Learning Outcomes" was approved by the AU IRB as "Exempt" under federal regulation 45 CFR 46.101(b)(1,2).

Official notice:

This e-mail serves as notice the protocol has been approved. By accepting this approval, you also accept your responsibilities associated with this approval. Details of your responsibilities are attached. Please print and retain.

Information Letter:

A copy of your approved protocol is attached. However you still need to add the following IRB approval information to your information letter(s): "The Auburn University Institutional Review Board has approved this document for use from November 5, 2020 to ------ Protocol #20-535 EX 2011, Lee.

You must use the updated document(s) to consent participants.

Expiration:

Continuing review of this Exempt protocol is not required; however, all modification/revisions to the approved protocol must be reviewed and approved by the IRB.

When you have completed all research activities, have no plans to collect additional data and have destroyed all identifiable information as approved by the IRB, notify Office of the IRB via e-mail. A final report is **not** required for Exempt protocols.

Best wishes for success with your research!

IRB Admin Office of Research Compliance Auburn University 115 Ramsay Hall Auburn, AL 36849

https://outlook.office.com/mail/id/AAQkAGIyNWFmNTM4LTgzYTItNGVINi1iMThiLTQzOWExYzg3MGYwMAAQAJvMYG4CUkMipwADUZZOEys%3D?... 1/2